

29 April 2022

(22-3329)

Page: 1/92

Original: English

EUROPEAN UNION – SAFEGUARD MEASURES ON CERTAIN STEEL PRODUCTS (DS595)

REPORT OF THE PANEL

- 2 -

TABLE OF CONTENTS

1	INTRODUCTION
1.1	Complaint by Turkey12
1.2	Panel establishment and composition12
1.3	Panel proceedings12
2	FACTUAL ASPECTS: THE MEASURES AT ISSUE
3	PARTIES' REQUESTS FOR FINDINGS AND RECOMMENDATIONS
4	ARGUMENTS OF THE PARTIES15
5	ARGUMENTS OF THE THIRD PARTIES
6	INTERIM REVIEW
7	FINDINGS
7.1	
7.2	
	 The applicable requirements of Article XIX:1(a) of the GATT 1994 and Articles 2.1, 3.1, .(c), 4.2(a), 4.2(b), and 4.2(c) of the Agreement on Safeguards
Art	2.2 Whether Turkey has established that the definitive safeguard is inconsistent with icle XIX:1(a) of the GATT 1994 and Articles 2.1, 3.1, 4.1(c), 4.2(a), 4.2(b), and 4.2(c) of the reement on Safeguards with regard to product scope21
a s	2.2.1 The alleged mismatch between the products on which the European Commission applied afeguard and the products for which the European Commission investigated the existence of a necessary circumstances and conditions
7.2	2.2.1.1 Article XIX:1(a) of the GATT 1994 and Article 2.1 of the Agreement on Safeguards22
7.2	2.2.1.2 Articles 4.1(c), 4.2(a), and 4.2(b) of the Agreement on Safeguards28
7.2	2.2.2 The alleged internal inconsistency in the approach to product scope
7.3	3 Unforeseen developments
7.3	3.1 The applicable requirements of Article XIX:1(a) of the GATT 199431
7.3 Art	B.2 Whether Turkey has established that the definitive safeguard is inconsistent with icle XIX:1(a) of the GATT 1994 with regard to unforeseen developments
7.3	3.2.1 Whether the European Commission did not identify any unforeseen developments33
7.3	3.2.2 Whether the developments were not unforeseen
	3.2.3 Whether the European Commission did not establish that the increase in imports was a result of" the unforeseen developments40
	3.2.3.1 Whether the European Commission did not do enough to establish how the unforeseen velopments had resulted in the increase in imports into the European Union
7.3	3.2.3.2 Unforeseen developments relating to steel in general
7.3	3.2.3.3 The timing of the Section 232 measures48
	3.2.3.4 Overall conclusion on the connection between the unforeseen developments and the rease in imports
7.4	The effect of obligations
7.4	1.1 The applicable requirements of Article XIX:1(a) of the GATT 1994
7.4 Art	4.2 Whether Turkey has established that the definitive safeguard is inconsistent with icle XIX:1(a) of the GATT 1994 with regard to the "effect of the obligations"50
7.5	5 Increase in imports

7.5.1 The applicable requirements of Article XIX:1(a) of the GATT 1994 and Articles 2.1 and 4.2(a) of the Agreement on Safeguards
7.5.2 Whether Turkey has established that the definitive safeguard is inconsistent with Article XIX:1(a) of the GATT 1994 and Articles 2.1 and 4.2(a) of the Agreement on Safeguards52
7.5.2.1 Whether the European Commission did not adequately account for decreases in imports in the most recent past
7.5.2.2 Whether the European Commission did not demonstrate that the increase in imports was sufficiently sharp, significant, sudden, and recent
7.5.2.3 Overall conclusion on the European Commission's determination of an increase in imports
7.6 Threat of serious injury
7.6.1 The applicable requirements of Articles 4.1(a), 4.1(b), and 4.2(a) of the Agreement on Safeguards
7.6.2 Whether Turkey has established that the definitive safeguard is inconsistent with Articles 4.1(a), 4.1(b), and 4.2(a) of the Agreement on Safeguards
7.6.2.1 Whether the European Commission did not provide a reasoned and adequate explanation of its determination of threat of serious injury
7.6.2.1.1 Whether any improvement in the domestic industry's situation necessarily negates a finding of threat of serious injury
7.6.2.1.2 Whether the European Commission adequately explained the basis for the domestic industry's improved performance
7.6.2.1.3 Whether the European Commission improperly inferred that improvements owing to the AD/CV measures would be temporary and reversible
7.6.2.1.4 Whether the European Commission identified other factors contributing to the domestic industry's improved performance that demonstrate its vulnerability to increased imports
7.6.2.1.5 Conclusion on the explanation for the determination of threat of serious injury
7.6.2.2 Whether the European Commission did not establish a high likelihood of serious injury in the very near future
7.6.2.3 Turkey's consequential arguments relating to the European Commission's determination of threat of serious injury
7.6.2.4 Overall conclusion on threat of serious injury
7.7 Causation
7.8 Applying the safeguard only to the extent and time necessary to prevent serious injury70
7.8.1 Article 5.1 of the Agreement on Safeguards71
7.8.1.1 The applicable requirements of Article 5.1 of the Agreement on Safeguards71
7.8.1.2 Whether Turkey has established that the definitive safeguard is inconsistent with Article 5.1 of the Agreement on Safeguards72
7.8.1.2.1 Whether the definitive safeguard is inconsistent with Article 5.1 because the European Commission did not take into account data from the first six months of 2018 in determining the size of the TRQs72
7.8.1.2.2 Whether the definitive safeguard is inconsistent with Article 5.1 because the European Commission did not conduct a proper causation analysis
7.8.1.2.3 Whether the double remedy regulation rendered the definitive safeguard inconsistent with Article 5.1
7.8.2 Article 7.1 of the Agreement on Safeguards and Article XIX:1(a) of the GATT 199476
7.9 The allocation of shares in the TRQs76
7.9.1 Article XIII:2 of the GATT 199476

7.9.1.1 The applicable requirements of Article XIII:2 of the GATT 199477
7.9.1.2 Whether Turkey has established that the definitive safeguard is inconsistent with Article XIII:2(d) of the GATT 199478
7.9.1.2.1 The representative period argument
7.9.1.2.2 The special factor argument81
7.9.1.3 Whether Turkey has established that the definitive safeguard is inconsistent with the <i>chapeau</i> of Article XIII:2 of the GATT 199482
7.9.2 Article 5.2(a) of the Agreement on Safeguards
7.10 Reduction in the pace of liberalization and increased restrictiveness
7.10.1 Article 7.4 of the Agreement on Safeguards
7.10.1.1 The applicable requirements of Article 7.4 of the Agreement on Safeguards
7.10.1.2 Whether Turkey has established that the definitive safeguard is inconsistent with Article 7.4 of the Agreement on Safeguards
7.10.1.2.1 Whether the reduction in the pace of liberalization was inconsistent with Article 7.4
7.10.1.2.2 Whether certain modifications to the definitive safeguard resulted in an inconsistency with Article 7.4
7.10.2 Article 5.1 of the Agreement on Safeguards
7.11 Article II:1 of the GATT 199490
8 CONCLUSIONS, RECOMMENDATION, AND REQUEST FOR A SUGGESTION

- 5 -

LIST OF ANNEXES

ANNEX A

PANEL DOCUMENTS

	Contents	Page
Annex A-1	Working Procedures of the Panel	4
Annex A-2	Additional Working Procedures of the Panel for the first substantive meeting	11
Annex A-3	Additional Working Procedures of the Panel for the second substantive meeting	15
Annex A-4	Additional Working Procedures of the Panel	18
Annex A-5	Interim Review	19

ANNEX B

ARGUMENTS OF THE PARTIES

Contents		
Annex B-1	Integrated executive summary of the arguments of Turkey	39
Annex B-2	Integrated executive summary of the arguments of the European Union	57

ANNEX C

ARGUMENTS OF THE THIRD PARTIES

	Contents	Page
Annex C-1	Integrated executive summary of the arguments of Argentina	78
Annex C-2	Integrated executive summary of the arguments of Brazil	81
Annex C-3	Integrated executive summary of the arguments of Canada	84
Annex C-4	Integrated executive summary of the arguments of Japan	86
Annex C-5	Integrated executive summary of the arguments of the Republic of Korea	90
Annex C-6	Integrated executive summary of the arguments of Switzerland	95
Annex C-7	Integrated executive summary of the arguments of Ukraine	100
Annex C-8	Integrated executive summary of the arguments of the United Kingdom	102
Annex C-9	Integrated executive summary of the arguments of the United States	105

- 6 -

CASES CITED IN THIS REPORT

Short Title	Full Case Title and Citation
Argentina – Footwear (EC)	Appellate Body Report, Argentina – Safeguard Measures on Imports of
	Footwear, WT/DS121/AB/R, adopted 12 January 2000, DSR 2000:I, p. 515
Argentina – Footwear (EC)	Panel Report, Argentina – Safeguard Measures on Imports of Footwear, <u>WT/DS121/R</u> , adopted 12 January 2000, as modified by Appellate Body Report WT/DS121/AB/R, DSR 2000:II, p. 575
Argentina – Poultry Anti-Dumping Duties	Panel Report, <i>Argentina – Definitive Anti-Dumping Duties on Poultry from</i> <i>Brazil</i> , <u>WT/DS241/R</u> , adopted 19 May 2003, DSR 2003:V, p. 1727
Argentina – Preserved Peaches	Panel Report, Argentina – Definitive Safeguard Measure on Imports of Preserved Peaches, <u>WT/DS238/R</u> , adopted 15 April 2003, DSR 2003:III, p. 1037
Argentina – Textiles and Apparel	Panel Report, <i>Argentina – Measures Affecting Imports of Footwear, Textiles, Apparel and Other Items</i> , <u>WT/DS56/R</u> , adopted 22 April 1998, as modified by Appellate Body Report WT/DS56/AB/R, DSR 1998:III, p. 1033
Australia – Anti-Dumping Measures on Paper	Panel Report, Australia – Anti-Dumping Measures on A4 Copy Paper, WT/DS529/R and Add.1, adopted 28 January 2020
Chile – Price Band System	Panel Report, <i>Chile – Price Band System and Safeguard Measures Relating to Certain Agricultural Products</i> , <u>WT/DS207/R</u> , adopted 23 October 2002, as modified by Appellate Body Report WT/DS207AB/R, DSR 2002:VIII, p. 3127
China – Electronic Payment Services	Panel Report, <i>China – Certain Measures Affecting Electronic Payment Services</i> , <u>WT/DS413/R</u> and Add.1, adopted 31 August 2012, DSR 2012:X, p. 5305
Dominican Republic – Safeguard Measures	Panel Report, <i>Dominican Republic – Safeguard Measures on Imports of Polypropylene Bags and Tubular Fabric</i> , <u>WT/DS415/R</u> , <u>WT/DS416/R</u> , <u>WT/DS417/R</u> , <u>WT/DS418/R</u> , and Add.1, adopted 22 February 2012, DSR 2012:XIII, p. 6775
<i>EC – Bananas III (Article 21.5 – Ecuador II) / EC – Bananas III (Article 21.5 – US)</i>	Appellate Body Reports, European Communities – Regime for the Importation, Sale and Distribution of Bananas – Second Recourse to Article 21.5 of the DSU by Ecuador, <u>WT/DS27/AB/RW2/ECU</u> , adopted 11 December 2008, and Corr.1 / European Communities – Regime for the Importation, Sale and Distribution of Bananas – Recourse to Article 21.5 of the DSU by the United States, <u>WT/DS27/AB/RW/USA</u> and Corr.1, adopted 22 December 2008, DSR 2008:XVIII, p. 7165
EC – Bananas III (Article 21.5 – Ecuador)	Panel Report, <i>European Communities – Regime for the Importation, Sale and Distribution of Bananas – Recourse to Article 21.5 of the DSU by Ecuador</i> , <u>WT/DS27/RW/ECU</u> , adopted 6 May 1999, DSR 1999:II, p. 803
EC – Export Subsidies on Sugar (Australia)	Panel Report, European Communities – Export Subsidies on Sugar, Complaint by Australia, <u>WT/DS265/R</u> , adopted 19 May 2005, as modified by Appellate Body Report WT/DS265/AB/R, WT/DS266/AB/R, WT/DS283/AB/R, DSR 2005:XIII, p. 6499
EC – Export Subsidies on Sugar (Brazil)	Panel Report, <i>European Communities – Export Subsidies on Sugar,</i> <i>Complaint by Brazil</i> , <u>WT/DS266/R</u> , adopted 19 May 2005, as modified by Appellate Body Report WT/DS265/AB/R, WT/DS266/AB/R, WT/DS283/AB/R, DSR 2005:XIV, p. 6793
EC – Export Subsidies on Sugar (Thailand)	Panel Report, European Communities – Export Subsidies on Sugar, Complaint by Thailand, <u>WT/DS283/R</u> , adopted 19 May 2005, as modified by Appellate Body Report WT/DS265/AB/R, WT/DS266/AB/R, WT/DS283/AB/R, DSR 2005:XIV, p. 7071
EC – Trademarks and Geographical Indications (Australia)	Panel Report, European Communities – Protection of Trademarks and Geographical Indications for Agricultural Products and Foodstuffs, Complaint by Australia, <u>WT/DS290/R</u> , adopted 20 April 2005, DSR 2005:X, p. 4603
EC – Trademarks and Geographical Indications (US)	Panel Report, <i>European Communities – Protection of Trademarks and Geographical Indications for Agricultural Products and Foodstuffs, Complaint by the United States</i> , <u>WT/DS174/R</u> , adopted 20 April 2005, DSR 2005:VIII, p. 3499
EU – PET (Pakistan)	Panel Report, <i>European Union – Countervailing Measures on Certain</i> <i>Polyethylene Terephthalate from Pakistan</i> , <u>WT/DS486/R</u> , Add.1 and Corr.1, adopted 28 May 2018, as modified by Appellate Body Report WT/DS486/AB/R, DSR 2018:IV, p. 1739

- 7 -

Short Title	Full Case Title and Citation
EU – Poultry Meat (China)	Panel Report, <i>European Union – Measures Affecting Tariff Concessions on Certain Poultry Meat Products</i> , <u>WT/DS492/R</u> and Add.1, adopted 19 April 2017, DSR 2017:III, p. 1067
Guatemala – Cement II	Panel Report, <i>Guatemala – Definitive Anti-Dumping Measures on Grey</i> <i>Portland Cement from Mexico</i> , <u>WT/DS156/R</u> , adopted 17 November 2000, DSR 2000:XI, p. 5295
India – Iron and Steel Products	Panel Report, <i>India – Certain Measures on Imports of Iron and Steel Products</i> , <u>WT/DS518/R</u> and Add.1, circulated to WTO Members 6 November 2018 [appealed by India 14 December 2018 – the Division suspended its work on 10 December 2019]
India – Quantitative Restrictions	Panel Report, <i>India – Quantitative Restrictions on Imports of Agricultural, Textile and Industrial Products</i> , <u>WT/DS90/R</u> , adopted 22 September 1999, upheld by Appellate Body Report WT/DS90/AB/R, DSR 1999:V, p. 1799
Korea – Dairy	Appellate Body Report, <i>Korea – Definitive Safeguard Measure on Imports of Certain Dairy Products</i> , <u>WT/DS98/AB/R</u> , adopted 12 January 2000, DSR 2000:I, p. 3
Mexico – Steel Pipes and Tubes	Panel Report, <i>Mexico – Anti-Dumping Duties on Steel Pipes and Tubes from Guatemala</i> , <u>WT/DS331/R</u> , adopted 24 July 2007, DSR 2007:IV, p. 1207
Pakistan – BOPP Film (UAE)	Panel Report, <i>Pakistan – Anti-Dumping Measures on Biaxially Oriented</i> <i>Polypropylene Film from the United Arab Emirates</i> , <u>WT/DS538/R</u> and Add.1, circulated to WTO Members 18 January 2021 [appealed by Pakistan 22 February 2021]
Ukraine – Passenger Cars	Panel Report, Ukraine – Definitive Safeguard Measures on Certain Passenger Cars, <u>WT/DS468/R</u> and Add.1, adopted 20 July 2015, DSR 2015:VI, p. 3117
US – Certain EC Products	Appellate Body Report, United States – Import Measures on Certain Products from the European Communities, <u>WT/DS165/AB/R</u> , adopted 10 January 2001, DSR 2001:I, p. 373
US – Continued Zeroing	Appellate Body Report, <i>United States – Continued Existence and Application of Zeroing Methodology</i> , <u>WT/DS350/AB/R</u> , adopted 19 February 2009, DSR 2009:III, p. 1291
US – Cotton Yarn	Panel Report, <i>United States – Transitional Safeguard Measure on Combed Cotton Yarn from Pakistan</i> , <u>WT/DS192/R</u> , adopted 5 November 2001, as modified by Appellate Body Report WT/DS192/AB/R, DSR 2001:XII, p. 6067
US – Fur Felt Hats	GATT Working Party Report, <i>Report on the Withdrawal by the United States of a Tariff Concession under Article XIX of the General Agreement on Tariffs and Trade</i> , adopted 22 October 1951, <u>GATT/CP/106</u>
US – Lamb	Appellate Body Report, United States – Safeguard Measures on Imports of Fresh, Chilled or Frozen Lamb Meat from New Zealand and Australia, <u>WT/DS177/AB/R</u> , <u>WT/DS178/AB/R</u> , adopted 16 May 2001, DSR 2001:IX, p. 4051
US – Lead and Bismuth II	Panel Report, United States – Imposition of Countervailing Duties on Certain Hot-Rolled Lead and Bismuth Carbon Steel Products Originating in the United Kingdom, <u>WT/DS138/R</u> and Corr.2, adopted 7 June 2000, upheld by Appellate Body Report WT/DS138/AB/R, DSR 2000:VI, p. 2623
US – Line Pipe	Appellate Body Report, United States – Definitive Safeguard Measures on Imports of Circular Welded Carbon Quality Line Pipe from Korea, <u>WT/DS202/AB/R</u> , adopted 8 March 2002, DSR 2002:IV, p. 1403
US – Line Pipe	Panel Report, United States – Definitive Safeguard Measures on Imports of Circular Welded Carbon Quality Line Pipe from Korea, <u>WT/DS202/R</u> , adopted 8 March 2002, as modified by Appellate Body Report WT/DS202/AB/, DSR 2002:IV, p. 1473
US – Offset Act (Byrd Amendment)	Panel Report, <i>United States – Continued Dumping and Subsidy Offset Act of 2000</i> , <u>WT/DS217/R</u> , <u>WT/DS234/R</u> , adopted 27 January 2003, as modified by Appellate Body Report WT/DS217/AB/R, WT/DS234/AB/R, DSR 2003:II, p. 489
US – Safeguard Measure on PV Products (China)	Panel Report, <i>United States – Safeguard Measure on Imports of Crystalline Silicon Photovoltaic Products</i> , <u>WT/DS562/R</u> and Add.1, circulated to WTO Members 2 September 2021 [appealed by China 16 September 2021]

- 8 -

Short Title	Full Case Title and Citation
US – Steel Safeguards	Appellate Body Report, United States – Definitive Safeguard Measures on Imports of Certain Steel Products, <u>WT/DS248/AB/R</u> , <u>WT/DS249/AB/R</u> , <u>WT/DS251/AB/R</u> , <u>WT/DS252/AB/R</u> , <u>WT/DS253/AB/R</u> , <u>WT/DS254/AB/R</u> , <u>WT/DS258/AB/R</u> , <u>WT/DS259/AB/R</u> , adopted 10 December 2003, DSR 2003:VII, p. 3117
US – Steel Safeguards	Panel Reports, United States – Definitive Safeguard Measures on Imports of Certain Steel Products, <u>WT/DS248/R</u> and Corr.1 / <u>WT/DS259/R</u> and Corr.1 / <u>WT/DS251/R</u> and Corr.1 / <u>WT/DS253/R</u> and Corr.1 / <u>WT/DS254/R</u> and Corr.1 / <u>WT/DS259/R</u> and Corr.1 / <u>WT/DS258/AB/R</u> , WT/DS259/AB/R, DSR 2003:VIII, p. 3273
US – Underwear	Panel Report, United States – Restrictions on Imports of Cotton and Man-made Fibre Underwear, <u>WT/DS24/R</u> , adopted 25 February 1997, as modified by Appellate Body Report WT/DS24/AB/R, DSR 1997:I, p. 31

- 9 -

EXHIBITS REFERRED TO IN THIS REPORT

Exhibit	Short title (if any)	Description
EU-2	Short title (Ir ally)	WTO statistics on antidumping initiations by sector
EU-3		
	Due soutetien et the	WTO statistics on countervailing initiations by sector
EU-7	Presentation at the 83 rd session of the OECD Steel Committee	H. Otsuka, "Recent developments in steel trade and trade policies" presentation at the 83 rd session of the OECD Steel Committee held in Paris on 28 September 2017
EU-8	Chair's statement at the 83 rd session of the OECD Steel Committee	Statement dated 28 September 2017 of L. Top, Chair of the OECD Steel Committee at the 83^{rd} session of the OECD Steel Committee
EU-9	Chair's statement at the 84 th session of the OECD Steel Committee	Statement dated 5 March 2018 of the Chair at the 84^{th} session of the OECD Steel Committee
EU-10		Cover letter (11 April 2018)
EU-12		Calculation deflection
EU-13		Estimate of imports to the US that will be deflected
EU-16	Presentation at the 84 th session of the OECD Steel Committee	OECD presentation, "Steel trade and trade policy developments: a closer look at NTMs", 84^{th} session of the OECD Steel Committee
EU-17		Global Forum on Steel Excess Capacity Report (30 November 2017)
EU-23	European Commission's Communication "Steel: Preserving Sustainable Jobs and Growth"	Communication from the Commission to the European Parliament, the European Council, the Council, the European Economic and Social Committee, the Committee of the Regions and the European Investment Bank, "Steel: Preserving sustainable jobs and growth in Europe" (16 March 2016)
TUR-1	Notice of initiation	Notice of initiation of a safeguard investigation concerning imports of steel products, Official Journal of the European Union, C Series, No. 111 (26 March 2018) pp. 29-35
TUR-2	Amendment to the notice of initiation	Notice amending the notice of initiation of a safeguard investigation concerning the imports of steel products, Official Journal of the European Union, C Series, No. 225 (28 June 2018), pp. 54-56
TUR-3	Provisional determination	Commission Implementing Regulation (EU) No. 2018/1013 of 17 July 2018 imposing provisional safeguard measures with regard to imports of certain steel products, Official Journal of the European Union, L Series, No. 181 (18 July 2018), pp. 39-83
TUR-5	Definitive determination	Commission Implementing Regulation (EU) No. 2019/159 of 31 January 2019 imposing definitive safeguard measures against imports of certain steel products, Official Journal of the European Union, Series L, No. 31 (1 February 2019), pp. 27-74
TUR-7	Double remedy regulation	Commission Implementing Regulation (EU) No. 2019/1382 of 2 September 2019 amending certain regulations imposing anti-dumping or anti-subsidy measures on certain steel products subject to safeguard measures, Official Journal of the European Union, Series L, No. 227 (3 September 2019), pp. 1-25
TUR-9	First review regulation	Commission Implementing Regulation (EU) No. 2019/1590 of 26 September 2019 amending Implementing Regulation (EU) No. 2019/159 imposing definitive safeguard measures against imports of certain steel products, Official Journal of the European Union, Series L, No. 248 (27 September 2019), pp. 28-64
TUR-10	European Commission Implementing Regulation (EU) 2020/35	Commission Implementing Regulation (EU) 2020/35 of 15 January 2020 amending Implementing Regulation (EU) 2019/159 imposing definitive safeguard measures against imports of certain steel products, Official Journal of the European Union, Series L, No. 12 (16 January 2020), pp. 13-16
TUR-12	Second review regulation	Commission Implementing Regulation (EU) No. 2020/894 of 29 June 2020 amending Implementing Regulation (EU) No. 2019/159 imposing definitive safeguard measures against imports of certain steel products, Official Journal of the European Union, Series L, No. 206 (30 June 2020), pp. 27-62
TUR-14	Questionnaire for exporting producers	European Commission, safeguard questionnaire for exporting producers
TUR-15	Questionnaire for domestic producers	European Commission, safeguard questionnaire for Union producers
TUR-18		Note to the file (11 April 2018)
TUR-19		Updated note to the file (2 July 2018)
TUR-21	Global Trade Alert Report, "Going Spare"	S. Evenett and J. Fritz, <i>Going Spare: Steel, Excess Capacity, and Protectionism (The 22nd Global Trade Alert Report)</i> (CEPR Press, May 2018)
TUR-39		Commission Regulation (EC) No 1694/2002 of 27 September 2002 imposing definitive safeguard measures against imports of certain steel products, Official Journal of the European Union, L Series, No. 261 (28 September 2002), pp. 1-123

- 10 -

Exhibit	Short title (if any)	Description
TUR-40		Questionnaire response of ThyssenKrupp Steel Europe GmbH
TUR-49		List of anti-dumping and countervailing measures in place at the time of the
		imposition of the provisional safeguard measures

- 11 -

ABBREVIATIONS USED IN THIS REPORT

Abbreviation	Description
AD/CV	anti-dumping/countervailing
Anti-Dumping Agreement	Agreement on Implementation of Article VI of the General Agreement on Tariffs and Trade 1994
DSB	Dispute Settlement Body
DSU	Understanding on Rules and Procedures Governing the Settlement of Disputes
GATT 1994	General Agreement on Tariffs and Trade 1994
MRP	most recent period
POI	period of investigation
PV	photovoltaic
SCM Agreement	Agreement on Subsidies and Countervailing Measures
SPS	sanitary and phytosanitary
TRQ	tariff-rate quota
WTO	World Trade Organization

1 INTRODUCTION

1.1 Complaint by Turkey

1.1. On 13 March 2020, Turkey requested consultations with the European Union pursuant to Articles 1 and 4 of the Understanding on Rules and Procedures Governing the Settlement of Disputes (DSU), Article XXIII:1 of the General Agreement on Tariffs and Trade 1994 (GATT 1994), and Article 14 of the Agreement on Safeguards with respect to the measures and claims set out below.¹

1.2. Consultations were held on 29 April 2020 but failed to resolve the dispute.

1.2 Panel establishment and composition

1.3. On 16 July 2020, Turkey requested the establishment of a panel pursuant to Article 6 of the DSU and Article 14 of the Agreement on Safeguards with standard terms of reference.² At its meeting on 28 August 2020, the Dispute Settlement Body (DSB) established a panel pursuant to the request of Turkey in document WT/DS595/3, in accordance with Article 6 of the DSU.³

1.4. The Panel's terms of reference are the following:

To examine, in the light of the relevant provisions of the covered agreements cited by the parties to the dispute, the matter referred to the DSB by Turkey in document WT/DS595/3 and to make such findings as will assist the DSB in making the recommendations or in giving the rulings provided for in those agreements.⁴

1.5. On 29 September 2020, the parties agreed that the panel would be composed as follows:

Chairperson: Mr William Davey Members: Ms Silvia Lorena Hooker Ortega Mr Marco Tulio Molina Tejeda

1.6. Argentina, Brazil, Canada, China, India, the Republic of Korea, Japan, Norway, the Russian Federation, Switzerland, Chinese Taipei, Ukraine, the United Arab Emirates, the United Kingdom, and the United States notified their interest in participating in the Panel proceedings as third parties.

1.3 Panel proceedings

1.7. After consultation with the parties, the Panel adopted its Working Procedures⁵ on 26 October 2020, the Additional Working Procedures of the Panel for the first substantive meeting⁶ on 31 March 2021, and the Additional Working Procedures of the Panel for the second substantive meeting⁷ on 2 June 2021. To give effect to certain requests made jointly by the parties on 22 March 2022, and after consulting the parties, the Panel also adopted further Additional Working Procedures on 1 April 2022.⁸

1.8. After consultation with the parties, the Panel adopted a partial timetable on 26 October 2020 and revised the timetable on 25 November 2020, 9 February 2021, 22 March 2021, 2 June 2021, 25 October 2021, and 3 November 2021. The timetable adopted by the Panel on 26 October 2020 envisaged that the European Union would file its first written submission by 22 December 2020. However, on 23 November 2020, the European Union requested the Panel to extend the deadline for the submission of its first written submission to 15 January 2021. On 24 November 2020,

¹ Request for consultations by Turkey, WT/DS595/1 (Turkey's consultation request).

² Request for establishment of a panel by Turkey, WT/DS595/3 (Turkey's panel request).

³ DSB, Minutes of the meeting held on 28 August 2020, WT/DSB/M/444.

⁴ Constitution note of the Panel, WT/DS595/4.

⁵ Working Procedures of the Panel (Annex A-1).

⁶ Additional Working Procedures of the Panel for the first substantive meeting (Annex A-2).

⁷ Additional Working Procedures of the Panel for the second substantive meeting (Annex A-3).

⁸ See Agreed procedures for arbitration under Article 25 of the DSU, WT/DS595/10; Additional Working Procedures of the Panel (Annex A-4); and paragraph 1.9 below.

Turkey opposed the European Union's request. After considering both parties' positions, the Panel granted the European Union's request and accordingly revised the timetable on 25 November 2020. On 8 February 2021, the European Union requested the Panel to extend (a) the deadline for filing responses to the questions sent by the Panel to the parties on 5 February 2021 from 12 February 2021 to 26 February 2021; and (b) the deadline for filing the second written submission from 5 March 2021 to 19 March 2021. On 9 February 2021, Turkey opposed the European Union's request. After considering both parties' positions, the Panel revised the timetable on 9 February 2021, partially granting the European Union's request by changing the two deadlines to 19 February 2021 and 12 March 2021, respectively. On 27 May 2021, the European Union asked the Panel to extend the deadline for filing comments on the other party's responses to the Panel's questions after the first substantive meeting from 4 June 2021 to 18 June 2021. On 27 May 2021, Turkey opposed the European Union's request. After considering both parties' positions, the Panel partially granted the European Union's request by changing the deadline to 8 June 2021. The revised timetable issued by the Panel on 25 October 2021 set the date for the parties to request review of precise aspects of the interim report and/or to request an interim review meeting as 5 November 2021, and the deadline for the parties to comment on the other party's review requests and for the interim review meeting (if requested) as 12 November 2021. On 2 November 2021, Turkey requested the Panel to change the two dates to 12 November 2021 and 19 November 2021 respectively. The European Union did not object to Turkey's request. In light of both parties' positions, the Panel revised the timetable on 3 November 2021, granting Turkey's request.

1.9. The Panel held a first substantive meeting with the parties on 4, 5, and 6 May 2021. A session with the third parties took place on 5 May 2021. According to paragraph 15 of the Working Procedures adopted by the Panel, the Panel retained the discretion to hold a second substantive meeting with the parties upon request by either party. On 17 May 2021, the European Union requested the Panel to hold a second substantive meeting. On 20 May 2021, Turkey opposed the European Union's request. After considering both parties' positions, the Panel granted the European Union's request and held a second substantive meeting with the parties on 22 and 23 June 2021. Because of restrictions due to the COVID-19 pandemic, the Panel conducted both substantive meetings via secure videoconference.9 On 19 August 2021, the Panel issued the descriptive part of its Report to the parties. The Panel issued its Interim Report to the parties on 29 October 2021. The Panel issued its Final Report to the parties on 10 December 2021. The Panel Report was scheduled to be circulated on 21 December 2021. On 20 December 2021, Turkey requested the Panel to suspend its work, pursuant to Article 12.12 of the DSU, until 21 January 2022, and the European Union did not object. The Panel granted the suspension request.¹⁰ Subsequently, Turkey filed three requests to extend the suspension of the Panel's work pursuant to Article 12.12 of the DSU, until 11 February 2022, 25 February 2022, and 25 March 2022, respectively. The European Union did not object to any of the three requests. The Panel granted these requests.¹¹ On 22 March 2022, the European Union and Turkey transmitted to the Panel a document signed by both parties entitled "Agreed Procedures for Arbitration under Article 25 of the DSU" (Agreed Procedures).¹² Through the Agreed Procedures, the parties jointly requested the Panel to extend indefinitely the suspension of its work pursuant to Article 12.12 of the DSU, except to the extent necessary for the Panel to effect certain steps set out in the Agreed Procedures. On 24 March 2022, the Panel decided to grant that suspension request.¹³ To give effect to the parties' requests, made through the Agreed Procedures, that the Panel undertake certain steps relating to the report of the Panel and to the record of the Panel proceedings, the Panel adopted the Additional Working Procedures referred to at paragraph 1.7 above. Through the Agreed Procedures, paragraph 6, the parties also jointly requested the Panel to resume its work if neither party initiated arbitration under the Agreed Procedures within 30 days from the date on which the Agreed Procedures were notified to the DSB. Neither party initiated such arbitration, and on 26 April 2022 the parties confirmed their request that the Panel resume its work. The Panel Report was therefore circulated on 29 April 2022.

⁹ The Panel adopted additional working procedures for each of these meetings. (See fns 6-7 above).

¹⁰ See Suspension of Panel Work dated 21 December 2021, WT/DS595/6.

¹¹ See Suspension of Panel Work dated 20 January 2022, WT/DS595/7; Suspension of Panel Work dated 10 February 2022, WT/DS595/8; and Suspension of Panel Work dated 25 February 2022, WT/DS595/9.

¹² See Agreed procedures for arbitration under Article 25 of the DSU, WT/DS595/10.

¹³ See Communication from the Panel dated 24 March 2022, WT/DS595/11.

2 FACTUAL ASPECTS: THE MEASURES AT ISSUE

2.1. In its panel request, Turkey identified the measures at issue as "the provisional and definitive safeguard measures imposed by the European Union on the imports of certain steel products and the investigation leading to the imposition of those measures". Turkey indicated that those measures "cover[ed] all decisions, notices, notifications and regulations" specifically mentioned in the panel request, and "include[d] any amendments, supplements, reviews, replacement, renewals, extensions, implementing measures and any other related measures taken by the European Union in relation to the investigation and/or the safeguard measures at issue".¹⁴ Turkey repeated this formulation in its first written submission.¹⁵

2.2. In response to a question from the Panel, Turkey clarified that its challenge to the "investigation" leading to the imposition of the provisional and definitive safeguards was not distinct from its challenge to the provisional and definitive safeguards themselves.¹⁶

2.3. The legal instrument setting out the provisional safeguard measures on certain steel products (the "provisional safeguard") is Commission Implementing Regulation (EU) No. 2018/1013 of 17 July 2018 imposing provisional safeguard measures with regard to imports of certain steel products (the "provisional determination").¹⁷

2.4. The legal instruments setting out the definitive safeguard measures on certain steel products ("definitive safeguard"¹⁸) include:

- a. Commission Implementing Regulation (EU) No. 2019/159 of 31 January 2019 imposing definitive safeguard measures against imports of certain steel products (the "definitive determination")¹⁹;
- b. Commission Implementing Regulation (EU) No. 2019/1382 of 2 September 2019 amending certain regulations imposing anti-dumping or anti-subsidy measures on certain steel products subject to safeguard measures (the "double remedy regulation")²⁰;
- c. Commission Implementing Regulation (EU) No. 2019/1590 of 26 September 2019 amending Implementing Regulation (EU) No. 2019/159 imposing definitive safeguard measures against imports of certain steel products (the "first review regulation")²¹;
- d. Commission Implementing Regulation (EU) No. 2020/35 of 15 January 2020 amending Implementing Regulation (EU) No. 2019/159 imposing definitive safeguard measures against imports of certain steel products²²; and
- e. Commission Implementing Regulation (EU) No. 2020/894 of 29 June 2020 amending Implementing Regulation (EU) No. 2019/159 imposing definitive safeguard measures against imports of certain steel products (the "second review regulation").²³

¹⁴ Turkey's panel request, para. 17. Paragraph 17 of the panel request refers to sections 1.1-1.4 of the panel request, which describe in more detail the legal instruments comprising the challenged measures. (See also Turkey's request for consultations, para. 17).

¹⁵ Turkey's first written submission, para. 34.

¹⁶ Turkey's response to Panel question No. 15, para. 1.

¹⁷ Provisional determination, (Exhibit TUR-3).

¹⁸ Turkey requested the Panel to refer to this as "definitive safeguards", in the plural.

⁽Turkey's comments on the descriptive part of the Panel Report, para. 2.4). The Panel has rejected the request, without prejudice to the Panel's consideration of the question whether there is a single definitive safeguard or multiple definitive safeguards, which the Panel discusses in paras. 7.43-7.57 below. Turkey made an equivalent request also for the provisional safeguard. (Turkey's comments on the descriptive part of the Panel Report, para. 2.3).

¹⁹ Definitive determination, (Exhibit TUR-5). The definitive determination incorporates by reference certain findings and reasoning set out in the provisional determination.

²⁰ Double remedy regulation, (Exhibit TUR-7).

²¹ First review regulation, (Exhibit TUR-9).

²² European Commission Implementing Regulation (EU) 2020/35, (Exhibit TUR-10).

²³ Second review regulation, (Exhibit TUR-12).

3 PARTIES' REQUESTS FOR FINDINGS AND RECOMMENDATIONS

3.1. Turkey requests the Panel to find that the measures at issue are inconsistent with Articles 2.1, 3.1, 4.1(a), 4.1(b), 4.1(c), 4.2(a), 4.2(b), 4.2(c), 5.1, 5.2(a), 6, 7.1, and 7.4 of the Agreement on Safeguards, and with Articles II:1(b), XIII:2 *chapeau*, XIII:2(d), and XIX:1(a) of the GATT 1994.²⁴

3.2. Turkey requests, pursuant to Article 19.1 of the DSU, that the Panel recommend that the European Union bring its measures into conformity with its obligations under the Agreement on Safeguards and the GATT 1994.²⁵ In addition, Turkey requests the Panel to suggest, pursuant to the second sentence of Article 19.1 of the DSU, that the European Union implement that recommendation by revoking the measures at issue.²⁶

3.3. The European Union requests that the Panel reject Turkey's claims in this dispute in their entirety. $^{\rm 27}$

4 ARGUMENTS OF THE PARTIES

4.1. The arguments of the parties are reflected in their executive summaries, provided to the Panel in accordance with paragraph 22 of the Working Procedures adopted by the Panel (see Annexes B-1 and B-2).

5 ARGUMENTS OF THE THIRD PARTIES

5.1. The arguments of Argentina, Brazil, Canada, Japan, the Republic of Korea, Switzerland, Ukraine, the United Kingdom, and the United States are reflected in their executive summaries, provided in accordance with paragraph 22 of the Working Procedures adopted by the Panel (see Annexes C-1, C-2, C-3, C-4, C-5, C-6, C-7, C-8, and C-9). China, India, Norway, the Russian Federation, Chinese Taipei, and the United Arab Emirates did not provide the Panel with a third-party submission and did not make an opening oral statement at the third-party session of the first substantive meeting with the Panel.

6 INTERIM REVIEW

6.1. On 29 October 2021, the Panel issued its Interim Report to the parties. On 12 November 2021, Turkey and the European Union submitted their written requests for review. On 19 November 2021, the parties submitted comments on the other parties' written requests for review.

6.2. The parties' requests made at the interim review stage as well as the Panel's discussion and disposition of those requests are set out in Annex A-5.

7 FINDINGS

7.1. Turkey claims that the provisional and definitive safeguards on certain steel products adopted by the European Union in July 2018 and January 2019, respectively, are inconsistent with a number of provisions of the Agreement on Safeguards and the GATT 1994.

7.2. Below, we begin by considering whether we should make findings on both these measures, or only on the definitive safeguard, and we decide to make findings only on the definitive safeguard (section 7.1).

7.3. Our subsequent discussion follows the structure of Turkey's arguments. We therefore examine²⁸, in turn, the following aspects of the definitive safeguard: the European Commission's approach to product scope (section 7.2); unforeseen developments (section 7.3); the effect of obligations under the GATT 1994 (section 7.4); the increase in imports (section 7.5); threat of injury (section 7.6); causation (section 7.7); whether the safeguard was

²⁴ Turkey's first written submission, para. 381; second written submission, para. 268.

²⁵ Turkey's first written submission, para. 382; second written submission, para. 269.

²⁶ Turkey's first written submission, para. 382; second written submission, para. 269.

²⁷ European Union's first written submission, para. 313.

²⁸ In some instances, we exercise economy.

applied beyond the extent and time necessary to prevent serious injury (section 7.8); the allocation of shares in the tariff rate quotas that are one element of the definitive safeguard (section 7.9); whether the European Union did not progressively liberalize the safeguard and in fact made it more restrictive over time (section 7.10); and whether the out-of-quota duty that is another element of the definitive safeguard is a duty or charge inconsistent with Article II:1(b) of the GATT 1994 (section 7.11).

7.1 The measures on which the Panel will make findings

7.4. We recall that Turkey challenged (a) the provisional safeguard and (b) the definitive safeguard applied by the European Union on imports of certain steel products, as set out in more detail in section 2 above. Below, we discuss whether we should make findings on the provisional safeguard.

7.5. In response to a question from the Panel, Turkey has confirmed that the provisional safeguard is no longer in force, having been replaced by the definitive safeguard.²⁹ The European Union has not argued otherwise. It is therefore undisputed that the provisional safeguard is no longer in force.

7.6. At the same time, Turkey has observed that the regulation that applied the provisional safequard (which we refer to as the provisional determination) is still in force.³⁰ In response to a further question from the Panel, Turkey has indicated that this means that "the legal basis remains to collect provisional safeguard measures retroactively" and, in support for this proposition, refers to findings of the panel in India – Iron and Steel Products.³¹

7.7. In this regard, we note that in these proceedings whether the provisional determination continues to provide the basis to collect out-of-quota safeguard duties is a question of fact. The findings of a different panel regarding the effects of a different safeguard adopted by a different Member cannot serve to establish as a matter of fact the effects of the provisional determination adopted by the European Union. In this case, we note that the reference to the findings of the panel in India – Iron and Steel Products is the only support that Turkey has provided for its statement that duties could still be collected retroactively under the provisional safeguard. In other words, Turkey has not provided the Panel with any evidence that duties are either continuing to be collected, or could be collected, by the European Union under the provisional safeguard at issue, and it has also stated that the provisional safeguard is no longer in force.³²

7.8. Article 6.2 of the DSU does not limit to measures still in existence "the specific measures at issue" that can be brought before a panel. A panel may, therefore, issue findings on an expired measure where the panel believes it is necessary to do so to resolve a dispute. Conversely, a panel need not make findings on an expired measure where such findings are not necessary to resolve the dispute.³³ In addition, we note that when a panel does make findings on an expired measure, the fact that the measure has expired may affect the recommendations a panel may make pursuant to Article 19.1 of the DSU, which requires panels to make recommendations "to bring the measure into conformity" with the covered agreements.³⁴

7.9. In this instance, Turkey concedes that the provisional safeguard is no longer in force. Turkey claims that the provisional safeguard is inconsistent with Article 6 for a subset of the reasons for which it claims the definitive safequard is inconsistent with other provisions of the Agreement on Safeguards. Indeed, Turkey makes no distinction between its arguments against the provisional safequard and the corresponding arguments against the definitive safeguard.³⁵ Given that the provisional safeguard is no longer in force, and that Turkey challenges the provisional safeguard for some of the same reasons for which it challenges the definitive safeguard, we do not consider that

²⁹ Turkey's response to Panel question No. 1, paras. 2 and 4.

³⁰ Turkey's response to Panel question No. 1, para. 4.

³¹ Turkey's response to Panel question No. 16, para. 2 (referring to Panel Report, India – Iron and Steel Products [appealed by India on 14 December 2018, appealed by Japan on 21 December 2018]).

³² We also note that Article 9 of the definitive determination provides that any amounts paid under the provisional determination "shall be definitively collected", which indicates that it is the definitive determination that provides the basis for the definitive collection of duties on imports covered by the provisional safeguard.

³³ See e.g. Panel Reports, China – Electronic Payment Services, paras. 7.226-7.229; and Argentina – Textiles and Apparel, para. 6.15. ³⁴ DSU, Article 19.1 (emphasis added). See e.g. Appellate Body Report, US – Certain EC Products,

para. 81; and Panel Report, *EU – PET (Pakistan)*, para. 8.3. ³⁵ See e.g. Turkey's first written submission, paras. 49-76.

making findings on the consistency of the provisional safeguard with the covered agreements is necessary to resolve the dispute, and we therefore refrain from making such findings.

7.10. Notwithstanding this, we note that the provisional determination forms part of the record in the underlying investigation that led to the adoption of the definitive safeguard, and that the definitive determination refers to part of the findings and reasoning set out in the provisional determination. Therefore, although we decline to make findings on the consistency with the covered agreements of the provisional safeguard as a challenged measure, we still consider the provisional determination as appropriate when examining Turkey's claims against the definitive safeguard.

7.2 Product scope

7.11. Turkey claims that the definitive safeguard is inconsistent with Article XIX:1(a) of the GATT 1994 and Articles 2.1, 3.1, 4.1(c), 4.2(a), 4.2(b), and 4.2(c) of the Agreement on Safeguards, based on two sets of arguments.

7.12. First, Turkey argues that the European Commission applied 26 distinct safeguards, on 26 products, but did not examine whether the circumstances and conditions for imposing a safeguard existed for each of those products individually. We refer to this as the "mismatch" argument. According to Turkey, this gives rise to an inconsistency with Article XIX:1(a) of the GATT 1994 and Articles 2.1, 4.1(c), 4.2(a), and 4.2(b) of the Agreement on Safeguards.³⁶

7.13. Second, Turkey argues that the European Commission adopted an internally inconsistent approach to product scope at different stages of the investigation and application of the measures. We refer to this as the "internal inconsistency" argument. According to Turkey, this gives rise to an inconsistency with Article XIX:1(a) of the GATT 1994 and Articles 2.1, 3.1, 4.1(c), 4.2(a), 4.2(b), and 4.2(c) of the Agreement on Safeguards.³⁷

7.14. We note that within each of these sets of arguments, Turkey presents its arguments for the most part without distinguishing among the various claims it has raised. That is, Turkey sets out its arguments typically without specifying to what claim(s) each refers, and then cites a string of provisions in its concluding paragraphs; the Panel has sought some clarification in this regard but has obtained only very limited clarification.³⁸

7.15. To Turkey's arguments, the European Union responds that it adopted a single definitive safeguard on certain steel products comprising 26 product categories, and that it ascertained the existence of the circumstances and conditions necessary for imposing a safeguard on the product as thus defined.³⁹ As regards the allegation of internal inconsistency, the European Union responds that it did, as required, conduct an objective and unbiased evaluation and set out its findings and reasoned conclusions, and that the Agreement on Safeguards does not contain, in addition, the "internal consistency" requirement invoked by Turkey.⁴⁰

7.16. Among the third parties, Canada, the Republic of Korea, the United Kingdom, and the United States argue that the authority has discretion to define the product under investigation but must then verify whether the circumstances and conditions for applying a safeguard to that product are met.⁴¹ Argentina and Switzerland question how the authority could demonstrate the existence

³⁶ See e.g. Turkey's first written submission, para. 67. See, more broadly, Turkey's first written submission, paras. 50-67; and second written submission, paras. 17-46.

³⁷ See e.g. Turkey's first written submission, para. 76; response to Panel question No. 17, para. 3. See, more broadly, Turkey's first written submission, paras. 68-76; and second written submission, paras. 48-54.

³⁸ The second set of arguments is presented in support of the same claims as the first set of arguments, plus Articles 3.1 and 4.2(c) of the Agreement on Safeguards. (Turkey's response to Panel question No. 17, para. 3).

para. 3). ³⁹ European Union's first written submission, paras. 27-46; second written submission, paras. 1, 3-5, 7-16, and 24-29.

⁴⁰ European Union's first written submission, paras. 25-58; second written submission, paras. 1, 6, 17-23, and 28-29.

⁴¹ Canada's third-party submission, paras. 12-13 and 15; Korea's third-party submission, paras. 7-9; United Kingdom's third-party submission, paras. 4-11; Canada's third-party statement, para. 9; United Kingdom's third-party statement, paras. 4-6; United States' third-party responses to Panel question

Nos. 1 and 2, paras. 1-7.

of the conditions for imposing a safeguard for the product under investigation if it modified the product groupings depending on the condition it was examining.⁴² Switzerland adds that the authority must use a method that is unbiased and objective and that allows for a reasoned and adequate explanation of how the facts support the determination that the imports covered by a safeguard measure satisfy the conditions for applying the measure.⁴³ Switzerland also takes the view that there are contradictions both in the record of the domestic proceedings and in the EU arguments.⁴⁴ Japan recalls that if a safeguard is applied to specific products, it is not enough to demonstrate the existence of the conditions for imposing a safeguard for a broad category of products that includes the product under investigation.⁴⁵ Argentina considers that the question before the Panel is primarily a factual one, i.e. whether the European Commission applied a single safeguard on a single product, or a number of safeguards. Argentina notes that this question is central because the European Commission was required to verify the existence of the necessary conditions for the product on which it applied a safeguard. Further, Argentina notes that the European Commission was required to conduct the investigation in an objective and unbiased manner, and that if an authority relies on different product groupings at different steps of its analysis, that might cast doubt on whether the investigation was objective and unbiased.⁴⁶ For its part, Ukraine cautions that "including a wide list of products may result in artificial and incorrect findings in the framework of such categories' injury examination and, as a result, may lead to distorted findings of an investigation".47

7.17. Canada considers that the Panel must assess whether the European Commission demonstrated that it was appropriate to exclude certain product categories from the analysis.⁴⁸ The United Kingdom takes the view that we must examine whether the exclusion of certain product categories affected the analysis, and cautions against finding that a Member errs by adopting a narrower safeguard than it is entitled to apply.⁴⁹ Argentina indicates that a Member may exclude a subset of the product concerned from the application of a safeguard and, in that case, the imports of the excluded subset are a factor for which the authority must perform a non-attribution analysis pursuant to Article 4.2(b) of the Agreement on Safeguards.⁵⁰ Brazil and Switzerland take the view that a Member may exclude a subset of the product concerned from the application of a safeguard and, if it does so, it must exclude the corresponding imports from the analysis of increased imports.⁵¹

7.18. We recall that Turkey sets out its arguments on product scope mostly without specifying to what claim or claims each argument refers, and then cites a string of provisions in its concluding paragraphs.⁵² Therefore, in this section, we first examine the requirements of all of the provisions with which Turkey claims an inconsistency, and we then examine, in turn, each of Turkey's two sets of arguments.

7.2.1 The applicable requirements of Article XIX:1(a) of the GATT 1994 and Articles 2.1, 3.1, 4.1(c), 4.2(a), 4.2(b), and 4.2(c) of the Agreement on Safeguards

7.19. As noted, Turkey brings a number of claims related to the product scope of the EU investigation and of the ensuing safeguard. That is, Turkey claims that the definitive safeguard is inconsistent with Article XIX:1(a) of the GATT 1994, and Articles 2.1, 3.1, 4.1(c), 4.2(a), 4.2(b), and 4.2(c) of the Agreement on Safeguards. We consider each of these provisions below.

7.20. <u>Article XIX:1(a)</u> of the GATT 1994 provides:

⁴² Argentina's third-party statement, para. 7; Switzerland's third-party submission, para. 11.

⁴³ Switzerland's third-party submission, para. 7.

⁴⁴ Switzerland's third-party submission, para. 10.

 $^{^{45}}$ Japan's third-party submission, paras. 9 and 17 (referring to the Panel Reports and Appellate Body Report in US – Steel Safeguards).

⁴⁶ Argentina's third-party statement, paras. 5-7.

⁴⁷ Ukraine's third-party statement, paras. 11-12.

⁴⁸ Canada's third-party submission, para. 16; third-party statement, paras. 5-6.

⁴⁹ United Kingdom's third-party statement, paras. 9-10. See also United Kingdom's third-party response to Panel question No. 2, paras. 6-8.

⁵⁰ Argentina's third-party response to Panel question No. 2.

⁵¹ Brazil's third-party response to Panel question No. 2; Switzerland's third-party response to Panel

question No. 2.

⁵² See para. 7.14 above.

If, as a result of unforeseen developments and of the effect of the obligations incurred by a contracting party under this Agreement, including tariff concessions, *any product* is being imported into the territory of that contracting party in such increased quantities and under such conditions as to cause or threaten serious injury to domestic producers in that territory of like or directly competitive products, the contracting party shall be free, *in respect of such product*, and to the extent and for such time as may be necessary to prevent or remedy such injury, to suspend the obligation in whole or in part or to withdraw or modify the concession.⁵³

7.21. Article XIX:1(a) states that "if" the circumstances and conditions specified therein are met for "any product", a Member is free to apply a safeguard "in respect of such product". Thus, under Article XIX:1(a), a safeguard may be applied on a product only if the circumstances and conditions necessary to apply the safeguard have been verified for that same product.

7.22. Our understanding accords with the findings in US – Steel Safeguards and India – Iron and Steel Products that if a required circumstance or condition is not found to exist specifically for the product(s) under investigation, the application of a safeguard on that product is inconsistent with Article XIX:1(a). In those two proceedings, the panels found that the authority had not explained how the unforeseen developments, which related to steel in general, had resulted in an increase in imports of the specific products under investigation, and that therefore the safeguards applied on those products were inconsistent with Article XIX:1(a).

7.23. Similarly, <u>Article 2.1</u> of the Agreement on Safeguards provides that a Member may apply a safeguard "to a product" "only if" that Member has determined that certain conditions are met in respect of "such product". That is, the product on which a safeguard is applied must be the same as the product for which the Member has determined that the conditions for applying a safeguard are met. If a Member applies a safeguard on a product for which it has not determined that the conditions are met, it acts inconsistently with Article 2.1.

7.24. The last sentence of <u>Article 3.1</u> of the Agreement on Safeguards provides:

The competent authorities shall publish a report setting forth their findings and reasoned conclusions reached on all pertinent issues of fact and law.

7.25. Article 3.1 thus requires authorities to set forth their findings and "reasoned" conclusions on "all pertinent issues of fact and law" in their published reports.

7.26. The ordinary meaning of "finding" includes an instance of "coming across or discovering something ... as the result of searching or enquiry", "something which is found", and "the result of a[n] ... enquiry".⁵⁵ The ordinary meaning of "conclusion" includes a "final result, outcome" and "a judgement or statement arrived at by any reasoning process".⁵⁶ And the ordinary meaning of "reasoned" includes "characterized by or based on reasoning; carefully studied".⁵⁷ The last sentence of Article 3.1 therefore requires authorities to set forth in their published reports what they have found, as the result of searching or enquiry, and the conclusions (outcome, final result) they have reached from their findings. These conclusions must be "based on reasoning", "carefully studied".

7.27. As to the scope of the necessary findings and reasoned conclusions, the last sentence of Article 3.1 specifies that these must cover "all pertinent issues of fact and law". The ordinary

⁵³ Emphasis added.

⁵⁴ Panel Reports, *US* – *Steel Safeguards*, paras. 10.126, 10.147, and 10.150; *India* – *Iron and Steel Products* [appealed by India on 14 December 2018, appealed by Japan on 21 December 2018], paras. 7.111-7.112 and 7.122; and Appellate Body Report, *US* – *Steel Safeguards*, paras. 314-319.

⁵⁵ Oxford Dictionaries online, definition of "finding", n., meanings 2(a), 2(b), and 6(a),

https://www.oed.com/view/Entry/70356?rskey=Qnw27c&result=2&isAdvanced=false#eid (accessed 20 September 2021).

⁵⁶ Oxford Dictionaries online, definition of "conclusion", n., meanings 2 and 5(a), <u>https://www.oed.com/view/Entry/38312?redirectedFrom=conclusion#eid</u> (accessed 20 September 2021). ⁵⁷ Oxford Dictionaries online, definition of "reasoned", adj.2,

https://www.oed.com/view/Entry/159078?rskey=0Idv1P&result=3&isAdvanced=false#eid (accessed 20 September 2021).

meaning of "pertinent" includes "referring or relating to; relevant; to the point; apposite".⁵⁸ Article 3 is a provision on the "investigation" that is required for a Member to be able to apply a safeguard measure, and Article 3.1 specifies certain procedural requirements relating to such an investigation. This suggests that the "pertinent", i.e. relevant, issues of fact and law are those that must be established through the investigation, and specifically those that must be established for a Member to be able to apply a safeguard measure. We therefore agree with previous findings to the effect that the pertinent issues of fact and law within the meaning of Article 3.1 include the circumstances and conditions that must be demonstrated in order to apply a safeguard and that are set forth in Article 2.1 of the Agreement on Safeguards and Article XIX:1(a) of the GATT 1994.⁵⁹

7.28. <u>Article 4.1(c)</u> of the Agreement on Safeguards defines the domestic industry. One of the conditions that must be determined to exist to apply a safeguard is serious injury or threat of serious injury "to the domestic industry that produces like or directly competitive products".⁶⁰ Article 4.1(c) defines the domestic industry as "the producers as a whole of the like or directly competitive products operating within the territory of a Member, or those whose collective output of the like or directly competitive products". Thus, given a product under investigation, the authority must ascertain injury or threat of injury to the domestic industry producing the products that are "like or directly competitive" with that product.

7.29. We note that the panel in Argentina - Footwear (EC) was called upon to assess whether the authority had erred by not examining each injury factor for each segment of the market for the like or directly competitive products. In that dispute, Argentina had defined the like or directly competitive products as all footwear except ski boots. In collecting data from the domestic industry, Argentina had requested a breakdown into five product segments, but it had not then examined each injury factor for each of the five product segments separately, and the European Union argued that, by not doing so, Argentina had acted inconsistently with Articles 2 and 4 of the Agreement on Safeguards. The panel rejected that argument of the European Union. The panel reasoned that, given that the definition of like or directly competitive products (all footwear except ski boots) was not disputed, that definition was decisive. Therefore Argentina was "required at a minimum to consider each injury factor with respect to all footwear" and was not required to consider each injury factor for each segment of the like or competitive products on a disaggregated basis, although it could do so.⁶¹ We agree with the assessment of that panel, because a Member is required to verify the existence of the necessary circumstances and conditions for the product under investigation as a whole (in the injury determination, for the corresponding like or directly competitive products as a whole), and not necessarily on a disaggregated basis.

7.30. <u>Article 4.2(a)</u> of the Agreement on Safeguards provides that in determining whether increased imports have caused or threatened serious injury to the domestic industry, the authorities "shall evaluate all relevant factors of an objective and quantifiable nature having a bearing on the situation of that industry", and it also lists certain relevant factors.

7.31. <u>Article 4.2(b)</u> of the Agreement on Safeguards provides that to determine causation, the authorities must "demonstrate[], on the basis of objective evidence, the existence of a causal link between the increased imports of the product concerned and serious injury or threat thereof", and must not attribute to the increased imports injury that is due to other factors. Thus, it must be increased imports "of the product" under investigation that cause injury or threat of injury to the industry producing the like or directly competitive products.

7.32. Finally, <u>Article 4.2(c)</u> of the Agreement on Safeguards provides:

The competent authorities shall publish promptly, in accordance with the provisions of Article 3, a detailed analysis of the case under investigation as well as a demonstration of the relevance of the factors examined.

⁵⁸ Oxford Dictionaries online, definition of "pertinent", adj., meaning 3,

https://www.oed.com/view/Entry/141602?redirectedFrom=pertinent#eid (accessed 20 September 2021). ⁵⁹ See e.g. Appellate Body Report, *US – Lamb*, para. 72.

⁶⁰ Agreement on Safeguards, Article 2.1.

⁶¹ Panel Report, *Argentina – Footwear (EC)*, paras. 8.112-8.116 and 8.135-8.137.

7.33. Article 4.2(c) thus reiterates, with reference to the analysis of injury and causation addressed in Article 4.2, the obligation set out in the last sentence of Article 3.1, and requires in particular "a detailed analysis of the case" and a demonstration of the relevance of the factors examined.

7.34. It is also worth noting what the provisions considered in this section do *not* discipline, namely the choice of the product under investigation, in itself. These provisions do not set out rules for selecting the product under investigation, and for example they do not prevent a Member from including a range of products in a single investigation. We thus agree with the panel in *Dominican Republic – Safeguard Measures* that while the Agreement on Safeguards requires the authority to define the product under investigation "solely to those products that are like or directly competitive with each other".⁶² In that instance, the product under investigation comprised polypropylene bags (the end product) and polypropylene tubular fabric (the input product for polypropylene bags). The panel in that dispute explained that the authority could include "two separate products" in the product under investigation, and was not under an "obligation to provide an explanation of why two separate products were treated as the product under investigation in the same proceeding". It therefore found that no inconsistency had been established on that basis.⁶³

7.2.2 Whether Turkey has established that the definitive safeguard is inconsistent with Article XIX:1(a) of the GATT 1994 and Articles 2.1, 3.1, 4.1(c), 4.2(a), 4.2(b), and 4.2(c) of the Agreement on Safeguards with regard to product scope

7.35. We now assess whether Turkey has established that the definitive safeguard is inconsistent with the provisions outlined in the previous section.

7.36. We recall that Turkey has grouped its arguments into two sets, i.e. (a) that the European Commission applied 26 distinct definitive safeguards, on 26 products, but did not examine whether the circumstances and conditions for imposing a safeguard existed for each of those products individually (the mismatch argument); and (b) that the European Commission's approach to product scope in the investigation and application of the safeguards was internally inconsistent (the internal inconsistency argument). We address these two sets of arguments in turn.

7.37. Before discussing the parties' specific arguments and setting out our findings, we note a point of terminology. In the underlying investigation, the European Commission used the terms "products" and "product categories" somewhat interchangeably to refer to the range of products/product categories covered by its investigation. In their arguments before us, the European Union mostly refers to "product categories" and Turkey mostly refers to "products". As we explain below, we do not consider that the use of either of these phrases decides the outcome of Turkey's claims.⁶⁴ For consistency of style, we will use the phrase "product categories" to refer to the range of products/product categories covered by the EU investigation, except in certain instances when referring to arguments of a party or citing record documents that use the term "products" instead.

7.2.2.1 The alleged mismatch between the products on which the European Commission applied a safeguard and the products for which the European Commission investigated the existence of the necessary circumstances and conditions

7.38. Turkey argues that the European Commission applied 26 distinct definitive safeguards, on 26 products, without examining whether the circumstances and conditions for imposing a safeguard existed for each of those products individually.⁶⁵ Specifically, Turkey argues that: the European Commission has applied distinct safeguards on 26 products⁶⁶; the European Commission did not examine unforeseen developments and their connection with the increase in imports for each of the 26 products individually⁶⁷; although the European Commission examined the data on increase

⁶² Panel Report, *Dominican Republic – Safeguard Measures*, paras. 7.177 and 7.181.

⁶³ Panel Report, Dominican Republic – Safeguard Measures, paras. 7.176-7.182.

⁶⁴ See para. 7.53 below.

⁶⁵ Turkey's first written submission, paras. 37-38, 42-48, and 50-67; second written submission, paras. 17-46.

⁶⁶ Turkey's first written submission, paras. 42-48 and 54-58; second written submission, paras. 32-46; and opening statement at the first meeting of the Panel, para. 6.

⁶⁷ Turkey's first written submission, paras. 61 and 141-142.

in imports for each of the 26 products individually, it did not base its determination of an increase in imports on such data⁶⁸; and the European Commission did not examine threat of injury⁶⁹ and causation⁷⁰ for each of the 26 products individually. Therefore, according to Turkey, there is a mismatch between the products to which the European Commission has applied the safeguard, on the one hand, and the products for which it has investigated the existence of the circumstances and conditions required to apply the safeguard, on the other.

7.39. The European Union disputes Turkey's characterization of the facts underpinning this line of argument, namely that it has applied a number of distinct safeguards on distinct products. The European Union asserts that it investigated "certain steel products" globally as the product under investigation, and applied provisional and definitive safeguards on that product. The European Union asserts that its examinations at the level of individual product categories and then at the level of three "product families" merely supplemented its examination of the circumstances and conditions at the level of the product as a whole; that it carried out such additional examinations out of diligence; and that where it did carry out these additional examinations, they confirmed the findings reached for the product concerned globally.⁷¹

7.40. Turkey responds to the European Union's argument that the European Commission adopted a single safeguard by asserting that if the European Commission had really applied a single safeguard on a single product under investigation, that product should have stayed the same throughout the investigation. Turkey asserts that this was not the case here, given that the European Commission began with 28 products, applied a provisional safeguard on 23 products, and applied definitive measures on 26 products.⁷² The European Union responds that it chose not to apply measures to those product categories for which it had not observed an increase in imports, and that therefore when verifying the existence of the circumstances and conditions necessary to apply a safeguard, it excluded those product categories.⁷³

7.41. Turkey brings a single set of arguments on product scope in support of its series of claims under Article XIX:1(a) of the GATT 1994 and Articles 2.1, 4.1(c), 4.2(a), and 4.2(b) of the Agreement on Safeguards, without specifying which argument supports which of these individual claims. We consider, first, the claims of inconsistency under Article XIX:1(a) of the GATT 1994 and Article 2.1 of the Agreement on Safeguards (section 7.2.2.1.1) and, second, the claims of inconsistency under Articles 4.1(c), 4.2(a), and 4.2(b) of the Agreement on Safeguards (section 7.2.2.1.2).

7.2.2.1.1 Article XIX:1(a) of the GATT 1994 and Article 2.1 of the Agreement on Safeguards

7.42. We recall that under Article XIX:1(a) of the GATT 1994 and Article 2.1 of the Agreement on Safeguards, a Member may apply a safeguard to a product only if it has verified the existence of the required circumstances and conditions for that same product. In other words, the product(s) to which the Member applies a safeguard must be the same as the product(s) for which the Member has verified the existence of the circumstances and conditions for adopting the safeguard.

7.43. With this in mind, we turn to the facts to assess the merits of the parties' respective arguments.

7.44. In its notice of initiation, the European Commission described the "products subject to th[e] investigation" as "certain steel products (the 'products concerned') ... listed in Annex I to th[e] Notice". That annex listed 26 "products concerned". The notice used interchangeably the phrases "products" and "product categories". The notice specified that there was sufficient evidence to initiate

⁶⁸ Turkey's first written submission, paras. 62 and 170-173. In other words, Turkey's argument is that the determination was based on the data for the 26 product categories taken together, and on the data for the three product families. (Turkey's first written submission, para. 62 (referring to Definitive determination, (Exhibit TUR-5), recitals 32-36 and 47)).

⁶⁹ Turkey's first written submission, paras. 63, 226-229, and 254-257.

⁷⁰ Turkey's first written submission, para. 64.

⁷¹ European Union's first written submission, paras. 21-23; second written submission, paras. 11-15.

⁷² Turkey's opening statement at the first meeting of the Panel, para. 12.

⁷³ European Union's responses to Panel question No. 6, paras. 18-19; and No. 7, paras. 21-22; opening statement at the first meeting of the Panel, paras. 22-23. See also European Union's second written submission, para. 46.

- 23 -

the investigation both for the product concerned as a whole and for some or all product categories, and that the investigation would "examine the situation of the products concerned, including the situation of each of the product categories individually".⁷⁴ The European Commission later amended the notice of initiation, to add two more "steel product categories" to the scope of the investigation, and in so doing it updated among others the figures on total volume of imports, to include the additional product categories.⁷⁵

7.45. In its questionnaires both for exporting producers and for domestic producers, the European Commission described the product under investigation as "certain steel products", and asked respondents to provide certain data "separately for each of the products concerned".^{76, 77, 78}

7.46. In the provisional determination, the European Commission explained that it found:

[A]n important interrelation and strong competition between products classified in different product categories and also between products at different production stages within certain categories as some of the categories contain the main raw or input material to produce other products in other product categories.⁷⁹

7.47. The European Commission observed that, "as a consequence, given this level of interrelation, competitive pressure can easily be shifted from one product to the other"⁸⁰, and further explained that:

Because of these interrelations and interconnections, and given the fact that – as will be explained below – the potential trade diversion resulting from the US Section 232 measures applies to all product categories ... the analysis for the purpose of the provisional determination has been carried out both globally for all 28 product categories

In notes to the file following the original notice of initiation and its amendment, which were made available to interested parties, the European Commission referred to "products concerned" or "product categories", and provided data for the various product categories concerned taken together. In addition to the data for all product categories together, in some instances, the European Commission also provided or referred to data for individual product categories. (Note to the file (11 April 2018), (Exhibit TUR-18); Updated note to the file (2 July 2018), (Exhibit TUR-19). See also European Union's response to Panel question No. 3; second written submission, para. 14; and request for interim review, p. 1).

⁷⁶ Questionnaire for exporting producers, (Exhibit TUR-14), cover page and section D, heading D.1; Questionnaire for domestic producers, (Exhibit TUR-15), cover page and section E, heading E.1. The questionnaire for exporting producers requested information by product category on production, capacity, captive use, stock variation, sales volume, value and price by destination market (the respondent's domestic market, the European Union, the United States, and other markets), consumption in the respondent's domestic market, current and foreseen production and capacity on the respondent's domestic market (in the European Union and worldwide), future plans for production, capacity, and sales in the event that the European Union did, or did not, adopt a safeguard, and information on factors other than trends in imports that might have caused or threatened serious injury to the EU industry. (Questionnaire for exporting producers, (Exhibit TUR-14), section D). The questionnaire for exporting producers also requested information for all product categories taken together on sales volume and value by destination markets, and on the comparability of the imported and domestic products. (Questionnaire for exporting producers, (Exhibit TUR-14), sections C.7, C.9, and C.10). The questionnaire for domestic producers requested information by product category on the comparability of the imported and domestic products, on production, capacity, sales, stocks, profitability, return on capital employed, cash flow, employment, "any information" on the exporters' behaviour in the EU market and their current or likely future export capacity, and on the expected consequences of the US Section 232 measures. (Questionnaire for domestic producers, (Exhibit TUR-15), section E). The questionnaire for domestic producers also requested information for all product categories taken together on turnover by destination and on whether safeguard measures would be in the Union interest. (Questionnaire for domestic producers, (Exhibit TUR-15), sections D.1 and F).

⁷⁷ Turkey submitted a questionnaire response showing that a responding domestic producer provided data on injury factors at the level of product categories. (Questionnaire response of ThyssenKrupp Steel Europe GmbH, (Exhibit TUR-40); Turkey's second written submission, para. 26; and request for interim review, para. 8).

⁷⁸ The cover letter for the questionnaire for domestic producers indicated that the information requested therein was meant to "assist the Commission in establishing global data relating to the Union steel industry". (Cover letter (11 April 2018), (Exhibit EU-10), p. 1; European Union's request for interim review, p. 1).

⁷⁹ Provisional determination, (Exhibit TUR-3), recital 13.

⁸⁰ Provisional determination, (Exhibit TUR-3), recital 16.

⁷⁴ Notice of initiation, (Exhibit TUR-1), p. 29 and annex.

⁷⁵ Amendment to the notice of initiation, (Exhibit TUR-2), p. 54.

as the product concerned (i.e. steel in various shapes and forms) and also at individual level for each product category. 81

7.48. In its provisional determination, having reviewed import trends for all 28 product categories, the European Commission found that imports of five product categories had not increased between 2013 and 2017. On that basis, it decided to exclude them from the scope of the provisional safeguard, without prejudice to the possibility of imposing a definitive safeguard on them. The European Commission then performed every step in its analysis underlying the provisional safeguard for the remaining 23 product categories only, taken together and, for some steps, also individually.⁸²

7.49. In its definitive determination, the European Commission reiterated its approach to the product under investigation as comprising a group of product categories, taken together.⁸³ In response to certain objections from interested parties regarding the product scope of the investigation, it also divided the product categories concerned into three "product families", and examined both the import trends and injury also at the level of those product families.⁸⁴ Further, having found that imports of two of the 28 product categories had not increased between 2013 and the first half of 2018, it excluded those two product categories both from the scope of the definitive safeguard and from the scope of its analysis of the circumstances and conditions required to apply the safeguard.⁸⁵

7.50. Specifically, as a basis for adopting the definitive safeguard, the European Commission examined the existence of the required circumstances and conditions as follows:

- a. <u>Increase in imports</u>. The European Commission reviewed data on increase in imports for each of the 28 product categories; having found that there had been no increase in imports between 2013 and the first part of 2018 for two product categories, it excluded those from the rest of the analysis, and only included in the rest of its analysis the product categories for which it found an increase in imports; relying on the data for the remaining 26 product categories globally, it found that there had been an increase in imports; in addition, it also considered the evolution of imports for those 26 product categories grouped into three product "families", "to supplement the global ... analysis"⁸⁶;
- b. <u>Unforeseen developments</u>. The European Commission examined unforeseen developments for all product categories "globally", and found that there were such unforeseen developments resulting in increased imports^{87, 88};
- c. <u>Threat of injury</u>. The European Commission reviewed data for the 26 product categories globally, and on that basis found that the domestic industry was threatened with material injury; it also "supplemented" its conclusions by reviewing the data for the three families into which it grouped the 26 product categories⁸⁹;

⁸¹ Provisional determination, (Exhibit TUR-3), recital 17. (fn omitted)

⁸² Provisional determination, (Exhibit TUR-3), recitals 24 and 26 ff.

⁸³ Definitive determination, (Exhibit TUR-5), section 2.

⁸⁴ Definitive determination, (Exhibit TUR-5), recitals 19-22.

⁸⁵ Definitive determination, (Exhibit TUR-5), recital 31; see also para. 7.50 below.

⁸⁶ Definitive determination, (Exhibit TUR-5), recitals 27-47 and annex II.

⁸⁷ Definitive determination, (Exhibit TUR-5), recitals 50 and 48-62; Provisional determination, (Exhibit TUR-3), section V.

⁸⁸ We consider that the question whether the European Union only examined unforeseen developments for steel in general and, if so, whether that suffices to establish unforeseen developments resulting in an increase in imports of the product under investigation under Article XIX:1(a), is a separate one. However, since we find that the European Commission has otherwise not established that the unforeseen developments resulted in the observed increase in imports, we do not reach this question. (See section 7.3.2.3.2 below).

⁸⁹ Definitive determination, (Exhibit TUR-5), recitals 63-110. In the provisional determination, the European Union had also relied on data for product categories individually, but those did not include four of the product categories on which the European Union applied the definitive safeguard. The European Union did not rely on this analysis by product categories in the definitive determination. (Provisional determination, (Exhibit TUR-3), recitals 38, 47-69, and annex III; Definitive determination, (Exhibit TUR-5), recitals 63-110).

- d. <u>Causation</u>. The European Commission examined causation for all product categories globally.⁹⁰
- 7.51. The definitive safeguard consists of the following elements:
 - a. the introduction of tariff-rate quotas (TRQs) for each of the 26 product categories subject to measures⁹¹, with the following characteristics:
 - for each product category, the European Commission based the initial level of the TRQ on the average imports in the period 2015-2017 plus 5%⁹²;
 - ii. the European Commission further split two of the 26 product categories into two subcategories, adopting a TRQ for each subcategory⁹³; and
 - iii. for most product categories, the European Commission allocated part of the TRQ on a country-specific basis.⁹⁴
 - b. an out-of-quota duty rate of 25% for all product categories⁹⁵; and
 - c. for each of the 26 product categories separately, the exclusion of developing country Members with *de minimis* exports.⁹⁶

7.52. Having set out the facts, we now assess whether Turkey has established that the definitive safeguard is inconsistent with Article XIX:1(a) and Article 2.1 on the basis of a mismatch between the products to which the European Commission has applied the safeguard, on the one hand, and the products for which it has investigated the existence of the circumstances and conditions required to apply the safeguard, on the other.

7.53. We recall, first, that Article XIX:1(a) and the Agreement on Safeguards do not constrain the choice of product(s) under investigation.⁹⁷ They did not therefore bar the European Commission from choosing to investigate a number of steel products taken together, and indeed Turkey does not argue that the European Commission's selection of the product under investigation, in itself, gave rise to an inconsistency. We note in this respect that Turkey's reliance on the fact that the European Commission has at times used the term "products", in the plural, instead of referring to a "product" (comprising a number of product categories) is not decisive, because an investigating authority can investigate and apply a safeguard on a range of products taken together, provided that the safeguard meets the other requirements of Article XIX:1(a) and the Agreement on Safeguards.⁹⁸

7.54. Next, we recall that Article XIX:1(a) and Article 2.1 of the Agreement on Safeguards require a Member to have ascertained the existence of the necessary *circumstances and conditions* for

⁹⁰ Definitive determination, (Exhibit TUR-5), recitals 111-127; Provisional determination, (Exhibit TUR-3), recitals 70-81 (confirmed in Definitive determination, (Exhibit TUR-5), recital 127).

⁹¹ Definitive determination, (Exhibit TUR-5), article 1.

⁹² Definitive determination, (Exhibit TUR-5), recital 144.

⁹³ Definitive determination, (Exhibit TUR-5), recital 160 and annex IV.

⁹⁴ Definitive determination, (Exhibit TUR-5), article 1(2) and annex IV.

⁹⁵ Definitive determination, (Exhibit TUR-5), article 6.

⁹⁶ Definitive determination, (Exhibit TUR-5), article 5 and annex III.2.

⁹⁷ See para. 7.53 above.

⁹⁸ Separately, we note that Turkey has also argued that the use of the plural in "measures" (e.g. "definitive safeguard measures") establishes that the European Union in fact applied 26 distinct safeguards on 26 products. (Turkey's first written submission, para. 58; request for interim review, para. 9). We note that it is not uncommon to describe a safeguard as "safeguard measures", and that the European Commission's references to "safeguard measures" appeared in a context that made it abundantly clear that the European Commission considered that it had investigated and was applying measures on a "single product" collectively rather than distinct safeguards on a number of distinct products. (See e.g. Definitive determination, (Exhibit TUR-5), section 2.1). Therefore, the use of the plural "safeguard measures" by the European Commission in this context does not assist us in answering the question before us.

applying a safeguard for the *same product* on which that Member *applies a safeguard*.⁹⁹ We therefore consider whether Turkey has established that the European Commission did not do so.

7.55. We begin by assessing whether Turkey has established that the European Commission has applied 26 distinct safeguards on 26 products, rather than a single safeguard as the European Union asserts.

7.56. The facts reviewed above indicate that the European Commission began an investigation into "certain steel products", together, and indicated that it would "examine [their] situation" both globally and at the level of "the product categories individually"¹⁰⁰; it accordingly collected certain data globally and other data for each product category individually.¹⁰¹ At the end of its investigation, the European Commission applied, through the definitive determination (Regulation (EU) 2019/159), a safeguard consisting of a combination of a zero-rate TRQ and an out-of-quota duty rate of 25% on 26 categories of steel products. The size of the TRQ was calculated at the level of each product categories, at the level of further subsets of those product categories, and according to the definitive determination it was set, among other things, to "ensur[e] that traditional trade flows are maintained".¹⁰² Whenever a product category (or subcategory) subject to the safeguard is imported in excess of the respective TRQ, it is subject to the out-of-quota duty of 25%.¹⁰³

7.57. These facts support the European Union's assertion that it applied a single safeguard on 26 steel product categories, together.

7.58. Turkey argues that the fact that the European Commission examined import trends also at the level of individual product categories, and excluded from the application of the safeguard those product categories for which, separately, it found no increase in imports, and the fact that, as a result, the total number of product categories concerned shifted, establish that the European Union imposed a number of distinct safeguards, each on one product category.¹⁰⁴ However, a Member is not barred from not applying a safeguard on a subset of the product or products¹⁰⁵ on which it initiates a safeguard investigation. What a Member applying a safeguard must ensure is that it verifies the existence of the conditions for imposing a safeguard for the same product on which it applies the safeguard, which is a question we address further below.¹⁰⁶ We therefore do not consider that the decision to refrain from imposing the safeguard on certain product categories establishes that the European Commission imposed distinct safeguards on each of the remaining product categories.

7.59. Turkey also argues that the fact that TRQs were established at the level of the product categories shows that the European Union applied distinct safeguards on each product category.¹⁰⁷

¹⁰² Definitive determination, (Exhibit TUR-5), recital 144. See also European Union's opening statement at the first meeting of the Panel, para. 20.

⁹⁹ See paras. 7.20-7.23 above.

¹⁰⁰ See para. 7.44 above.

¹⁰¹ For example, the European Commission collected data on imports both globally and by product category; information on unforeseen developments globally; data on volume and value of sales both globally and by product category; information on Union interest globally; data on injury factors, and data on the exporters' current and expected capacity and production, by product category. (See e.g. Definitive determination, (Exhibit TUR-5), table 2 and section 4; and para. 7.45 above).

¹⁰³ See para. 7.51 above.

¹⁰⁴ Turkey's first written submission, para. 57; second written submission, paras. 32 and 44-45; and opening statement at the first meeting of the Panel, para. 12. The European Union argues that it adopted "a conservative approach" by choosing not to include in the safeguard those product categories for which, taken separately, it did not observe an import surge, even though it had found an import surge for all the product categories initially considered, taken together. The European Union observes that, having made this choice, it then verified the existence of the conditions for imposing a safeguard for the same product on which it applied the safeguard. (European Union's responses to Panel question No. 6, paras. 18-19; and No. 7, paras. 21-22; second written submission, para. 46; and opening statement at the first meeting of the Panel, paras. 22-23).

¹⁰⁵ See para. 7.53 above.

¹⁰⁶ See paras. 7.60-7.67 below.

¹⁰⁷ Turkey's first written submission, paras. 54-56; second written submission, paras. 32 and 34-38. The European Union explains that its safeguard "is calibrated so as to affect in the minimum way possible steel trade into the EU, reflecting historical trade flows in volumes and origins, making a more effective administration of the TRQs". (European Union's opening statement at the first meeting of the Panel, paras. 19-20).

- 27 -

However, we do not consider that the choice of calculating the size of the TRQ at the level of individual product categories (or further subdivisions¹⁰⁸), to better match traditional trade flows, establishes the existence of 26 distinct different safeguards, rather than a single safeguard covering 26 product categories. Turkey also refers to the fact that the European Commission excluded developing country Members with *de minimis* exports on a category-by-category basis, rather than for the 26 product categories as a whole¹⁰⁹, and points out that "the product" referred to in Articles 2.1 and 9.1 of the Agreement on Safeguards must be the same.¹¹⁰ According to the European Union, it applied Article 9.1 at the level of product categories to ensure "calibration", and that choice is not evidence of the existence of distinct safeguards for each product category.¹¹¹ We note that Turkey does not ask us to assess whether the European Commission's approach is consistent with Article 9.1 of the Agreement on Safeguards, and we therefore do not do so. Whether or not the exclusion of developing country Members at the level of subsets of the product concerned is consistent with Article 9.1, we do not view this approach as establishing that the European Commission in effect adopted 26 distinct safeguards. We therefore do not consider that Turkey has demonstrated that the European Commission adopted 26 distinct safeguards. On this factual question, we therefore conclude that the European Commission has applied a single definitive safeguard to 26 steel product categories, taken together.¹¹²

7.60. The next question under Article XIX:1(a) and Article 2.1, therefore, is whether the European Commission has investigated the existence of the circumstances and conditions required to apply a safeguard for that *same* product, namely the 26 steel product categories on which it has applied the challenged definitive safeguard, taken together.

7.61. The facts reviewed above¹¹³ indicate that, without prejudice to the substantive adequacy of the European Commission's examination on other grounds (which we consider in subsequent sections), the European Commission has examined, for the 26 steel product categories taken together, whether there had been an increase in imports; whether there existed unforeseen developments resulting in that increase in imports¹¹⁴; whether there existed injury or threat of injury; and whether the injury or threat of injury were caused by the increase in imports; and it found in each respect that this was the case.¹¹⁵

7.62. In assessing whether there had been an increase in imports, the European Commission also examined the trends in imports at the level of the product categories. First, the European Commission examined import trends for all 28 product categories under investigation. By so doing, it found that imports of 2 of the 28 product categories had not increased between 2013 and the first half of 2018; as a consequence, it chose both to remove those two product categories from its examination of the circumstances and conditions necessary to apply a safeguard, and not to apply a safeguard to them.¹¹⁶ Because the European Commission did not apply a safeguard to those two product categories, and because it excluded them from every step in its examination of the necessary circumstances and conditions for imposing a safeguard, we do not consider that

¹¹² As noted above, in this context, the wording "product categories" or "products" is not decisive, because a Member may adopt a safeguard on multiple product categories, provided that it complies with all the requirements of Article XIX:1(a) of the GATT 1994 and of the Agreement on Safeguards. (See para. 7.53 above).

¹¹³ See para. 7.50 above.

¹¹⁴ See fn 88 above.

¹¹⁵ See para. 7.50 above.

¹⁰⁸ See para.7.56 above.

¹⁰⁹ Turkey's second written submission, paras. 32 and 39-43.

 ¹¹⁰ Turkey's comments on the European Union's response to Panel question No. 19, para. 5; second written submission, para. 42.
 ¹¹¹ European Union's response to Panel question No. 19, paras. 5-6; opening statement at the

¹¹¹ European Union's response to Panel question No. 19, paras. 5-6; opening statement at the first meeting of the Panel, paras. 20-21. Among the third parties, the United States argues that: "[w]hen the products under investigation encompass different types of imported articles, it may be appropriate for a Member to impose a safeguard measure consisting of tariffs, quotas, or [TRQs] specific to each type of imported article or to different groupings of imported articles. The suitability of such a safeguard measure will depend upon the relevant facts". (United States' third-party response to Panel question No. 3, para. 11). The United Kingdom "does not consider that evidence relating to the manner in which a safeguard measure is applied, including developing country exceptions, is relevant to the factual determination of the product concerned adopted by the authority". (United Kingdom's third-party response to Panel question No. 3, para. 11).

¹¹⁶ Definitive determination, (Exhibit TUR-5), recital 31 and annex II.

Turkey has established that the removal of these two categories, in this case, gave rise to an inconsistency with Article XIX:1(a) and Article 2.1.117

7.63. The European Commission also examined the evolution of imports at the level of the three product families into which it grouped the 26 product categories to which it applied the safeguard, "to supplement the global import analysis".¹¹⁸

7.64. We do not consider that the fact that the European Commission also examined the evolution of imports at the level of product categories and product families detracts from the fact that the European Commission assessed the existence of the increase in imports for the 26 product categories taken together. An authority is not barred from examining the existence of the necessary conditions, including increase in imports, for segments of the product under investigation, provided that it does ascertain that the conditions exist for the product as a whole.¹¹⁹

7.65. Similarly, in assessing whether there was a threat of serious injury, the European Commission also performed its assessment at the level of the three product families into which it grouped the 26 product categories. Again, this does not detract from the fact that the European Commission assessed the existence of threat of injury for the 26 product categories taken together.

7.66. That is, the European Commission has verified the existence of each of the circumstances and conditions necessary for applying a safeguard, at least, for the 26 product categories taken together, i.e. the same product to which it has applied the definitive safeguard.

7.67. We therefore find that Turkey has not established that the European Union applied 26 distinct safequards on 26 products without examining the existence of the required circumstances and conditions for each of those products, inconsistently with Article XIX:1(a) and Article 2.1.

7.2.2.1.2 Articles 4.1(c), 4.2(a), and 4.2(b) of the Agreement on Safeguards

7.68. We recall that Turkey argues that by imposing 26 distinct safeguards on 26 products without examining the existence of the required circumstances and conditions for each of those products, the European Union also violated Articles 4.1(c), 4.2(a), and 4.2(b). Thus, the factual premise of these additional claims is the same as reviewed in the previous section (section 7.2.2.1.1). Because we have found that Turkey has not established that factual premise, we also find that Turkey has not established that the European Union applied 26 distinct safeguards on 26 products without examining the existence of the required circumstances and conditions for each of those products, inconsistently with Articles 4.1(c), 4.2(a), and 4.2(b) of the Agreement on Safeguards.

7.2.2.2 The alleged internal inconsistency in the approach to product scope

7.69. Turkey argues that the European Commission took an inconsistent approach to the products under investigation because it first gave notice of an investigation into 28 different products, and sought information by product¹²⁰, but then examined sometimes all products together, sometimes each product individually or by family, and sometimes took a combination of these approaches.¹²¹ Turkey adds that the European Commission also acted in a biased manner, because it first reviewed the import trends for each product and excluded the products whose imports had not increased, and then continued the analysis only including those products whose imports had increased.¹²² According

¹¹⁷ As noted in para. 7.73 below, we do not exclude that modifying the scope of a

safeguard investigation to remove product categories that do not, individually, meet certain conditions, could lead to a biased outcome in certain cases. However, we do not consider that Turkey has established that, in this case, the EU approach gave rise to a mismatch between the product to which the European Union applied the safeguard and the product for which it verified the existence of the necessary circumstances and conditions.

¹¹⁸ Definitive determination, (Exhibit TUR-5), recitals 33-47.

¹¹⁹ We also note in this regard that the panel in Argentina – Footwear (EC) addressed a similar question, noting that an authority could examine the existence of injury for segments of the domestic industry, although it was not required to. (See para. 7.29 above).

¹²⁰ Turkey's first written submission, paras. 42-46 and 70; second written submission, paras. 19-31; and opening statement at the first meeting of the Panel, paras. 8-11. ¹²¹ Turkey's first written submission, paras. 47-48 and 71-74; second written submission, para. 52; and

opening statement at the first meeting of the Panel, paras. 14-15. ¹²² Turkey's first written submission, para. 75; second written submission, para. 53.

to Turkey, for these reasons, the definitive safeguard is inconsistent with Article XIX:1(a) of the GATT 1994 and Articles 2.1, 3.1, 4.1(c), 4.2(a), 4.2(b), and 4.2(c) of the Agreement on Safeguards. Turkey specifies that the "consistency requirement" it relies upon "is supported by Article 4.2(a)", which according to Turkey establishes an "obligation of objectivity"¹²³, and by the requirements in Articles 3.1 and 4.2(c) to publish "reasoned conclusions" and "a detailed analysis", respectively.¹²⁴ As for the claims under Article XIX:1(a) and Articles 2.1, 4.1(c), and 4.2(b), Turkey has not explained in which way its arguments on "consistency" differ from its arguments on the alleged mismatch between the products subject to measures and the investigated products. We have examined and rejected the latter in the previous section, and therefore, in this section, we do not further consider the claims under Article XIX:1(a) and Articles 2.1, 4.1(c), and 4.2(b).

7.70. The European Union responds under Articles 3.1 that and 4.2(a), the authority's "evaluation ... must be objective and unbiased", and that Article 3.1(a) requires the authority to set forth "reasoned conclusions" in its report, and that this is what the European Commission has done. According to the European Union, Articles 3.1 and 4.2(a) do not prevent an authority from examining individual product categories or product families in addition to examining a product globally, as argued by Turkey.¹²⁵ Regarding the exclusion of product categories whose imports, individually, had not increased, the European Union responds that: the very purpose of a safeguard investigation is to investigate on what products a measure should be applied; even when a Member has a right to apply a safeguard, it does not have an obligation to do so^{126} ; the two categories that were excluded at the definitive stage represented only 0.36% of total imports, and "a very small amount of the total EU production", and therefore their "impact ... would not have changed the findings with regard to the injury"¹²⁷; and what matters is that the European Commission verified the existence of the necessary circumstances and conditions on the product comprising the 26 product categories to which it applied the definitive safeguard.¹²⁸

7.71. In essence, insofar as this second set of Turkey's arguments differs from the first set, discussed in the previous section, Turkey's contentions are that: (a) the European Commission examined the investigated products sometimes together, sometimes individually, and sometimes took a mixed approach; and (b) the European Commission's analysis was biased because it removed from the analysis those products whose imports had not increased.

7.72. As regards the first of these two contentions, we have found, $above^{129}$, that the European Commission examined the existence of each of the requisite circumstances and conditions for the 26 product categories to which it applied the challenged definitive safeguard, taken together, and that for some conditions it supplemented that analysis with an analysis at the level of product categories and/or product families. Given that it applied the definitive safeguard on a product comprising 26 product categories, taken together, the European Commission was required, at a minimum, to investigate the existence of the necessary circumstances and conditions for that product, additionally, at a more disaggregated level.¹³⁰ This being the case, Turkey has not demonstrated that merely by conducting both a global analysis and, in some instances, conducting in addition a more disaggregated analysis, the European Union acted inconsistently with the requirements to publish "findings and reasoned conclusions" (Article 3.1), to "evaluate all relevant factors of an objective and quantifiable nature" (Article 4.2(a)), or to publish "a detailed analysis" (Article 4.2(c)).

7.73. Turkey also argues that the decision to exclude (from the safeguard and from the underlying analysis) the two product categories whose imports had not increased between 2013 and June 2018 was biased, because, according to Turkey, it made it more likely to find an increase in imports and

¹²³ Turkey's first written submission, para. 40.

¹²⁴ Turkey's first written submission, para. 41.

¹²⁵ European Union's first written submission, paras. 47-57.

¹²⁶ European Union's second written submission, para. 6 (referring to Appellate Body Report, US - Line *Pipe*, paras. 83-84 ("[t]he Appellate Body confirmed in US - Line *Pipe* that the taking of a safeguard measure, when all the conditions are satisfied, is a right of WTO Members, not an obligation")).

 ¹²⁷ European Union's response to Panel question No. 7, para. 23; first written submission, para. 19.
 ¹²⁸ European Union's response to Panel question No. 18, paras. 1-4. See also Brazil's third-party response to Panel question No. 2(b), para. 10.

¹²⁹ See paras. 7.37-7.45 and 7.50-7.51 above.

¹³⁰ See para. 7.29 above.

threat of injury.¹³¹ We note that once the European Commission decided to exclude these product categories, it performed every step of its analysis without including them, and it did not include these product categories within the scope of the safeguard, either. Despite alleging bias, Turkey has not shown how the exclusion of certain product categories *both* from the scope of the measure and from the analytical steps of the investigation supporting the measure led to bias in the present case.¹³² Therefore, while we do not discount the possibility that modifying the scope of a safeguard investigation to remove product categories that do not, individually, meet certain conditions, could lead to a biased outcome in certain cases, we find that Turkey has not established that this was so in this instance.

7.74. We also observe that the European Commission explained its approach in the provisional and definitive determinations. In particular, the European Commission explained why it considered it appropriate to examine the product categories concerned taken together¹³³; it explained that it carried out certain additional analyses at the level of individual product categories or of product families to supplement the global analysis¹³⁴; and when it excluded certain product categories from the safeguard and consequently from the analysis of the required circumstances and conditions, it explained this choice.¹³⁵

7.75. Turkey argues that the European Commission failed to reach a reasoned and objective determination because its approach to the product under investigation lacked internal consistency and carried bias; however, we have found above that Turkey has not established that the European Commission was not entitled to complement the analysis for the product as a whole with a disaggregated analysis, or to exclude from both the analysis and from the scope of its safeguard those product categories for which there was no increase in imports, which are the grounds on which Turkey argues that the EU approach lacked consistency. Given that we have found that the European Commission's approach to the product under investigation was permissible, and noting that the European Commission has also explained that approach in some detail, we find that Turkey has not established that the definitive safeguard was inconsistent with Article XIX:1(a) of the GATT 1994 and Articles 2.1, 3.1, 4.1(c), 4.2(a), 4.2(b), and 4.2(c) of the Agreement on inconsistencies Safequards because of the alleged internal and bias in the European Commission's approach to the product under investigation.

7.3 Unforeseen developments

7.76. Turkey argues that the definitive safeguard is inconsistent with Article XIX:1(a) of the GATT 1994 because: (a) the European Commission did not identify unforeseen developments; (b) if it did identify developments, they were not unforeseen; and (c) it did not demonstrate that the injurious increase in imports occurred "as a result of" the unforeseen developments.¹³⁶ The European Union disagrees on all counts.

7.77. Below, we first consider the applicable requirements of Article XIX:1(a) (section 7.3.1). We then examine whether Turkey has established that the European Union acted inconsistently with this

¹³¹ See e.g. Turkey's first written submission, para. 75.

¹³² Turkey has suggested a parallel between the facts in this instance and zeroing in anti-dumping investigations. However, in this instance the product categories excluded from the investigation of the circumstances and conditions necessary for imposing a safeguard are also excluded from the application of the safeguard, and there is no calculation of the equivalent of a dumping margin. (Turkey's comments on the European Union's response to Panel question No. 50, para. 62; European Union's opening statement at the second meeting of the Panel, paras. 53-55).

¹³³ See paras. 7.46-7.47 above. See also Provisional determination, (Exhibit TUR-3), recitals 11-17; and Definitive determination, (Exhibit TUR-5), recitals 12-18. In response to a question from us, Turkey indicated that this explanation was "irrelevant in the context of Turkey's claims", although "in any event" Turkey did not agree with the European Commission's explanation that there were an important interrelation and strong competition between product categories. (Turkey's response to Panel question No. 8, paras. 27-28; request for interim review, para. 13).

¹³⁴ Provisional determination, (Exhibit TUR-3), recital 47; Definitive determination, (Exhibit TUR-5), recitals 19-22, 28, and 34.

¹³⁵ See paras. 7.48-7.49 above. See also Provisional determination, (Exhibit TUR-3), recital 24, article 1.1, and annex II; and Definitive determination, (Exhibit TUR-5), recital 31, article 1.1, and annex II.

¹³⁶ Turkey's first written submission, paras. 87-112 and 123-150; second written submission, paras. 59-76 and 80-95. Turkey also argues that the European Commission "should have made its analysis of the unforeseen developments for each product category". (Turkey's first written submission, para. 141). We have addressed this argument in section 7.2 above.

provision because the European Commission did not identify unforeseen developments (section 7.3.2.1), because the developments were not unforeseen (section 7.3.2.2), or because the European Commission did not establish that the injurious increase in imports was "as a result of" the unforeseen developments (section 7.3.2.3).

7.3.1 The applicable requirements of Article XIX:1(a) of the GATT 1994

7.78. Article XIX:1(a) of the GATT 1994 provides:

If, as a result of unforeseen developments and of the effect of the obligations incurred by a contracting party under this Agreement, including tariff concessions, any product is being imported into the territory of that contracting party in such increased quantities and under such conditions as to cause or threaten serious injury to domestic producers in that territory of like or directly competitive products, the contracting party shall be free, in respect of such product, and to the extent and for such time as may be necessary to prevent or remedy such injury, to suspend the obligation in whole or in part or to withdraw or modify the concession.137

7.79. Article XIX:1(a) is a two-part conditional sentence, and each part contains several clauses. The first part of Article XIX:1(a) begins with the conjunction "[i]f" and recites certain circumstances or conditions. The second part of Article XIX:1(a), which begins with the phrase "the contracting party shall be free", describes what happens in the event that the circumstances and conditions contained in the first part of the sentence are satisfied. That is, the Member concerned may ("shall be free ... to") suspend the relevant obligation(s) of the GATT 1994, thus imposing a safeguard.

7.80. Part of the circumstances and conditions that permit a Member to apply a safeguard is that "as a result of unforeseen developments ... [a] product is being imported into the territory of that [Member] in such increased quantities and under such conditions as to cause or threaten serious injury". We first consider the meaning of the phrase "unforeseen developments", and then the meaning of the phrase "as a result of".

7.81. The ordinary meaning of "development" includes "an event constituting a new stage in a changing or evolving situation; a fact or circumstance emerging or coming to light".¹³⁸ The ordinary meaning of "unforeseen" includes "not anticipated or expected: ... unexpected"¹³⁹, and not "seen beforehand".¹⁴⁰ Thus, an unforeseen development is an event, fact, or circumstance that emerges or comes to light, including a new stage in an evolving situation, that was not anticipated or expected. This interpretation accords with findings in previous disputes that "unforeseen developments" are developments that were "unexpected".¹⁴¹ Because the ordinary meaning of the word "development" includes a "new stage in a changing or evolving situation", a known event may develop into an unforeseen development.¹⁴²

7.82. Regarding the point in time at which the development must be "unforeseen", we note that Article XIX:1(a) requires that the injurious increase in imports result from two elements, i.e. unforeseen developments and "the effect of the obligations incurred" under the GATT 1994. We further note that the remedy allowed by Article XIX:1(a) is the suspension of the relevant obligations. To us, this means that the point in time at which a development must have been "unforeseen" is when the relevant obligations were incurred, because the Member concerned would not have undertaken the obligation had it foreseen a development that would result in an injurious increase in imports. This understanding also accords with findings reached in previous proceedings,

webster.com/dictionary/unforeseen (accessed 22 September 2021).

¹³⁷ Emphasis added.

¹³⁸ Oxford Dictionaries online, definition of "development", n., meaning II.11

https://www.oed.com/view/Entry/51434?redirectedFrom=development#eid (accessed 22 September 2021). ¹³⁹ Merriam Webster, online edition, definition of "unforeseen", adj., https://www.merriam-

¹⁴⁰ Oxford Dictionaries online, definition of "foreseen", adj., https://www.oed.com/view/Entry/73150?rskey=ykyd38&result=2&isAdvanced=false#eid (accessed 22 September 2021).

¹⁴¹ Appellate Body Reports, Korea – Dairy, para. 84; Argentina – Footwear (EC), para. 91.

¹⁴² See also Panel Reports, US - Steel Safequards, para. 10.84 ("there may be instances where an event which is already known will develop into a situation initially unforeseen").

namely that the point in time at which a development must be unforeseen is when the relevant obligations were incurred. $^{\rm 143}$

7.83. Whether a given development was unforeseen at the time the obligations were incurred will of course depend on the facts. We note that, in the past, panels have found that certain market developments, in the circumstances of the cases before them, could constitute unforeseen developments. In US - Steel Safeguards, for example, the panel found that the Russian financial crisis, the Asian financial crisis, the continued strength of the US market, and the persistent appreciation of the dollar, taken together, could constitute unforeseen developments.¹⁴⁴ As regards specifically the Russian financial crisis, the panel considered arguments to the effect that the crisis could not be unforeseen during the Uruguay Round, because the Union of Soviet Socialist Republics (USSR) had dissolved prior to the conclusion of the Uruguay Round and Russia's then-current financial circumstances could be traced to that dissolution. In considering this argument, the panel observed that the United States had distinguished between the anticipated financial difficulties flowing from the dissolution of the USSR and the additional financial disruption that was unforeseen, and stated that "an unforeseen development may evolve from well-known prior facts".¹⁴⁵ In India – Iron and Steel Products, the panel reviewed India's conclusion that a significant increase in steel excess capacity, decreased demand for steel in important markets, currency depreciation in Russia and Ukraine, and increased demand and prices in India, were unforeseen developments. In doing so, the panel reasoned that while "changes in production capacity or demand are not necessarily extraordinary circumstances, and can occur as part of normal business cycles, the extent and timing of such changes as well as the degree of their impact on the competitive situation in the market can be unforeseen". The panel found that having explained the extent of the developments and having noted that they were occurring at the same time, India had reasonably concluded that they constituted unforeseen developments.¹⁴⁶

7.84. Turning to the phrase "as a result of" we note that the ordinary meaning of the word "result" includes "[t]he effect, consequence, or outcome of some action, process, or design".¹⁴⁷ In Article XIX:1(a) the phrase "as a result of" connects the unforeseen developments and the injurious increase in imports, i.e. it envisages that the injurious increase in imports occurs "as a result of" – i.e. is the effect, consequence or outcome of – the unforeseen developments. We note that previous reports have described this as a "logical connection" between the unforeseen developments and the injurious increase in imports.¹⁴⁸ The phrase "logical connection" does not appear in the text of Article XIX:1(a). However, our understanding of this provision accords with this concept.¹⁴⁹ We also note that in describing the relationship required between the increased imports and the injury suffered by the domestic industry, Article XIX:1(a) uses the word "cause" in the phrase "cause or threaten serious injury". In contrast, the word "cause" is not used in the first part of Article XIX:1(a), which uses instead the phrase "as a result of". The different choice of terms indicates that the

¹⁴³ See e.g. Appellate Body Report, *Korea – Dairy*, para. 86; Panel Reports, *US – Steel Safeguards*, paras. 10.79, 10.80, and 10.85; and Working Party Report, *US – Fur Felt Hats*, GATT/CP/106, adopted 22 October 1951, para. 12.

¹⁴⁴ Panel Reports, US – Steel Safeguards, paras. 10.78-10.100.

¹⁴⁵ Panel Reports, *US – Steel Safeguards*, para. 10.84. The panel in that case therefore accepted for the sake of argument that the Russian financial crisis could constitute an unforeseen development.

⁽Ibid. para. 10.85 ("the Panel will accept, arguendo")).

¹⁴⁶ Panel Report, *India – Iron and Steel Products* [appealed by India on 14 December 2018, appealed by Japan on 21 December 2018], paras. 7.90-7.99.

¹⁴⁷ Oxford Dictionaries online, definition of "result", n., meaning II.4.a <u>https://www.oed.com/view/Entry/164061?rskey=Ft4Kjb&result=1&isAdvanced=false#eid</u> (accessed 22 September 2021).

¹⁴⁸ Appellate Body Reports, *Korea – Dairy*, para. 85; *Argentina – Footwear (EC)*, para. 92; *US – Lamb*, para. 72; and *US – Steel Safeguards*, paras. 317-318 and 322; and Panel Reports, *US – Steel Safeguards*, para. 10.104; and *Argentina – Preserved Peaches*, paras. 7.17 and 7.24. See also, with different wording, Panel Report, *India – Iron and Steel Products* [appealed by India on 14 December 2018, appealed by Japan on 21 December 2018], paras. 7.105-7.106, 7.110, and 7.114. The report of the GATT Working Party in *US – Fur Felt Hats* described the requirement as being that "the increased imports must be the result of unforeseen developments". (Working Party Report, *US – Fur Felt Hats*, GATT/CP/106, para. 4(a)(ii); see also ibid. paras. 7 and 11-12).

¹⁴⁹ Appellate Body Reports, *Korea – Dairy*, para. 85; *US – Steel Safeguards*, paras. 317-318. See also Panel Reports, *US – Steel Safeguards*, para. 10.104; and *India – Iron and Steel Products* [appealed by India on 14 December 2018, appealed by Japan on 21 December 2018], para. 7.86.

relationship required between unforeseen developments and the increase in imports is not the same as causation, which is instead required between the increase in imports and serious injury.

7.85. In US - Steel Safeguards, the panel reasoned that the "nature of the facts, including their complexity, will dictate the extent to which the relationship between the unforeseen developments and increased imports causing injury needs to be explained".¹⁵⁰ We agree that what is required to establish the connection between the unforeseen developments and the increase in imports will depend on the specific facts of each case. As the panel observed in US - Steel Safequards, ' '[i]n some cases, the explanation may be as simple as bringing two sets of facts together", whereas "in other situations, it may require much more detailed analysis".¹⁵¹

7.86. We now assess whether Turkey has demonstrated that the European Commission did not establish the existence of unforeseen developments that resulted in the injurious increase in imports.

7.3.2 Whether Turkey has established that the definitive safeguard is inconsistent with Article XIX:1(a) of the GATT 1994 with regard to unforeseen developments

7.87. To recall, Turkey claims that the definitive safeguard is inconsistent with Article XIX:1(a) of the GATT 1994 because the European Commission adopted that measure without establishing the existence of unforeseen developments that resulted in the injurious increase in imports. Turkey advances three sets of arguments in support of this claim: first, that the European Commission did not even nominally identify unforeseen developments; second, that if the European Commission did identify certain developments, these developments were not "unforeseen"; and third, that the European Commission did not establish that the unforeseen developments resulted in the increase in imports of the product under investigation into the European Union. We examine each of these three sets of arguments in turn, below.

7.3.2.1 Whether the European Commission did not identify any unforeseen developments

7.88. Turkey notes that the European Commission's provisional and definitive determinations stated that the increase in imports "had been the result of unforeseen developments that found their source in a number of factors", and then went on to describe the factors in question.¹⁵² Turkey argues that the fact that the European Commission described the factors that were allegedly the source of unforeseen developments, means that the European Commission did not identify the unforeseen developments themselves, which must be different from their "source".¹⁵³ Turkey adds that the "failure to clearly identify the events ... is compounded by the failure to clearly state when ... the factors identified would have occurred", with the European Commission making imprecise references to different time periods, one of which spanned 16 years.¹⁵⁴

7.89. The European Union asserts that Turkey's arguments are mere "semantic efforts", and explains that the developments that the European Commission described as the "source" are indeed the unforeseen developments.¹⁵⁵ Regarding timing, the European Union responds that the provisional and definitive determinations do identify the timeframe of each development, and that not all events that form a confluence of developments must take place simultaneously.¹⁵⁶

7.90. Among the third parties, the United States argues that an authority is not required to demonstrate in its published report the existence of unforeseen developments before it applies a safeguard; the developments need only exist as a matter of fact.¹⁵⁷ Japan, Switzerland, and Ukraine argue that an authority must identify the unforeseen developments in its published report before

 ¹⁵⁰ Panel Reports, US - Steel Safeguards, para. 10.115.
 ¹⁵¹ Panel Reports, US - Steel Safeguards, para. 10.115.

¹⁵² Definitive determination, (Exhibit TUR-5), recital 48 (emphasis added). See also Provisional determination, (Exhibit TUR-3), recital 30.

¹⁵³ Turkey's first written submission, paras. 88-89; second written submission, para. 59.

¹⁵⁴ Turkey's first written submission, para. 90; second written submission, paras. 60-61.

¹⁵⁵ European Union's first written submission, paras. 72-74.

¹⁵⁶ European Union's first written submission, paras. 75-79 (referring to Panel Report, India – Iron and Steel Products [appealed by India on 14 December 2018, appealed by Japan on 21 December 2018], para. 7.114).

¹⁵⁷ United States' third-party submission, paras. 8-14; third-party statement, paras. 7-10.

applying a safeguard.¹⁵⁸ Concerning Turkey's reliance on the fact that the European Commission referred to "factors", Switzerland argues that the particular label used by the authority to describe the developments does not matter.¹⁵⁹

7.91. We turn for our assessment to the provisional and definitive determinations.

7.92. The provisional¹⁶⁰ and definitive determinations each contain a section entitled "Unforeseen Developments".¹⁶¹ In those sections, both determinations identify three cumulative developments, namely: (a) "unprecedented" overcapacity in the steel sector, which persisted despite measures taken to narrow it¹⁶², and which was fuelled by government support¹⁶³; (b) greater use of trade-restrictive measures in third-country markets, including applied tariff increases, minimum import prices, mandatory national standards, and local content requirements, and a steady increase in trade defence measures on steel, a significant portion of which was by the United States, one of the world's largest steel importers¹⁶⁴; and (c) the United States' Section 232 measures.¹⁶⁵

7.93. Turkey grounds its claim that the European Commission did not identify any unforeseen development on the fact that both determinations refer to unforeseen developments that "found their source in a number of factors"¹⁶⁶, and then list the developments set out above. Turkey points to this statement to reason that if these developments are "factors" that were the "source" of the unforeseen developments, they could not be the unforeseen developments themselves, and therefore that the European Commission cannot be considered to have even nominally identified any unforeseen developments. We disagree. The developments in guestion are listed and discussed in a section clearly entitled "unforeseen developments"¹⁶⁷; they are referred to interchangeably not only as "factors" but also as "unforeseen developments"¹⁶⁸; the European Commission addressed arguments of the interested parties tending to establish that the developments were not unforeseen, explaining why, in its view, they were indeed unforeseen¹⁶⁹; and the European Commission indicated that these developments led to an increase in imports.¹⁷⁰ It is thus clear from the European Commission's published determinations that it considered these to be the relevant unforeseen developments. In this context, that the European Commission referred at times to the unforeseen developments as "factors" that were a "source" of unforeseen developments does not negate the fact that it identified the developments at issue as unforeseen developments. A Member is not necessarily required to use the exact wording of the World Trade Organization (WTO) Agreement in its domestic determinations.

7.94. On Turkey's argument that the European Commission did not "clearly state when ... the factors identified ... occurred"¹⁷¹, we observe that the provisional and definitive determinations indicated that: (a) global steelmaking capacity had more than doubled from 2000 to 2017 and global

¹⁵⁸ Japan's third-party submission, para. 6; Switzerland's third-party submission, para. 13; and Ukraine's third-party submission, para. 11.

¹⁵⁹ Switzerland's third-party submission, para. 17.

¹⁶⁰ The definitive determination incorporates by reference the findings on unforeseen developments of recitals 30 to 36 of the provisional determination, which thus also form part of the basis for the definitive safeguard. (Definitive determination, (Exhibit TUR-5), recital 48; see also para. 7.10 above).

¹⁶¹ Provisional determination, (Exhibit TUR-3), section V; Definitive determination, (Exhibit TUR-5),

section 4. ¹⁶² Provisional determination, (Exhibit TUR-3), recital 31; Definitive determination, (Exhibit TUR-5), recitals 48-49 and 62.

¹⁶³ Provisional determination, (Exhibit TUR-3), recital 32; Definitive determination, (Exhibit TUR-5), recitals 48-49 and 62.

¹⁶⁴ Provisional determination, (Exhibit TUR-3), recitals 33-34; Definitive determination, (Exhibit TUR-5), recitals 48-49 and 62.

¹⁶⁵ Provisional determination, (Exhibit TUR-3), recital 35; Definitive determination, (Exhibit TUR-5), recitals 48-49 and 62.

¹⁶⁶ Definitive determination, (Exhibit TUR-5), recital 48; Provisional determination, (Exhibit TUR-3), recital 30 ("that finds its source in").

¹⁶⁷ Provisional determination, (Exhibit TUR-3), section V, heading; Definitive determination,

⁽Exhibit TUR-5), section 4, heading.

¹⁶⁸ See e.g. Provisional determination, (Exhibit TUR-3), recital 36 ("the abovementioned unforeseen developments").

¹⁶⁹ Definitive determination, (Exhibit TUR-5), recitals 51-52.

¹⁷⁰ Provisional determination, (Exhibit TUR-3), recital 36; Definitive determination, (Exhibit TUR-5),

recital 62. ¹⁷¹ See para. 7.88 above.

steel production in 2016 was 100 million tonnes higher than global steel demand¹⁷²; (b) several countries had made greater use of trade restrictive measures on steel products since 2014/2015 and throughout 2017¹⁷³; (c) there had been significantly more initiations of steel-related trade defence investigations in 2015-2016 than in 2011-2013, and in 2018 the United States had 169 anti-dumping and countervailing (AD/CV) duty orders in place on steel and 25 ongoing investigations¹⁷⁴; and (d) the US Section 232 measures on steel had been adopted in March 2018, following an investigation initiated in April 2017 and a report published in January 2018.¹⁷⁵ Thus, the European Commission described the timing of each development either by reference to a timeframe over which it had manifested itself (e.g. the doubling of capacity since 2000), or by reference to the specific time at which it was introduced (e.g. the initiation of a US Section 232 investigation in April 2017 and introduction of measures in March 2018). Therefore, we do not see that Turkey has substantiated its allegation that the European Commission failed to identify unforeseen developments because it did not sufficiently identify their timing.

7.95. Given this, we reject Turkey's arguments that the European Commission did not *identify* the unforeseen developments on which the definitive safeguard is based.

7.3.2.2 Whether the developments were not unforeseen

7.96. Turkey argues that even if the European Commission had identified developments, none of them was unforeseen. We consider whether Turkey has so established for each of the three sets of developments identified by the European Commission (the increase in steel overcapacity, the increase in trade restrictive and trade defence measures on steel, and the US Section 232 measures on steel), in turn.

7.97. First, Turkey submits that steel overcapacity existed for decades and predated the Uruguay Round, and therefore was not unforeseen.¹⁷⁶

7.98. The European Union responds that the European Commission explained why the overcapacity in question was unforeseen: according to the European Commission's explanations, this was because steel production capacity continued to increase after 2011 (following a dip between 2009 and 2011) despite being already excessive then and despite the fact that it was economically expected to decline, and because this overcapacity persisted "despite the important number of measures taken to reduce it".¹⁷⁷ The European Union also observes that the panel in India – Iron and Steel Products considered that the increase in production capacity for steel in 2015 was, together with other factors, an unforeseen development.¹⁷⁸

7.99. Turkey responds to the European Union that if an authority relies on developments in the market such as changes in capacity as an unforeseen development, it must provide "a reasonable and detailed explanation regarding exactly how ... [they] are unforeseen". According to Turkey, "[t]he European Commission's findings and explanations that '[post-2011] it was expected that total crude steel capacity would decrease or at least remain stable', do not constitute such a reasonable and detailed explanation, as this statement is provided without evidence." Turkey also asserts that the dip in overcapacity between 2009 and 2011 was due to the financial crisis, and therefore that it

¹⁷² Provisional determination, (Exhibit TUR-3), recital 31. In addition, in the definitive determination the European Commission discussed intervening trends (comparing 2009-2011 with 2011-2016) to address arguments to the effect that excess steelmaking capacity was not unexpected. (Definitive determination, (Exhibit TUR-5), recital 52).

¹⁷³ Provisional determination, (Exhibit TUR-3), recital 33.

¹⁷⁴ Provisional determination, (Exhibit TUR-3), recital 34; Definitive determination, (Exhibit TUR-5), recital 56.

¹⁷⁵ Definitive determination, (Exhibit TUR-5), recitals 49 and 58-59.

¹⁷⁶ Turkey's first written submission, paras. 93-96; see also response to Panel question No. 9, on the issue of the treatment of one piece of evidence. Turkey argues in particular that because overcapacity has existed for decades and predated the Uruguay Round, "[p]eriods of overcapacity in commodities ... cannot be characterized as an unexpected development that was unforeseen by the European Union during the Uruguay Round." (Turkey's first written submission, para. 94; request for interim review, para. 15). ¹⁷⁷ European Union's first written submission, paras. 82-87; Definitive determination, (Exhibit TUR-5),

recitals 49 and 52. See also European Union's response to Panel question No. 9, para. 30. ¹⁷⁸ European Union's first written submission, para. 85.

was to be expected that capacity would then have resumed its upward trend observed in the previous years. $^{\rm 179}$

7.100. Among the third parties, Canada submits that overcapacity may be an unforeseen development but that an authority must provide a reasoned and adequate explanation of why this is so.¹⁸⁰ Brazil submits that if a Member invokes market dynamics, such as overcapacity, as an unforeseen development, it must provide a careful explanation of why they were unforeseen.¹⁸¹ The Republic of Korea and Ukraine submit that overcapacity has existed in the steel sector for a long time, and the Republic of Korea therefore considers it "[i]n principle ... doubtful" that it could constitute an unforeseen development.¹⁸² More generally, on the interpretation of "unforeseen", Japan submits that an authority must have, "at a minimum, some discussion as to why the alleged 'unforeseen developments' were 'unexpected'", and the United States argues that unforeseen means not actually foreseen.¹⁸³

7.101. We turn again for our assessment to the published determinations. With respect to overcapacity, we understand the explanation in the published determinations to be that what was unexpected was that overcapacity would continue to increase, reaching "unprecedented" levels, contrary to economic logic and efforts to contain the increase.¹⁸⁴ Turkey relies on the fact that overcapacity in the steel sector has existed for a long time and argues that therefore periods of overcapacity were not unforeseen when concluding the Uruguay Round.¹⁸⁵ However, Turkey has not directly contested the European Commission's assertions that overcapacity had continued to increase and reached unprecedented levels, with capacity having more than doubled since 2000, despite measures to contain overcapacity. Turkey has contested the relevance of the 2009-2011 dip in overcapacity in the assessment conducted by the European Commission, and has noted that the European Commission's statement that capacity would have been expected to decrease or remain stable after 2011 did not contain a reference to supporting material on the record.¹⁸⁶ This however does not contradict the proposition relied on by the European Commission that it was contrary to economic logic for capacity to continue to increase when there was already considerable overcapacity. The European Commission drew a distinction between an underlying situation - i.e. overcapacity per se, which was known at the time of the Uruguay Round - and its development into a new situation, i.e. the fact that overcapacity greatly increased, contrary to economic expectations and measures taken to address it. We do not consider that more was required in this case to identify the unprecedented overcapacity as a development that was unforeseen in this instance. A development that was unforeseen at the time of contracting an obligation can evolve from a situation that was known at that time. We therefore do not consider that Turkey has established that the European Commission erred in concluding that increased overcapacity constituted a development that was unforeseen at the conclusion of the Uruguay Round.

7.102. Second, Turkey argues that the increase in trade defence measures on steel products could not be unforeseen, because: "the adoption of trade defence measures is contemplated by WTO rules"¹⁸⁷; periodic increases in the use of anti-dumping measures had taken place before the Uruguay Round¹⁸⁸; data on the use of anti-dumping measures since 1995 also show that there are periodic peaks in the number of anti-dumping investigations, both generally and for "base metals and articles"¹⁸⁹; and the European Union itself contributed to the alleged increase both by adopting

¹⁷⁹ Turkey's first written submission, paras. 94-96; second written submission, paras. 65-67.

¹⁸⁰ Canada's third-party submission, para. 23.

¹⁸¹ Brazil's third-party submission, paras. 5-6.

¹⁸² Korea's third-party submission, para. 17; Ukraine's third-party submission, para. 15; and third-party statement, para. 7.

¹⁸³ Japan's third-party submission, para. 7; United States' third-party submission, para. 5; and third-party statement, paras. 4-6.

¹⁸⁴ Provisional determination, (Exhibit TUR-3), recital 31; Definitive determination, (Exhibit TUR-5), recitals 49 and 52.

¹⁸⁵ See para. 7.97 and fn 176 above.

¹⁸⁶ See para. 7.99 above.

¹⁸⁷ Turkey's first written submission, para. 100.

¹⁸⁸ In this regard, Turkey relies on data concerning all anti-dumping investigations, regardless of the product concerned. (Turkey's first written submission, paras. 101-102 and 106, and figure 2).

¹⁸⁹ Turkey's first written submission, paras. 102-104 and (for iron and steel products in the United States) 108, and figures 2-4.

trade defence measures and because its steel exports were also the target of trade defence measures. $^{\rm 190}$

7.103. The European Union responds that increases in trade restrictive and trade defence measures are not necessarily unforeseen developments, but can be. According to the European Union, the extent, timing, and degree of impact of such increases can make them an unforeseen development, and did in this case.¹⁹¹ The European Union likewise contests the relevance of some of Turkey's factual allegations. According to the European Union, of three datasets that Turkey has submitted showing the evolution of anti-dumping measures, two are not relevant because they are not specific to steel, and one is not relevant because it postdates the Uruguay Round and therefore cannot serve as an indicator of what was unforeseen when the European Communities incurred the relevant obligation of the GATT 1994.¹⁹² The European Union also observes that the trade defence measures either initiated by the European Union or against EU steel exports account for a small proportion of the "worldwide use of trade defence measures in the steel sector".¹⁹³

7.104. To the European Union's objection that the data submitted by Turkey on the use of anti-dumping measures after the Uruguay Round does not help assess whether a particular development was unforeseen when the European Communities incurred the relevant obligation of the GATT 1994, Turkey responds that the data it has provided are intended to show that the use of trade defence measures, in general and for the "steel-related" sector, "follows cycles of varying intensity" and has done so both before and after the Uruguay Round.¹⁹⁴

7.105. In the context of its discussion of post-Uruguay Round data, Turkey also notes that the number of anti-dumping initiations had decreased in 2014 compared to 2013, and in 2017-2018 compared to 2015-2016, and that as a result the number of initiations in 2017-2018 was only slightly 2011-2013. According to Turkey, these facts hiaher than in contradict the European Commission's statement that there had been a steady increase in the recourse to trade defence measures and that the situation constituted an unforeseen development.¹⁹⁵ The European Union responds that the slight decrease in the number of anti-dumping initiations in 2014 was compensated by an increase in countervailing duty initiations the same year¹⁹⁶, that the number of initiations increased again in 2015¹⁹⁷, and that the decrease in the number of initiations in 2017-2018 compared to 2015-2016 did not call into question the unforeseen nature "of the overall significant increase of the use of trade defence measures".¹⁹⁸

7.106. Finally, Turkey argues that the fact that the United States had in place a large number of trade defence measures on steel in 2018 could not be unforeseen because the United States was one of the largest steel importers, and had had a similarly high number of trade defence measures on steel in place in 2002-2006.¹⁹⁹ The European Union responds that this does not preclude a significant increase in trade defence measures by the United States, "as part of a global increase in the use of such instruments", from constituting an unforeseen development. In response to Turkey's argument, the European Union also provides an overview of the evolution of the number

¹⁹⁶ European Union's first written submission, paras. 92 and 95 (referring to WTO statistics on antidumping initiations by sector, (Exhibit EU-2); and WTO statistics on countervailing initiations by sector, (Exhibit EU-3)). The data on initiations per year in the provisional determination, recital 34, refer to anti-dumping initiations. (WTO statistics on antidumping initiations by sector, (Exhibit EU-2)).

¹⁹⁰ Turkey's first written submission, para. 107.

¹⁹¹ European Union's first written submission, paras. 89-90.

¹⁹² European Union's first written submission, para. 91 (referring to Turkey's first written submission, figures 2 and 3) and para. 92 (referring to Turkey's first written submission, figure 4). The data in figure 3 also postdates the Uruguay Round.

¹⁹³ European Union's first written submission, paras. 88-106.

¹⁹⁴ Turkey's second written submission, paras. 68-75; request for interim review, para. 19. Turkey also reiterates its earlier arguments and notes that reliance by the European Union on an OECD presentation is an *ex post* explanation. (Turkey's second written submission, para. 75).

¹⁹⁵ Turkey's first written submission, para. 105; second written submission, paras. 70-71; and request for interim review, para. 19.

¹⁹⁷ European Union's first written submission, para. 95.

¹⁹⁸ European Union's first written submission, para. 96. The European Union also notes that the WTO statistics on base metals and articles "show a significant increase [in the number of initiations] during 2014-2018 compared to the period 2011-2013", and also show that 498 antidumping and 61 countervailing measures were adopted worldwide in the (six-year) period 2013-2018 compared to 163 and 25, respectively, in the (seven-year) period 2006-2012. (European Union's first written submission, para. 93).

¹⁹⁹ Turkey's first written submission, para. 108.

of anti-dumping and countervailing duty investigations and orders by the United States over several years.²⁰⁰

7.107. Among the third parties, Brazil argues that accepting that trade defence measures, which are "an integral part of the WTO system"²⁰¹, may constitute an unforeseen development risks leading to an escalation in trade barriers; and that to establish that they do constitute unforeseen developments, there must at least be solid supporting data and an "adequate threshold".²⁰² Switzerland shares the concern of an escalation of trade restrictions, and adds that an exercise of rights under the covered agreements cannot be an unforeseen development.²⁰³ The Republic of Korea notes that the use of trade defence measures is presumably "lawful".²⁰⁴

7.108. We do not consider that the fact that the WTO Agreement may contemplate particular events or circumstances establishes that they cannot constitute an unforeseen development. By way of illustration, we note that while the GATT 1994 refers to "action ... taken in time of war", this statement, standing alone, does not necessarily preclude actions taken in time of war from constituting an unforeseen development.

7.109. Regarding the European Union's contribution to the increase, the European Union asserts that any such increase was marginal, and Turkey has not provided us with information rebutting this assertion.²⁰⁵

7.110. We turn to consider Turkey's argument that there are cycles in the use of trade defence measures²⁰⁶, and that for this reason increases in their use are not unexpected. We note that the published determinations place the increase in trade defence measures in the broader context of increased overcapacity and the US Section 232 measures, and cite an increase in the average number of steel-related anti-dumping investigations initiated per year from 77 in 2011-2013 to 117 in 2015-2016 and the fact that the United States had, in February 2018, 169 AD/CV orders in place on steel and 25 ongoing investigations.²⁰⁷ In response to objections that the adoption of trade defence measures could not be unforeseen, the European Commission further explained in the definitive determination that the "issue" was "the unprecedented and increased number of such measures taken by third countries, which ha[d] created trade diversion resulting in increase of imports into the [European Union]".²⁰⁸ Thus, the European Commission did not assert that any uptick in trade defence measures would constitute an unforeseen development, but instead based its finding that these developments were unforeseen on the concrete circumstances before it. The fact that the use of trade defence measures may "follow cycles", or fluctuate, does not mean that an increased use of such measures cannot ever constitute an unforeseen development. The European Commission found that in the particular circumstances before it, the increase in trade defence measures together with the other developments was unforeseen within the meaning of Article XIX:1(a), and we consider that the fact that the use of trade defence measures may fluctuate does not call into question that assessment.

7.111. We next turn to Turkey's arguments regarding the number of worldwide initiations of anti-dumping investigations into base metals and articles in 2014 and 2017-2018, and regarding trade defence investigations and orders by the United States.²⁰⁹ We recall that the European Commission relied on the significant increase in the number of anti-dumping initiations

- ²⁰³ Switzerland's third-party submission, paras. 18-21.
- ²⁰⁴ Korea's third-party submission, para. 17.

²⁰⁵ Turkey notes that the European Union had initiated 29 trade defence investigations for iron and steel between 2013 and 2019, and compares the trade defence measures on steel targeting the European Union with the *total* trade defence measures *targeting the European Union*. The European Union responds that 72 of 723 "trade defence measures of all types" in force in 2020 targeted the European Union, and notes that it initiated 29 trade defence investigations for iron and steel between 2013-2018 out of a total of 582 anti-dumping and 118 countervailing initiations in the "Base metals and articles" sector in the same period 2013-2018. (Turkey's first written submission, para. 107; European Union's first written submission, para. 98).

²⁰⁰ European Union's first written submission, paras. 100-106.

²⁰¹ Brazil's third-party submission, para. 7.

²⁰² Brazil's third-party submission, paras. 7-10.

²⁰⁶ We note in this regard the European Union's observation that much of the data provided by Turkey on the fluctuations in the use of trade defence measures postdate the Uruguay Round or do not concern the steel sector. (See para. 7.103 above).

²⁰⁷ Provisional determination, (Exhibit TUR-3), recital 34.

²⁰⁸ Definitive determination, (Exhibit TUR-5), recital 56.

²⁰⁹ See paras. 7.105 and 7.106 above.

worldwide in the steel sector in the period 2015-2016 as compared to the period 2011-2013, and on the number of steel-related trade defence investigations and orders in place in the United States in February 2018, as evidence that there had been a significant increase in recourse to trade defence measures on steel, in a broader context of overcapacity, trade restrictive measures on steel, and the US Section 232 measures on steel. That the number of worldwide initiations was not as high in 2014 or 2017-2018 does not upset those observations and the conclusions drawn from them by the European Commission that the developments in question were unforeseen. Similarly, it is not merely because the United States is a large market, or because it had already had a high number of measures in place in 2002-2006, that its high number of orders and investigations in 2018 did not contribute to the overall situation that the European Commission found to be "unforeseen".

7.112. We therefore do not consider that Turkey has shown that the European Commission acted inconsistently with Article XIX:1(a) in concluding that the increase in trade defence measures it identified, together with the other developments it identified, was unforeseen.

7.113. Turning to the increase in trade restrictive measures on steel products, Turkey asserts that the European Commission did not clearly identify the measures and did not explain how an increase in a most-favoured-nation tariff or a tariff increase adopted as a trade defence measure would be unforeseen.²¹⁰ In response, the European Union recalls that the provisional determination listed the types of measures taken (tariff increases, standards, local content requirements) and countries adopting such measures (India, Indonesia, Mexico, South Africa, Turkey, and the United States).²¹¹ The European Union also states that more detail is provided in an OECD presentation made at the 83rd session of the OECD Steel Committee, and notes that the provisional determination refers to the reports from the 83rd and 84th sessions of the OECD Steel Committee. However, as that reference is not made in the context of increases in trade restrictive measures, we do not rely on the OECD presentation in this context.²¹² Among the third parties, the Republic of Korea submits that the adoption of government restrictions is also part of the reality in the steel sector, and it is therefore "doubtful" that it may constitute an unforeseen development.²¹³

7.114. Regarding the identification of the trade-restrictive measures, we note that the published determinations identify both the types of measures at issue and a number of countries having adopted them. While Turkey submits that "it is difficult to understand how"²¹⁴ the increase in these measures could be unforeseen, it has not articulated why this is the case. Similar to the increase in trade defence measures, the European Commission cited the increase in trade-restrictive measures in a context that included unprecedented overcapacity and the US Section 232 measures, which constrained the universe of outlets for steel products around the world, thereby "establishing and aggravating imbalances in the international trade of the products concerned".²¹⁵ We do not consider that Turkey has established that the European Commission erred in finding that these events, including the increased recourse to trade-restrictive measures, in this context, by a number of countries, were unforeseen.

7.115. Third, Turkey also submits that the Section 232 measures that the United States adopted on steel could not constitute an unforeseen development because Section 232 of the Trade Expansion Act, the US legislation that allows the adoption of such measures in the United States, long predated the Uruguay Round.²¹⁶ The European Union responds that indeed measures adopted under Section 232 will not always be unforeseen developments, but they might be, and that the European Commission explained why they were in these circumstances.²¹⁷ Among the third parties, Canada takes the view that there could be instances in which Section 232 measures could be unforeseen developments, but that the Panel must assess whether the

²¹⁰ Turkey's first written submission, para. 110.

²¹¹ Provisional determination, (Exhibit TUR-3), recital 33.

²¹² European Union's first written submission, para. 108 (referring to Presentation at the 83rd session of the OECD Steel Committee, (Exhibit EU-7)). In the context of overcapacity and government support measures, fns 1-2 to recitals 31-32 of the provisional determination refer to "reports from the 83rd and 84th OECD Steel Committee" and provide the following link: <u>http//:www.oecd.org/stid/ind/steel</u>.htm. (Provisional determination, (Exhibit TUR-3), fns 1-2).

²¹³ Korea's third-party submission, para. 17.

²¹⁴ Turkey's first written submission, para. 110.

²¹⁵ Provisional determination, (Exhibit TUR-3), recital 30.

²¹⁶ Turkey's first written submission, para. 111; second written submission, para. 76.

²¹⁷ European Union's first written submission, para. 110 (referring to Provisional determination,

⁽Exhibit TUR-3), recital 35).

European Commission gave a reasoned and adequate explanation of why this was so in the case before it. $^{\rm 218}$

7.116. The fact that existing legislation of a Member may authorize that Member to adopt a particular measure does not in itself establish that all measures that may be applied by virtue of that legislation are foreseen. By way of illustration, a Member's legislation may establish certain rules regarding military conflict, but that does not mean that any military conflict engaged in by that Member cannot be "unforeseen" within the meaning of Article XIX:1(a). Turning to the published determinations in the present case, we note that the European Commission referred to the "level and scope" of the US Section 232 measures on steel, noting that it consisted of "a single across-product tariff ... with almost no country exclusion", and that the measure was expected to reduce imports of steel products into the United States by 13 million tonnes. The European Commission also took the view that the measure was "illegal", and observed that it was being introduced "in the context of the prevailing persistent ... overcapacity".²¹⁹ The European Commission thus articulated several reasons why it regarded these specific Section 232 measures, and the events that included these measures, as unforeseen. The fact that the Section 232 measures were adopted in accordance with longstanding US legislation does not establish that the European Commission erred in considering those specific Section 232 measures to be an unforeseen development.

7.117. We therefore find that Turkey has not established that the European Commission erred in finding that the developments it relied upon were unforeseen.

7.3.2.3 Whether the European Commission did not establish that the increase in imports was "as a result of" the unforeseen developments

7.118. Turkey's third set of arguments is that the European Commission did not establish the requisite "logical connection" between the unforeseen developments and the increase in imports. Specifically, Turkey argues that: (a) the European Commission did not do enough to establish how the unforeseen developments resulted in the increase in imports into the European Union²²⁰; (b) the unforeseen developments related to steel in general, and the European Commission did not demonstrate that they resulted in the increase in imports of the specific steel products under investigation²²¹; and (c) the US Section 232 measures, introduced in March 2018, could not have resulted in the increase in imports observed during 2013-2017.²²² We consider these arguments in turn.

7.3.2.3.1 Whether the European Commission did not do enough to establish how the unforeseen developments had resulted in the increase in imports into the European Union

7.119. We begin by considering Turkey's argument that the European Commission did not do enough to establish how the unforeseen developments (i.e. overcapacity, the increase in trade restrictive and trade defence measures, and the Section 232 measures) resulted in the increase in imports at issue, and instead made assertions unsupported by evidence.

7.120. In considering these arguments, we note that the parties disagree on how much was required in this case to establish that the unforeseen developments resulted in the injurious increase in imports into the European Union. On the one hand, Turkey argues that as the European Commission relied on a complex set of facts to establish the "logical connection" between the unforeseen developments and the resulting increase in imports, the European Commission was required to provide a "detailed explanation ... of how the alleged unforeseen developments resulted in increases

²¹⁸ Canada's third-party submission, para. 25.

²¹⁹ Provisional determination, (Exhibit TUR-3), recital 35.

²²⁰ Turkey's first written submission, paras. 124-135; second written submission, paras. 80-92; response to Panel question No. 22(a); and comments on the European Union's response to Panel question No. 22(a)-(b).

²²¹ Turkey's first written submission, paras. 136-143; second written submission, paras. 93-95; opening statement at the first meeting of the Panel, para. 22; response to Panel question No. 22(a), paras. 4-8; and comments on European Union's response to Panel question No. 21, paras. 8-10. Turkey also argues that the European Union failed to carry out an analysis at the level of each specific product category: we have addressed this argument in section 7.2 above.

²²² Turkey's first written submission, paras. 132 and 144-149; second written submission, paras. 87-88.

in imports".²²³ The European Union, on the other hand, considers that the facts of this case were such that it was sufficient to just "put together" the unforeseen developments and the increase in imports to show their logical connection, including because they took place during the same period.²²⁴ Among the third parties, Canada submits that we must determine whether it was sufficient to "bring [the] two sets of facts together" as the European Union argues, given Turkey's arguments on the complexity of the facts.²²⁵ Canada asserts that although "a competent authority enjoys a certain latitude in choosing the appropriate method" to examine the relationship between the unforeseen developments and the increase in imports, "it must provide a reasoned and adequate explanation of its findings".²²⁶ Japan argues *inter alia* that the authority must demonstrate how the unforeseen developments resulted in the increase in imports, and that for this to be the case, the unforeseen developments must modify the conditions of competition in favour of the imported products.²²⁷ The Republic of Korea submits that we must focus on whether the explanation in the published reports is sufficiently reasoned and adequate, and that we must consider "whether complex phenomena can be reduced to a simple matter of 'putting facts together'".²²⁸ Switzerland argues that merely putting together the unforeseen developments and the increase in imports is not enough.229

7.121. With this in mind, we consider the parties' specific arguments on each of the unforeseen developments relied upon by the European Commission.

7.122. We begin with the increase in overcapacity. Turkey argues that the European Commission only offered general and vague statements, unsupported by evidence, regarding the connection between overcapacity and the increase in imports.²³⁰ Among other things, Turkey argues that although the European Commission asserted that "many steel producers, notably in countries where the State distorts the normal play of market forces, kept capacity utilisation at high rates and flooded third country markets with their products at low prices", and that "[t]his ha[d] resulted in increasing imports in the EU and overall price depression", the European Commission did not support these assertions with evidence.²³¹ Turkey also argues that the European Commission did not provide evidence to support the assertion that where there is spare capacity, exporting producers will seek opportunities on export markets.²³² Turkey notes that, insofar as it refers to China, this statement also "fails to take into account the existence of numerous trade remedies in place".²³³ Turkey also recalls that the panel in US - Steel Safeguards found that even if large volumes of foreign steel production were displaced from foreign consumption, this would not suffice to establish that imports into the Member concerned increased as a result.²³⁴ To support its position that overcapacity does not necessarily lead to increased imports, Turkey notes that a report on the record of the underlying proceeding indicated that the trade effects of overcapacity were "systemically unimportant". 235

7.123. The European Union responds that when one considers side by side the increase in overcapacity and the increase in imports into the European Union, their logical connection is apparent.²³⁶ The European Union adds that the increase in overcapacity (and the other developments) occurred during the same period as the increase in imports: overcapacity had been increasing since 2000 and reached high levels in 2014-2016, and the increase in imports had been observed between 2013 and June 2018.²³⁷ The European Union also submits that the

- ²²⁵ Canada's third-party submission, para. 30.
- ²²⁶ Canada's third-party submission, para. 31.

²²⁷ Japan's third-party submission, paras. 9-10; third-party statement, paras. 2-4. See also

Ukraine's third-party statement, para. 9.

²³⁰ Turkey's first written submission, paras. 126-127.

(Exhibit TUR-21), p. 6). On the latter, see European Union's response to Panel question No. 9, paras. 29-35. ²³⁶ European Union's first written submission, para. 116.

²²³ Turkey's second written submission, paras. 80-81. See also Turkey's first written submission, para. 125. ²²⁴ European Union's first written submission, paras. 115-117.

²²⁸ Korea's third-party submission, para. 19.

²²⁹ Switzerland's third-party submission, paras. 23-24.

²³¹ Turkey's first written submission, para. 126 (referring to Provisional determination, (Exhibit TUR-3), recital 32).

²³² Turkey's first written submission, para. 127 (referring to Definitive determination, (Exhibit TUR-5), recital 54).

²³³ Turkey's first written submission, para. 127.

²³⁴ Turkey's first written submission, para. 128.

²³⁵ Turkey's first written submission, para. 129 (referring to Global Trade Alert Report, "Going Spare",

²³⁷ European Union's first written submission, para. 117.

European Commission's findings on overcapacity were "justified by evidence". The European Union refers "for example" to the European Commission's communication "Steel: Preserving Sustainable Jobs and Growth", and to the Chair's statements at the 83rd and 84th sessions of the OECD Steel Committee.²³⁸ Turkey submits that these materials were not cited in the determinations to support the European Commission's findings on the logical connection between the unforeseen developments and the increase in imports, and that in any event they do not establish that logical connection.²³⁹ Turkey also argues that the European Union's choice of record evidence is biased, because the European Union is omitting to refer to a report according to which the trade effects of overcapacity were unimportant.²⁴⁰ The European Union notes that the passage of the report that Turkey is referring to does not refer specifically to the steel sector.²⁴¹

7.124. To evaluate these arguments, we again turn to the published determinations. These determinations indicate that the European Commission found that global steelmaking capacity had more than doubled since 2000 (and in particular increased unexpectedly between 2011 and 2016²⁴²), that actual steel production far exceeded demand²⁴³, and that there had been a sudden, sharp, and significant increase in imports between 2013 and both 2017 and the first half of 2018.²⁴⁴ Regarding the connection between overcapacity and the increase in imports, the provisional determination stated that "given the important fixed costs in the steel sector, many steel producers, notably in countries where the State distorts the normal play of market forces, kept capacity utilisation at high rates and flooded third country markets with their products at low prices when they could not be absorbed by domestic consumption", and that this "has resulted in increasing imports into the EU and overall price depression".²⁴⁵ The provisional determination also stated that import prices "generally" undercut prices of EU producers in 2017. In the definitive determination, the European Commission observed that:

As far as the link between the unforeseen development of steel overcapacity and the increase in imports is concerned, it is clear that exporting producers have an interest in maximizing their capacity utilization. In situations where spare capacity is available after supplying their domestic market, they will seek other business opportunities on export markets and thus generate an increase in import volumes on such markets.²⁴⁶

7.125. In the provisional determination, the European Commission referenced "reports from the 83rd and 84th Steel Committee" in support of its statements that overcapacity had more than doubled since 2000, and that distortive government support accentuated the imbalances in the steel sector.²⁴⁷ In the definitive determination, the European Commission referred to its communication "Steel: Preserving Sustainable Jobs and Growth", and to the Global Trade Alert Report "Going Spare", as supporting its statements that overcapacity had "unexpectedly continued to increase", and that it had increased between 2011 and 2016 after decreasing from 2009 to 2011.248

7.126. In discussing overcapacity, the European Commission also stated that "overcapacity is inherently closely linked to dumped and subsidized imports".249

²³⁸ European Union's first written submission, paras. 118-121 (referring to the

European Commission's Communication "Steel: Preserving Sustainable Jobs and Growth", (Exhibit EU-23); and to Chair's statements at the 83rd and 84th session of the OECD Steel Committee, respectively, (Exhibits EU-8 and EU-9)).

²³⁹ Turkey's second written submission, paras, 83-84,

²⁴⁰ Turkey's second written submission, para. 85; request for interim review, para. 21 (referring to Global Trade Alert Report, "Going Spare", (Exhibit TUR-21)). ²⁴¹ European Union's response to Panel question No. 9, para. 34. We agree with the European Union

that the passage at issue refers to overcapacity in general and not specifically in the steel sector. (See also ibid. paras. 26-36).

²⁴² Definitive determination, (Exhibit TUR-5), recital 52.

²⁴³ Provisional determination, (Exhibit TUR-3), recital 31.

²⁴⁴ Provisional determination, (Exhibit TUR-3), recital 29; Definitive determination, (Exhibit TUR-5), recital 47.

²⁴⁵ Provisional determination, (Exhibit TUR-3), recital 32.

²⁴⁶ Definitive determination, (Exhibit TUR-5), recital 54.

²⁴⁷ Provisional determination, (Exhibit TUR-3), recitals 31-32, fn 1 ("Cf. reports from the 83rd and 84th OECD Steel Committee, available at http://www.oecd.org/sti/ind/steel.htm"), and fn 2 ("Idem, 83rd report").

²⁴⁸ Definitive determination, (Exhibit TUR-5), recital 52. The same communication was also cited in the context of the analysis of threat of injury, in Provisional determination, (Exhibit TUR-3), recitals 58-60. ²⁴⁹ Definitive determination, (Exhibit TUR-5), recital 53.

7.127. We recall that Article XIX:1(a) provides that a Member may adopt a safeguard where, among other things, a product is being imported in increased quantities "as a result of" unforeseen developments and the effect of GATT obligations, and that the phrase "as a result of" does not establish a causation requirement.²⁵⁰ This means that an investigating authority need not provide the same quantum of reasoning and evidence to substantiate the "logical connection" between unforeseen developments and the increase in imports as it would be if Article XIX:1(a) had contained a causation requirement. Further, we recall that what precisely is required to establish that the increase in imports is "as a result of" the unforeseen developments will depend on the "nature of the facts" in any given case.²⁵¹

7.128. In this instance, we note that the European Commission established that global steel overcapacity greatly increased, reaching 2,29 billion tonnes in 2016 and remaining steady at 2,17 billion tonnes in 2017²⁵², and that Turkey does not dispute this fact. We also note that the European Commission established that imports into the European Union of the 26 product categories on which it imposed the definitive safeguard increased significantly between 2013 and June 2018²⁵³, and although Turkey takes issue with this assessment, we do not find fault with this finding.²⁵⁴ We further note that the European Commission established that for 16 of the steel product categories subject to the definitive safeguard, import prices undercut the prices of EU producers²⁵⁵, and that the European Commission provided a reference to support its statement that distortive government support was accentuating the structural imbalances in the steel sector, and Turkey has not contested this reference.²⁵⁶ The European Commission also explained the connection between the increase in overcapacity and the increase in imports, by stating that because of the high fixed costs in the steel sector, many steel producers kept capacity utilization rates high, especially in countries with distortive government support, and that as a result of this, they "flooded third country markets with their products at low prices when they could not be absorbed by domestic consumption"; in turn, this "resulted in increasing imports into the EU and overall price depression".²⁵⁷

7.129. The published determinations thus indicate that the European Commission provided evidentiary support for its statements regarding the increase in overcapacity, the unexpected nature of its continued increase, and its links to government support. However, the determinations do not point to evidence in support of the European Commission's statements regarding the connection between the overcapacity and the increase imports. Instead, in the European Commission's reasoning on the connection between overcapacity and the increase in imports into the European Union was limited to noting the magnitude of global overcapacity and excess production, asserting that "it is clear" that overcapacity leads steel producers to seek other export opportunities, observing that import prices were typically lower than prices of EU producers, and concluding that this has resulted in increased imports into the European Union.²⁵⁸

7.130. It may well be that overcapacity led to pressure to export, and that that pressure to export led to increased exports to the European Union, so that EU imports of the product under investigation increased as a result of overcapacity. Notwithstanding this, we note that, except for the description of the magnitude of the overcapacity and the increase in imports, the

²⁵² Provisional determination, (Exhibit TUR-3), recital 31 and fn 1; Definitive determination, (Exhibit TUR-5), recital 52 and fns 19-20.

²⁵³ Definitive determination, (Exhibit TUR-5), section 3.

²⁵⁴ See section 7.5 below.

²⁵⁵ Provisional determination, (Exhibit TUR-3), recital 32 and annex III. The European Commission also found import prices undercut the prices of EU producers for product category 23, to which it did not apply the definitive safeguard. (Definitive determination, (Exhibit TUR-5), recital 31).
²⁵⁶ Provisional determination, (Exhibit TUR-3), recital 32 and fn 2. Turkey contests the

²⁵⁶ Provisional determination, (Exhibit TUR-3), recital 32 and fn 2. Turkey contests the European Union's reliance on the OECD Chair's statement at the 83rd session of the OECD Steel Committee, (Exhibit EU-8) on the basis that the European Commission did not rely on it as evidence of the connection between the increase in overcapacity and the increase in imports. Turkey does not dispute the European Commission's reliance on it for the proposition that government support has accentuated the imbalances in the steel sector. (See para. 7.124 above).

²⁵⁷ See paras. 7.124-7.125 above.

²⁵⁸ As noted, the European Commission offered evidence for its statements about the evolution of overcapacity and about distortive government support, but not for its statements regarding the logical connection between overcapacity and the increase in imports. (See para. 7.125 above).

²⁵⁰ See para. 7.84 above.

²⁵¹ See para. 7.85 above.

European Commission's determinations do not provide evidence that would otherwise support its reasoning on the connection between the two.

7.131. We do not consider this to be a circumstance in which a great deal of analysis was required to establish a logical connection between the unforeseen development (i.e. overcapacity) and the increase in imports.²⁵⁹ The European Commission, however, did not satisfy this requirement, because it merely asserted that the connection existed, without referencing any evidence for its assertion.

7.132. In reaching this conclusion, we have considered a recent WTO dispute where no Article XIX:1(a) inconsistency was found to flow from an authority's determination that certain unforeseen developments resulted in an increase in imports based on their coincidence in time. In *US – Safeguard Measure on PV Products (China)*, one of the unforeseen developments that the US authorities cited was the rapid change in the global supply chains and manufacturing processes that followed the imposition of US AD/CV duties on photovoltaic (PV) products from China and Chinese Taipei. The panel found that the US authorities had shown that Chinese firms had made significant investments in PV production in several third countries, and that PV imports into the United States from those countries had significantly increased. China argued that it was inappropriate for the US authorities to infer that the increase in production capacity of Chinese-affiliated PV producers contributed to the increased imports into the United States, suggesting that the increase could have come from other producers. The panel rejected that argument, explaining:

[E]ven if this were a hypothetical possibility, the circumstantial evidence relied upon by the [United States International Trade Commission] suggests otherwise. Indeed, in line with the United States' argument, the fact that Chinese-affiliated companies significantly increased their production capacity in third countries in the same years when imports from those countries into the US market significantly increased suggests to us that there was a meaningful connection between these developments.²⁶⁰

7.133. This analysis appears to support the view that in some instances demonstrating a coincidence in time between certain unforeseen developments and an increase in imports, and explaining their connection, is sufficient to establish their logical connection for purposes of Article XIX:1(a).²⁶¹ In that case there was, among other elements²⁶², an increase in production capacity of Chinese-affiliated producers in four countries that matched a simultaneous increase in imports from those same countries, all over a two-year period. Thus, there was a closer link in that case between the unforeseen development at issue and the increase in imports than there is in the present case between overcapacity in the global steel industry in general and an increase in imports in general.

7.134. We next consider the European Commission's examination of the connection between the <u>increase in trade restrictive and trade defence measures</u> and the increase in imports. Turkey submits that the European Commission "provided no detailed factual or statistical record" on how steel flows into the European Union were impacted by the trade restrictive and trade defence measures it referred to, and did not identify those measures.²⁶³ The European Union responds that nothing more was required other than establishing the increase in trade defence measures and the increase in imports "during the same period".²⁶⁴

7.135. We turn again to the published determinations. In the provisional determination, the European Commission stated that the increasing imports into the European Union and overall price depression had "been exacerbated by trade-restrictive practices in third country markets". The European Commission gave the example of tariff increases on certain steel products that "were typically imported in increasing quantities over the period of investigation [(POI)]". The

²⁵⁹ We recall that Article XIX:1(a) does not require a causation relationship between the unforeseen developments and the increase in imports. (See para. 7.84 above).

²⁶⁰ Panel Report, *US – Safeguard Measure on PV Products (China)* [appealed by China 16 September 2021], para. 7.42.

²⁶¹ See also Panel Reports, US – Steel Safeguards, para. 10.115.

²⁶² See Panel Report, *US – Safeguard Measure on PV Products (China)* [appealed by China 16 September 2021], paras. 7.21 and 7.39-7.44.

²⁶³ Turkey's first written submission, paras. 130-131; second written submission, para. 86.

²⁶⁴ European Union's first written submission, paras. 117 and 123.

European Commission went on to list other trade restrictive practices; and then it noted that while in 2011/2013 there had been 77 trade defence initiations related to steel per year, that had increased to 117 per year in 2015/2016. It then referred to the large number of AD/CV duty orders and investigations in the United States, and stated that "as the U.S. is one of the world's largest steel-importing countries ... the impact of such a large number of trade remedies has been strongly felt globally".²⁶⁵ In the definitive determination, addressing arguments that it had not established the connection between the trade restrictive and trade defence measures and the increase in imports, the European Commission recalled that while in 2011/2013 there had been 77 trade defence initiations per year related to steel, that had increased to 117 per year in 2015/2016, and stated that "no party [had] questioned these figures which indicate an unforeseen development, leading to the increase of imports established above".²⁶⁶

7.136. Except for the trade defence measures adopted by the United States, the European Commission mainly described these unforeseen developments and asserted that they had led to the increase in imports into the European Union. As regard the trade defence measures adopted by the United States, the European Commission reasoned that because the United States is one of the largest steel importers, the effect of the large number of measures adopted by the United States had "been strongly felt globally".²⁶⁷

7.137. We note that each trade restrictive and trade defence measure typically concerns only certain steel products imported into and exported from certain countries. Therefore, recalling that the type of analysis required to establish the connection between the unforeseen developments and the increase in imports depends on the nature of the facts, we consider that this is a situation where a "more detailed analysis" is required.²⁶⁸ The European Commission did not offer such an analysis. Other than for the US trade defence measures, the European Commission merely asserted that the trade restrictive and trade defence measures had led to the increase in imports into the European Union; for the US trade defence measures, the European Commission asserted that their impact had been felt globally, but offered no evidence and made no attempt to establish a link between the measures and the increase in imports into the European Union. For example, while it indicated the increased number of US trade actions, it gave no information on the volume of imports affected.

7.138. Therefore, we find that the European Commission did not establish that the increase in imports was "as a result of" the increase in trade-restrictive and trade defence measures.

7.139. We next consider the US Section 232 measures on steel. Turkey argues that the European Commission did not establish that those measures resulted in the increase in imports of the 26 product categories concerned into the European Union, because: (a) the Section 232 measures excluded several products but the European Commission did not carry out a product-specific analysis; (b) the European Commission did not conduct a country-specific analysis to show that there had been increases in imports into the European Union from the countries whose exports to the United States had decreased; (c) the European Commission offered no evidence for its assertion that the EU market was attractive, and in any event that would not have been sufficient to establish that the Section 232 measures resulted in the increase in imports into the European Union; and (d) the Section 232 measures entered into force after the POI over which the increase in imports was observed.²⁶⁹

7.140. The European Union responds that: (a) the exclusions of products from certain countries did not change the authority's conclusions²⁷⁰; (b) a country-specific analysis was not necessary here, given the global nature of the unforeseen developments²⁷¹; (c) the EU market is attractive because it is the largest steel importing market, with high prices, and this is "public knowledge"²⁷²; (d) there

²⁶⁵ Provisional determination, (Exhibit TUR-3), recitals 33-34.

²⁶⁶ Definitive determination, (Exhibit TUR-5), recital 56.

²⁶⁷ Provisional determination, (Exhibit TUR-3), recital 35.

²⁶⁸ Panel Reports, US – Steel Safeguards, para. 10.115.

²⁶⁹ Turkey's first written submission, paras. 132-135; second written submission, paras. 87-92;

responses to Panel question Nos. 22 and 23; and comments on European Union's response to Panel question No. 22, para. 12.

²⁷⁰ European Union's first written submission, para. 127.

²⁷¹ European Union's first written submission, para. 128.

²⁷² European Union's response to Panel question No. 22(a), para. 9.

is evidence of the attractiveness of the EU market²⁷³; and (e) the attractiveness of the EU market, when combined with the unforeseen developments and their overlap in time with the increase in imports, establishes the logical connection between the unforeseen developments and the increase in imports.²⁷⁴ As regards the timing of the Section 232 measures, the European Union responds that: the POI for the increase in imports ended in June 2018, after the imposition of the Section 232 measures; the European Commission ascertained the existence of actual trade diversion based on data on monthly imports into the United States and the European Union from January to September 2017 and 2018; and the very initiation of the Section 232 investigation in April 2017 had already created uncertainty in the market.²⁷⁵

7.141. We turn once more to the published determinations to make our assessment. In the provisional determination, the European Commission stated that: the "U.S. Section 232 measures, given their level and scope, are likely to cause substantial trade diversion of steel products into the Union"; the United States had calculated that the Section 232 measures would decrease imports by an amount that corresponded to 7% of EU consumption; the EU market "is generally a very attractive market for steel products both in terms of demand and prices"; and "some of the main exporters to the US are also traditional steel suppliers to the Union and there is no doubt that these countries, as well as others whose exports and production will be affected by the U.S. measures and the foreseeable trade diversion cascade, will redirect their exports to the Union".²⁷⁶ In the definitive determination, the European Commission noted that the Section 232 investigation had been initiated in April 2017 and the resulting report published in January 2018, and asserted that even the initiation of the investigation "did undoubtedly create uncertainty on the market and caused effects on steel trade flows".²⁷⁷ Further, the European Commission examined monthly imports of steel into the United States and the European Union from January to September 2017 and 2018; it found that, in 2018, monthly imports into the United States had been "consistently lower" than in 2017, with the exception of April, and that correspondingly monthly imports into the European Union had been higher, each month, than in 2017; according to the European Commission, this established that trade diversion was already taking place, and more specifically that the Section 232 measures "sped up the increase in imports by adding further trade diversion".²⁷⁸ Next, addressing arguments regarding "product exclusions", the European Commission stated that only Australia had obtained an unconditional exemption, and accounted for around 1% of total US imports, whereas other countries had been allocated tariff-free quotas, many of which had been exhausted upon allocation, and moreover those countries accounted for less than 20% of 2017 imports.²⁷⁹

7.142. Thus, the European Commission's finding that the US Section 232 measures resulted in the increase in imports (by "adding further trade diversion") rested on the following: the "level and scope" of US measures, which according to the US Department of Commerce were expected to cut out of imports into the United States a volume equivalent to 7% of EU consumption²⁸⁰; the assertions that the European Union is "generally a very attractive market for steel products both in terms of demand and prices", and that some of the suppliers of the US market were also "traditional suppliers" of the European Union²⁸¹; and data showing that between the imposition of the Section 232 tariff on steel in March 2018 and September 2018, monthly imports of steel into the United States had decreased compared to the same month of 2017, except in April, and monthly imports into the European Union had correspondingly increased.²⁸²

7.143. Similar to the trade restrictive and trade defence measures that we have just considered, we take the view that establishing the connection between the Section 232 measures on steel and the increase in imports into the European Union "require[s] a much more detailed analysis" than "bringing two sets of facts together".²⁸³ This is particularly in view of the fact that the Section 232 measures have been adopted by a single Member.

²⁷³ European Union's response to Panel question No. 22(a), para. 10.

²⁷⁴ European Union's response to Panel question No. 22(b), para. 11.

²⁷⁵ European Union's first written submission, paras. 124-129 and 139-145.

²⁷⁶ Provisional determination, (Exhibit TUR-3), recital 35.

²⁷⁷ Definitive determination, (Exhibit TUR-5), recital 58.

²⁷⁸ Definitive determination, (Exhibit TUR-5), recitals 58-59 and tables 12 and 14.

²⁷⁹ Definitive determination, (Exhibit TUR-5), recitals 60-61.

²⁸⁰ Provisional determination, (Exhibit TUR-3), recital 35 and fn 2.

²⁸¹ Provisional determination, (Exhibit TUR-3), recital 35.

²⁸² Definitive determination, (Exhibit TUR-5), recital 59 and tables 12 and 14.

²⁸³ See para. 7.85 above, and Panel Reports, US – Steel Safeguards, para. 10.115.

- 47 -

7.144. The Section 232 analysis that the European Commission conducted with regard to the Section 232 measures does not satisfy this more detailed standard. In terms of evidence, we note that the European Commission relied on an estimate by the United States according to which the measures would have reduced steel imports into the United States by a total amount corresponding to 7% of EU steel consumption, and on six months of data following the imposition of the Section 232 steel tariffs that showed some negative correlations between trends in imports into the European Union and the United States. For the remainder of its analysis, the European Commission did not offer evidence or sufficiently detailed information.²⁸⁴

7.145. In particular, a key element of the European Commission's analysis was that the EU market was attractive for exporters forced out of the US market. While there appears to be evidence on the record that would support this proposition, this evidence is limited to a model presented to the European Commission by the EU industry²⁸⁵, which the European Commission used in its fashioning of the safeguard it applied, but made no mention of in discussing the connection between the Section 232 measures and the increase in imports.²⁸⁶ The other evidence pointed to by the European Union in this context was either not referred to by the European Commission as establishing the attractiveness of the EU market for exporters forced out of the US market, or is not on the record. First, the European Union now refers to the EU importing market shares reported in presentations made at the 83rd and 84th sessions of the OECD Steel Committee, showing that the "was the main importing economy worldwide".287 But although the European Union provisional determination, in discussing overcapacity, referred to "reports from the 83rd and 84th sessions of the OECD Steel Committee"²⁸⁸, the investigating authority did not refer to these "reports" as establishing the size of the EU market.²⁸⁹ Second, the European Union refers to the European Commission's findings on undercutting, but this was described in the published determinations as an effect of the unforeseen developments, and not as evidence of the attractiveness of the EU market.²⁹⁰ Third, the European Union refers to "publicly available data in the Global Trade Atlas database", but these data were not on the record of the investigation.²⁹¹ Fourth, the European Union asserts that the existence of trade defence measures establishes that the European Union was an attractive target for exports, but this assertion does not feature anywhere in the reasoning of the investigating authority.²⁹²

7.146. Thus, while it appears that evidence of the attractiveness of the EU market for exporters excluded from the US market might have been readily available to the European Commission, the European Commission in its published determinations did not rely on that evidence to establish that its own market was attractive to exporters excluded from the US market because of the Section 232

178-180). ²⁸⁷ European Union's opening statement at the first meeting of the Panel, para. 56; response to Panel question No. 22(a), para. 10 (referring to Presentation at the 83rd session of the OECD Steel Committee, (Exhibit EU-7); and Presentation at the 84th session of the OECD Steel Committee, (Exhibit EU-16)).

²⁸⁸ Provisional determination, (Exhibit TUR-3), fn 1 ("Cf. reports from the 83rd and 84th OECD Steel Committee, available at <u>http://www.oecd.org/sti/ind/steel.htm</u>"). Turkey points out that the two OECD documents that the European Union is referring to as establishing the attractiveness of the EU market, submitted to us as Exhibits EU-7 and EU-16, were not on the record of the safeguard investigation. (Turkey's comments on the European Union's response to Panel guestion No. 22, para. 12).

²⁸⁹ We also note that the OECD documents cited by the European Union in this context (Exhibits EU-7 and EU-16) are not the same OECD documents that the European Union has put forward as the relevant "reports" in the context of the discussion of overcapacity (Chair's statements at the 83rd and 84th session of the OECD Steel Committee, respectively, (Exhibits EU-8 and EU-9)).

²⁹¹ European Union's response to Panel question No. 22(a), para. 10; Turkey's comments on the European Union's response to Panel question No. 22, para. 12.

²⁹² European Union's opening statement at the first meeting of the Panel, para. 57; response to Panel question No. 22(a), para. 10. See also Provisional determination, (Exhibit TUR-3); and Definitive determination, (Exhibit TUR-5).

²⁸⁴ See para. 7.142 above.

²⁸⁵ Calculation deflection, (Exhibit EU-12); Estimate of imports to the US that will be deflected, (Exhibit EU-13).

²⁸⁶ Before us, the European Union has referred to this model as evidence of the connection between the Section 232 measures and the increase in imports. (European Union's response to Panel question No. 11; comments on Turkey's response to Panel question No. 22).

However, the published determinations indicate that that analysis was used to determine the appropriate level of the measures, and not to establish the connection between the unforeseen developments and the increase in imports; therefore, we do not consider it in this context. (Provisional determination, (Exhibit TUR-3), recitals 101-111; Definitive determination, (Exhibit TUR-5), recitals 167-169, 174-176, and 178-180).

²⁹⁰ Provisional determination, (Exhibit TUR-3), recital 32.

measures. Yet, this proposition was central to the European Commission's conclusion that the Section 232 measures resulted in the increase in imports into the European Union, by "speeding up" the pre-existing increase.

7.147. Similarly, we note that while the European Commission asserted that some of the suppliers to the United States were also suppliers of the EU market, and that there was "no doubt that these countries as well as others" would redirect their exports from the United States to the European Union²⁹³, the European Commission did not reference evidence in its determination that supported this assertion.

7.148. We therefore consider that the European Commission did not establish that the Section 232 measures resulted in the increase in imports into the European Union.

7.3.2.3.2 Unforeseen developments relating to steel in general

7.149. Turkey argues that the unforeseen developments relied upon by European Commission pertained to steel in general, and not to the specific products under investigation, and that therefore the European Commission did not demonstrate that they resulted in the increase in imports of the specific products under investigation.²⁹⁴

7.150. Having already found that the European Commission has not established that those unforeseen developments resulted in the increase in imports, we do not examine this further argument advanced by Turkey.

7.3.2.3.3 The timing of the Section 232 measures

7.151. Turkey argues that the increase in imports could not have been "as a result of" the Section 232 measures, because these were introduced in March 2018, after the observed increase in imports.²⁹⁵

7.152. The European Union responds *inter alia* that the POI for the increase in imports ended in June 2018, thus after the imposition of the Section 232 measures; that nothing requires a Member to rely on evidence that predated *initiation* of a safeguard investigation; that the European Commission also ascertained the existence of trade diversion based on data on monthly imports into the United States and into the European Union in January to September 2017 and 2018; and also that the European Commission found that the very initiation of the investigation in April 2017 had created uncertainty and affected trade flows.²⁹⁶

7.153. Having found that the European Commission has not established that the Section 232 measures resulted in the increase in imports, we do not examine this further argument advanced by Turkey.

7.3.2.3.4 Overall conclusion on the connection between the unforeseen developments and the increase in imports

7.154. We therefore find that the European Union acted inconsistently with Article XIX:1(a) of the GATT 1994 by imposing the definitive safeguard without establishing that the increase in imports took place "as a result of" the unforeseen developments.

7.4 The effect of obligations

7.155. Turkey argues that the definitive safeguard is inconsistent with Article XIX:1(a) of the GATT 1994 because the European Commission did not identify the relevant "obligations incurred" under the GATT 1994, and because the European Commission did not explain how any

²⁹³ Provisional determination, (Exhibit TUR-3), recital 35.

²⁹⁴ Turkey's first written submission, paras. 136-140 and 142-143; second written submission, para. 94. We recall that Turkey makes a related but separate argument that the European Commission was required to

carry out its examination at the level of the individual product categories. We have examined that argument in section 7.2 above.

²⁹⁵ Turkey's first written submission, paras. 132 and 144-149; second written submission, paras. 87-88.

²⁹⁶ European Union's first written submission, paras. 124-129 and 139-145.

such obligations had the effect of constraining the European Union's ability to prevent the threatened injury.²⁹⁷

7.156. In its first written submission, the European Union identified the relevant obligations as "the obligation under the GATT 1994 to not impose any tariffs above 0% (or quantitative restrictions)", and their "effect" as the inability to deviate from those same obligations to prevent the threat of serious injury from the increased imports.²⁹⁸ According to the European Union, those obligations "are self-evident" and do not require additional explanation, and their effects do not require additional demonstration.²⁹⁹ In answer to a question from the Panel, the European Union describes the relevant obligations as "the tariff binding of 0% for all imports included in the product concerned", which "is an obligation on the European Union under Article II:1(a) and II:1(b), in conjunction with Article II:7 of the GATT 1994 and the European Union's Schedule of Concessions and Commitments".³⁰⁰ The European Union acknowledges that it did not identify these obligations in its published determinations.³⁰¹

7.157. In response to the European Union's argument that the relevant obligations are self-evident, Turkey points to the finding of the panel in *Ukraine – Passenger Cars* to assert that the fact that an obligation is "known or knowable" does not relieve the authorities from the requirement to identify the obligation and its effects.³⁰² As to the effects of the obligations, Turkey refers to the finding of the panel in *India – Iron and Steel Products* according to which the authority must "explain how that obligation constrains [the Member's] ability to react to the import surge causing injury to its domestic industry".³⁰³

7.158. Among the third parties, Japan argues that the authority must both identify in its published report the relevant obligations and explain how they "prevented the Member concerned from taking WTO-consistent measures in order to prevent or remedy the change generated by the 'unforeseen developments' in the competitive relationship between imports and domestic like or directly competitive products".³⁰⁴ The United States argues that the authority is not required to demonstrate "the effect of obligations incurred" in its published report, and that "a Member establishes that increased imports are the 'effect of obligations incurred' by identifying a commitment, such as a tariff concession, that prevents it from raising duties on imports".³⁰⁵

7.159. In this section, we first examine the applicable requirements of Article XIX:1(a), and we then assess whether Turkey has established the claimed inconsistency.

7.4.1 The applicable requirements of Article XIX:1(a) of the GATT 1994

7.160. We recall that Article XIX:1(a) of the GATT 1994 provides:

If, as a result of unforeseen developments and of the effect of the obligations incurred by a contracting party under this Agreement, including tariff concessions, any product is being imported into the territory of that contracting party in such increased quantities and under such conditions as to cause or threaten serious injury to domestic producers in that territory of like or directly competitive products, the contracting party shall be free, in respect of such product, and to the extent and for such time as may be necessary

²⁹⁷ Turkey's first written submission, paras. 151-158; second written submission, paras. 98-103.

²⁹⁸ European Union's first written submission, para. 65. The European Union also submits that the relevant concessions need not have been negotiated during the Uruguay Round, but Turkey does not appear to have argued that they should. (European Union's first written submission, paras. 61-62).

²⁹⁹ European Union's first written submission, para. 66.

³⁰⁰ European Union's response to Panel question No. 24(a)-(b), paras. 12-13.

³⁰¹ European Union's response to Panel question No. 24(c), para. 15.

³⁰² Turkey's second written submission, para. 102 (referring to Panel Report, *Ukraine – Passenger Cars*, para. 7.96).

³⁰³ Turkey's second written submission, para. 100 (referring to Panel Report, *India – Iron and Steel Products* [appealed by India on 14 December 2018, appealed by Japan on 21 December 2018], para. 7.89).

³⁰⁴ Japan's third-party submission, paras. 20 and 22-24; third-party statement, paras. 5-6.
³⁰⁵ United States' third-party submission, paras. 14 and 16. See also United States' third-party

statement, paras. 13-14.

to prevent or remedy such injury, to suspend the obligation in whole or in part or to withdraw or modify the concession. 306

7.161. As we have already noted, Article XIX:1(a) is structured as a two-part conditional sentence, and both parts of this conditional sentence contain several clauses. The first part of Article XIX:1(a) is the dependent clause of the conditional sentence: it is introduced by the conjunction "if", and sets out certain circumstances or conditions. The second part of Article XIX:1(a), beginning with "the contracting party shall be free", describes the consequence of those circumstances and conditions. Specifically, that consequence is that the Member concerned "shall be free" to suspend the relevant obligation(s) of the GATT 1994, thus imposing a safeguard.

7.162. Under this conditional sentence, part of the circumstances that trigger a Member's ability to apply a safeguard are an injurious increase in imports "as a result of unforeseen developments and of the effect of the obligations incurred by a contracting party under this Agreement".³⁰⁷ That is, under Article XIX:1(a) the circumstances that lead to the imposition of a safeguard measure include a combination of unforeseen developments and "the effect of the obligations" under the GATT 1994, both of which together result in the increase in imports. The term "obligation" reappears in the last clause of Article XIX:1(a), because the remedy provided for in Article XIX:1(a) is the suspension of "the obligation" "as a result of" which the increase in imports has occurred.

7.163. As we have already noted³⁰⁸, the phrase "as a result of" connects the effect of the obligations incurred under the GATT 1994, on the one hand, and the injurious increase in imports, on the other, by providing that the latter occurs "as a result of" – i.e. is the effect, consequence or outcome of – the former. As we have also already noted³⁰⁹, the phrase "as a result of" is different from "cause".

7.164. We also note that Article 3.1 of the Agreement on Safeguards requires Members to "publish a report setting forth their findings and reasoned conclusions reached on all pertinent issues of fact and law". The existence of the circumstances and conditions that are necessary for a Member to be allowed to adopt a safeguard is "pertinent" to the decision to adopt a safeguard, and the existence of obligations of the GATT 1994 whose effect resulted in the injurious increase in imports is one such circumstance. It is therefore in a published report that Members must identify those obligations.³¹⁰

7.4.2 Whether Turkey has established that the definitive safeguard is inconsistent with Article XIX:1(a) of the GATT 1994 with regard to the "effect of the obligations"

7.165. The parties agree that the European Commission did not identify the relevant obligations of the GATT 1994 in its published determinations. 311

7.166. The European Union asserts that those obligations "are self-evident and do not require any additional explanation".³¹² The European Union also notes that the definitive determination "refers in several instances to a 'free of duty quota' thus identifying the products included in the product concerned as products with a tariff binding of 0%", because "[p]roducts that are free of duty under the safeguards measure were necessarily free of duty before the imposition of this measure".³¹³

7.167. We note, first, that the existence of obligations of the GATT 1994 whose effect results in the injurious increase in imports is one of the cumulative circumstances that must exist for a Member

³⁰⁶ Emphasis added.

³⁰⁷ We thus agree with the finding of the Appellate Body in *Korea – Dairy* that the clause "as a result of unforeseen developments and of the effects of obligations" "describes ... *circumstances* which must be demonstrated as a matter of fact ... for a safeguard measure to be applied consistently with the provisions of Article XIX". (Appellate Body Report, *Korea – Dairy*, para. 85 (emphasis original); see also Appellate Body Report, *Argentina – Footwear (EC)*, para. 85; and Panel Report, *Ukraine – Passenger Cars*, para. 7.52).

³⁰⁸ See para. 7.84 above.

³⁰⁹ See para. 7.84 above.

³¹⁰ We also note that the panel in *Ukraine – Passenger Cars* found that referring to the relevant obligations in the published determinations but in a different context (in that case, when discussing causation) was not sufficient. (Panel Report, *Ukraine – Passenger Cars*, paras. 7.96-7.99).

³¹¹ See paras. 7.155-7.156 above.

³¹² European Union's first written submission, para. 66.

³¹³ European Union's response to Panel question No. 24(c), para. 14 (referring to Definitive determination, (Exhibit TUR-5), recitals 152, 156, and 188). (fn omitted)

to be able to apply a safeguard measure, and those obligations must therefore be identified in the Member's published report as part of the "pertinent issues of fact and law".³¹⁴

7.168. Although the European Union suggests that other Members, on reviewing the safeguard, can understand which obligations of the GATT 1994 the European Union decided to suspend, and therefore which obligations had the effect of stopping the European Union from preventing or remedying serious injury, we do not think that it would be appropriate to require other Members to carry out this interpretive exercise, when the European Union was better positioned to identify, and should have identified, the relevant obligations itself. In this regard, we note that the Member that finds that certain obligations of the GATT 1994 constrain its ability to prevent or remedy injury from an increase in imports, and that suspends those obligations as a result, is best placed to identify with certainty those obligations. Conversely, the Members affected by the suspension should not be required to do so, second-guessing the Member adopting the safeguard.

7.169. Given this, we reject the European Union's suggestion that the European Commission's references to a "free of duty quota" (to which the out-of-quota duty does not apply) in the sections of the definitive determination discussing the form, level, and duration of the measures³¹⁵ was sufficient to identify the relevant obligations of the GATT 1994.

7.170. We therefore find that the definitive safeguard is inconsistent with Article XIX:1(a) because the European Commission did not identify in its published reports the obligations whose effect resulted in the injurious increase in imports. Having so found, we do not proceed to examine Turkey's further argument that the European Commission did not explain how the obligations in question resulted in the injurious increase in imports.

7.5 Increase in imports

7.171. Turkey claims that the European Commission acted inconsistently with Articles 2.1 and 4.2(a) of the Agreement on Safeguards and Article XIX:1(a) of the GATT 1994 because it failed to make reasoned and adequate findings with respect to its determinations regarding an increase in imports, including with respect to alleged increases in certain product categories and product families.³¹⁶ The European Union responds that: (a) the European Commission was not required to examine increases by reference to individual product categories and product families, but rather the examination of the product as a whole was the "relevant determination"³¹⁷; and (b) the examination of the product as a whole showed a sharp increase in imports of 59% between 2013 and 2016 followed by a continued increase after this timeframe, which reached 71% overall at the end of the most recent period (MRP), namely mid-2017 to mid-2018.³¹⁸

7.172. We begin by setting out the applicable provisions forming the basis of Turkey's claims (section 7.5.1). We then evaluate Turkey's arguments that the European Commission did not provide a reasoned and adequate explanation for this determination due to decreases in import levels for some individual product categories and product families (section 7.5.2.1), and that the European Commission did not show that the increase was sufficiently significant, sudden, sharp, and recent (section 7.5.2.2).

 $^{^{314}}$ Agreement on Safeguards, Article 3.1; see also para. 7.164 above and Appellate Body Report, US – Lamb, paras. 72 and 76.

³¹⁵ See fn 313 above.

³¹⁶ Turkey's first written submission, para. 159. Turkey also argues that the definitive determination is inconsistent with Articles 2.1 and 4.2(a) and Article XIX:1(a) because the European Commission did not determine that there had been the requisite increase in imports for each product category, and took an inconsistent approach to the product concerned at different steps of its analysis. (Turkey's first written submission, paras. 170-173). We have already examined and rejected these arguments in section 7.2 above, and therefore we do not discuss them again in this section.

³¹⁷ European Union's first written submission, para. 161.

³¹⁸ European Union's first written submission, paras. 154, 157, 159, 161, 171, 180, and 194.

- 52 -

7.5.1 The applicable requirements of Article XIX:1(a) of the GATT 1994 and Articles 2.1 and 4.2(a) of the Agreement on Safeguards

7.173. Turkey claims that the European Commission's determination of an increase in imports fails to satisfy the requirements of Articles 2.1 and 4.2(a) of the Agreement on Safeguards and Article XIX:1(a) of the GATT 1994. Article XIX:1(a) of the GATT 1994 provides, in relevant part:

If, as a result of unforeseen developments and of the effect of the obligations incurred by a contracting party under this Agreement, including tariff concessions, any product *is being imported* into the territory of that contracting party *in such increased quantities* and under such conditions as to cause or threaten serious injury to domestic producers ... [.]³¹⁹

7.174. Article 2.1 of the Agreement on Safeguards provides, in relevant part:

A Member may apply a safeguard measure to a product only if that Member has determined, pursuant to the provisions set out below, that such *product is being imported into its territory in such increased quantities, absolute or relative to domestic production*, and under such conditions as to cause or threaten to cause serious injury to the domestic industry ... [.]³²⁰

7.175. Article 4.2(a) of the Agreement on Safeguards provides, in relevant part:

In the investigation to determine whether increased imports have caused or are threatening to cause serious injury to a domestic industry under the terms of this Agreement, the competent authorities shall evaluate all relevant factors of an objective and quantifiable nature having a bearing on the situation of that industry, in particular, *the rate and amount of the increase in imports of the product concerned in absolute and relative terms* ... [.]³²¹

7.176. We proceed now to examine Turkey's claims in relation to these provisions.

7.5.2 Whether Turkey has established that the definitive safeguard is inconsistent with Article XIX:1(a) of the GATT 1994 and Articles 2.1 and 4.2(a) of the Agreement on Safeguards

7.5.2.1 Whether the European Commission did not adequately account for decreases in imports in the most recent past

7.177. Turkey argues that the European Commission failed to provide a reasoned and adequate explanation for its finding of an increase in imports in two main ways. First, with respect to product categories, Turkey argues that the European Commission only compared imports at the beginning and end of the POI.³²² Second, Turkey argues that the European Commission did not properly account for the fact that imports of certain individual product categories and product families decreased during the most recent part of the POI.³²³ The European Union responds that Turkey's contention incorrectly focuses on a limited number of individual product categories and product families, whereas the "relevant determination" analysed the product under investigation as a whole.³²⁴ For the European Union, the European Commission's analysis of the product categories and product families was merely "additional" to this "relevant determination".³²⁵ The European Union also asserts that nothing in the GATT 1994 or in the Agreement on Safeguards requires the "increase" to manifest as a continuous acceleration of imports.³²⁶ Brazil, as a third party, argues that "investigating authorities must analyse trends over the entire POI with a particular focus on the

³¹⁹ Emphasis added.

³²⁰ Emphasis added; fn omitted.

³²¹ Emphasis added.

³²² Turkey's first written submission, paras. 174-181.

³²³ Turkey's first written submission, para. 182.

³²⁴ European Union's first written submission, paras. 157-161.

³²⁵ European Union's first written submission, para. 161.

³²⁶ European Union's first written submission, paras. 167-170 and 178-179.

most recent data in order to comply with the requirement expressed by the phrase 'is being imported' present in Article 2.1 of the Agreement on Safeguards".³²⁷

7.178. In light of our findings in section 7.2.2 above, we consider it significant that this argument by Turkey pertains only to the European Commission's examination of increases in imports based on certain individual product categories and product families. Turkey has not presented an equivalent challenge in respect of the European Commission's examination of increases in imports based on the product under consideration as a whole. In its examination of the product as a whole, the European Commission found that "[i]mports increased in absolute terms by 71% during the period of analysis, and in relative terms with market shares increasing from 12,7% to 18,8%" and that "[t]he most significant increase took place in the period 2013-2016" whereupon "[s]ubsequently, imports continued to increase at a slower pace before picking up again in the MRP".³²⁸ Thus, for the product as a whole, the European Commission examined these trends and found that import volumes continuously increased at varying rates.

7.179. Accordingly, the European Commission's determination sets out a basis for its finding of an increase in imports that has not been challenged by Turkey, namely its finding on the basis of the product as a whole, comprising all the product categories on which the European Commission applied the safeguard. The European Commission characterized its examination by reference to the individual product categories and product families as "confirm[atory]", "complement[ary]", and "supplementa[I]" *vis-à-vis* its examination of the product as a whole.³²⁹ This is also reflected in the general structure and order of the European Commission's analysis.

7.180. In this context, we note that Turkey's argument is based on the fact that for certain product categories and product families imports peaked either in 2016 or 2017, after which they decreased somewhat, although remaining at a higher level than at the beginning of the POI.³³⁰ The European Commission considered that the overall increase for the included product categories and product families confirmed the findings relating to the product as a whole.³³¹ The European Commission also observed that although for some product families the greatest increase took place between 2013 and 2016, with imports peaking and then decreasing to an extent, those imports still remained "at a much higher level" than at the beginning of the POI.³³²

7.181. Taking into account that, as set out above, the central analysis conducted by the European Commission related to the product as a whole, the analysis of product categories and product families was supplemental, and this granular analysis still disclosed trends that were similar to those observed for the product as a whole, we do not consider that Turkey has established that the European Commission did not provide a reasoned and adequate explanation of its finding of increase in imports in light of the fluctuations in import levels observed for certain individual product categories and product families.

7.5.2.2 Whether the European Commission did not demonstrate that the increase in imports was sufficiently sharp, significant, sudden, and recent

7.182. Turkey contends that the European Commission "failed to demonstrate an increase in imports that is sharp enough, significant enough, sudden enough and recent enough" as required by Articles 2.1 and 4.2(a) of the Agreement on Safeguards and Article XIX:1(a) of the GATT 1994.³³³ The European Union responds that there are no "absolute standards" governing the requisite rate of increase and, accordingly, the sufficiency of an increase cannot be examined "in the abstract" without considering the situation of the domestic industry.³³⁴ Brazil, as a third party, argues that the "standard for the determination of the increase of imports 'in such quantities'" has been "elucidated by previous case law", namely that the increase of imports must have been recent enough, sudden

(Exhibit TUR-5), recitals 35-36 and 39.

³²⁷ Brazil's third-party submission, para. 14.

³²⁸ Definitive determination, (Exhibit TUR-5), recital 33.

³²⁹ Definitive determination, (Exhibit TUR-5), recitals 19, 22, 28, and 34.

³³⁰ Turkey's first written submission, tables 3-11. See also ibid. table 14.

³³¹ Provisional determination, (Exhibit TUR-3), recitals 23 and 28; Definitive determination,

³³² Definitive determination, (Exhibit TUR-5), recital 36.

³³³ Turkey's first written submission, paras. 169 and 197.

³³⁴ European Union's first written submission, paras. 173-174; second written submission para. 48.

enough, sharp enough, and significant enough, both quantitatively and qualitatively to cause or threaten to cause serious injury. 335

7.183. In our view, if a complainant in WTO dispute settlement alleges that the increase in imports found by an authority is not sudden, significant, sharp, or recent *enough*³³⁶, it bears the onus of adducing the reference point or benchmark against which to ascertain what may or may not be "enough". This is because the concept of "enough" is relational. Something is "enough" in relation to something else. What qualifies as "enough" cannot exist in the abstract untethered from some reference point or benchmark for ascertaining what is "enough".

7.184. We note that our understanding in this regard accords with that of the panel in US – Steel Safeguards, which observed that the question of whether, in that instance, an increase in imports was "sudden, sharp and significant *enough* as to cause serious injury is a question that is appropriately to be addressed in the context of causation of serious injury, not in the context of the condition of the increase, where no well-founded judgment in this regard can be made".³³⁷

7.185. Thus, the question of whether an increase is sudden, significant, sharp, and recent enough may be determined by evaluating whether the increase is sufficient to cause or threaten to cause serious injury to the domestic industry. This question may also conceivably be determined by reference to some other benchmark proposed by the complainant, such as whether the increase is merely reflective of a reversion to normal market conditions after a disruption.³³⁸

7.186. In the present case, having structured its claim around whether the increase was sudden, significant, sharp, or recent enough, we do not consider that Turkey has adduced a viable benchmark or reference point to ascertain what was "enough". In its first written submission, Turkey refers to another panel's finding that a 37.9% relative increase in imports was not "enough", but that reference point has no relevance to the present case involving a different product, market, and set of facts.³³⁹ Turkey also refers to the imports increasing by only 7% in the present case between 2017 and the MRP, and to the most recent data in the POI evincing only a 3% or 4% year-on-year increase³⁴⁰, but Turkey does not explain *why* these rates of increase were not "enough".

7.187. We asked Turkey the following question on this point:

The European Union contends that the sufficiency of increased imports cannot be assessed "*in the abstract*" but [the assessment] must instead be made in relation to "*other conditions necessary for imposition of a safeguard*" (European Union's first written submission, para. 174). Is it Turkey's view that the Panel should consider Turkey's claims regarding increased imports in isolation, and if so, what would be the benchmark for determining whether the increase was sufficient?³⁴¹

7.188. Turkey responded that "the condition of 'increase in imports' should not be seen in isolation but relating to the other conditions necessary for the imposition of safeguard measures".³⁴² Despite this, Turkey's submissions in relation to the present claim do not address *why* the increase in imports identified by the European Commission was not sudden, significant, sharp, or recent enough to

³³⁵ Brazil's third-party submission, para. 12.

³³⁶ We note that these terms are not found in the text of Articles 2.1 and 4.2(a) of the Agreement on Safeguards and Article XIX:1(a) of the GATT 1994, but are rather derived from prior DSB reports. In view of our ultimate conclusion in this section, we need not express a view on the extent to which these terms correspond to the textual requirements of those provisions.

³³⁷ Panel Reports, *US – Steel Safeguards*, paras. 10.217, 10.226, 10.236, and 10.255 (original emphasis omitted; emphasis added); see also Appellate Body Report, *US – Steel Safeguards*, paras. 351, 358, and 360 (noting that "[t]he question whether 'such increased quantities' of imports will suffice as 'increased imports' to justify the application of a safeguard measure is a question that can be answered only in the light of 'such conditions' under which those imports occur" and cannot be determined by reference to "absolute standards" nor "in the abstract").

³³⁸ Panel Report, Argentina – Preserved Peaches, para. 7.60.

³³⁹ Turkey's first written submission, para. 198 (referring to Panel Report, *Ukraine – Passenger Cars*, para. 7.147).

³⁴⁰ Turkey's first written submission, paras. 204-205.

³⁴¹ Panel question No. 10. (emphasis original)

³⁴² Turkey's response to Panel question No. 10, para. 34.

threaten to cause serious injury.³⁴³ Turkey points to characteristics exhibited by the increase itself. For instance, Turkey refers to the 3% year-on-year rate of increase at the end of the POI. Turkey also considers it significant that the bulk of the increase in imports found by the European Commission took place earlier in the POI and subsequently decelerated.³⁴⁴ However, Turkey does not link these characteristics to the situation of the domestic industry to show that the increase was not sudden, significant, sharp, or recent enough to threaten to cause serious injury.

7.189. We therefore find that Turkey has not established that the European Commission did not demonstrate a sufficient increase in imports under Articles 2.1 and 4.2(a) of the Agreement on Safeguards and Article XIX:1(a) of the GATT 1994. We note, however, that Turkey has presented other claims in these proceedings that encompass the impact of increased import volumes on the situation of the domestic industry, which we address elsewhere.

7.5.2.3 Overall conclusion on the European Commission's determination of an increase in imports

7.190. Based on the above, we find that Turkey has not established that the European Commission acted inconsistently with Article XIX:1(a) of the GATT 1994 and Articles 2.1 and 4.2(a) of the Agreement on Safeguards in its assessment of an increase in imports.

7.6 Threat of serious injury

7.191. Turkey claims that the definitive safeguard is inconsistent with Articles 4.1(a), 4.1(b), and 4.2(a) of the Agreement on Safeguards because the European Commission improperly determined that increased imports were threatening to cause serious injury to the domestic industry.³⁴⁵ Turkey argues that the European Commission neither demonstrated the existence of a significant overall impairment to the domestic industry that was clearly imminent, nor the existence of a high degree of likelihood of serious injury in the very near future.³⁴⁶

7.192. Turkey also argues that purely as a consequence of these inconsistencies with Articles 4.1(a), 4.1(b), and 4.2(a), the definitive safeguard is also inconsistent with Article 2.1 of the Agreement on Safeguards and Article XIX:1(a) of the GATT 1994.³⁴⁷

7.193. The European Union responds that the European Commission's finding of a threat of serious injury to the domestic industry was based on facts and Turkey has not demonstrated an inconsistency with the provisions it invokes.³⁴⁸

7.194. We begin by setting out the applicable requirements of Articles 4.1(a), 4.1(b), and 4.2(a) of the Agreement on Safeguards (section 7.6.1). We then evaluate whether Turkey has demonstrated that the European Commission acted inconsistently with these provisions by not demonstrating that

³⁴³ Turkey's response to Panel question No. 10, paras. 34-40; first written submission, paras. 197-213; and second written submission, paras. 132 and 136.

³⁴⁴ Turkey's second written submission, paras. 131-134; response to Panel question No. 10, paras. 35-40.

³⁴⁵ Turkey's first written submission, para. 215. When presenting its arguments on this point, Turkey did not clearly distinguish its arguments under Articles 4.1(a), 4.1(b), and 4.2(a), respectively, but rather mostly advanced a single set of arguments in respect of those provisions collectively.

³⁴⁶ Turkey's first written submission, paras. 231 and 245.

³⁴⁷ Further, Turkey argues that the definitive safeguard is inconsistent with Articles 2.1, 4.1(a), 4.1(b), and 4.2(a) and Article XIX:1(a) for the reasons already discussed in section 7.2 above, i.e. that, according to Turkey, the European Commission was required to determine that each of the conditions and circumstances to impose a safeguard was met for each product category individually. (Turkey's first written submission, paras. 226-229 and 254-257). Having already addressed these arguments in section 7.2, we do not discuss them again in this section.

We note that at paras. 226-229 and 254-257 of its first written submission, Turkey also relies on these arguments in support of a claim under Article 4.1(b). We have not addressed Article 4.1(b) directly in section 7.2, but these arguments rely on the same basic proposition that we have rejected in section 7.2, namely that the European Commission was required to determine that each of the conditions and circumstances to impose a safeguard was met for each product category individually. We therefore reject these arguments under Article 4.1(b). We note that we find that the definitive safeguard is otherwise inconsistent with Article 4.1(b).

³⁴⁸ European Union's first written submission, paras. 222 and 225; opening statement at the first meeting of the Panel, para. 79.

a significant overall impairment was clearly imminent (section 7.6.2.1), and that serious injury would be highly likely to materialize in the very near future (section 7.6.2.2). Last, we address Turkey's consequential arguments (section 7.6.2.3).

7.6.1 The applicable requirements of Articles 4.1(a), 4.1(b), and 4.2(a) of the Agreement on Safeguards

7.195. Turkey contends that the European Commission's determination of a threat of serious injury is inconsistent with the requirements set out in Articles 4.1(a), 4.1(b), and 4.2(a) of the Agreement on Safeguards.³⁴⁹

7.196. Article 4.1(a) defines "serious injury" as follows:

"[S]erious injury" shall be understood to mean a significant overall impairment in the position of a domestic industry[.]

7.197. Article 4.1(b), which defines "threat of serious injury" and sets out certain requirements for a determination of such threat, reads:

"[T]hreat of serious injury" shall be understood to mean serious injury that is clearly imminent, in accordance with the provisions of [Article 4.2]. A determination of the existence of a threat of serious injury shall be based on facts and not merely on allegation, conjecture or remote possibility[.]

7.198. Thus a competent authority making a determination of a threat of serious injury must establish (i) the clear imminence of (ii) a significant overall impairment in the position of a domestic industry.

7.199. Article 4.2(a) of the Agreement on Safeguards provides as follows:

In the investigation to determine whether increased imports have caused or are threatening to cause serious injury to a domestic industry under the terms of this Agreement, the competent authorities shall evaluate all relevant factors of an objective and quantifiable nature having a bearing on the situation of that industry, in particular, the rate and amount of the increase in imports of the product concerned in absolute and relative terms, the share of the domestic market taken by increased imports, changes in the level of sales, production, productivity, capacity utilization, profits and losses, and employment.

7.200. We proceed now to examine Turkey's claims under these provisions.

7.6.2 Whether Turkey has established that the definitive safeguard is inconsistent with Articles 4.1(a), 4.1(b), and 4.2(a) of the Agreement on Safeguards

7.6.2.1 Whether the European Commission did not provide a reasoned and adequate explanation of its determination of threat of serious injury

7.201. Turkey claims that the European Commission failed to provide a reasoned and adequate explanation for its finding that the domestic industry was "in a fragile and vulnerable position", given that the data indicate that the industry's performance had improved towards the end of the POI.³⁵⁰ Turkey argues that the very fact of this improved performance precludes a "threat of serious injury" determination.³⁵¹ Turkey also argues that the European Commission failed to adequately explain how this improved performance came about and why it would be short-lived in the face of further imports.³⁵² Because the finding of the domestic industry as "fragile and vulnerable" despite its improved performance was at the heart of the European Commission's determination of a threat of

³⁴⁹ Turkey's first written submission, para. 225.

³⁵⁰ Turkey's first written submission, paras. 231-232, 239, and 241-243.

³⁵¹ Turkey's response to Panel question No. 26, para. 13. See also Turkey's opening statement at the second meeting of the Panel, para. 12; and first written submission, para. 244.

³⁵² Turkey's response to Panel question No. 12, paras. 41, 44-45, and 50; comments on the European Union's response to Panel question No. 27, paras. 18-24.

serious injury, Turkey argues that the European Commission failed to properly reach this determination under Articles 4.1(a), 4.1(b), and 4.2(a) of the Agreement on Safeguards.³⁵³

7.202. The European Union responds that the European Commission acknowledged the domestic industry's improved performance but found it to be partial, temporary, and readily reversible, such that the domestic industry remained vulnerable to a continued increase in imports.³⁵⁴ For the European Union, the domestic industry's position was fragile in view of the continued downward trends in the domestic industry's market share, employment, and stocks, despite improvements in other factors.³⁵⁵ The European Union contends that the domestic industry's deterioration in 2013-2016 illustrates its vulnerability to increased imports, and that the post-2017 data indicate that import levels were accelerating due to, *inter alia*, the Section 232 measures, and that prices were also starting to decline.³⁵⁶

7.203. Brazil, as a third party, argues that while not every injury factor needs to be declining at the end of the POI, positive trends must be explained, particularly given that the most recent data will provide the most reliable indication of the situation of the domestic industry in the near future.³⁵⁷

7.204. Our evaluation of Turkey's claims and the European Union's rebuttal proceeds as follows. We first address Turkey's contention that the very fact that the domestic industry performance improved precludes a "threat of serious injury" determination (section 7.6.2.1.1). We then assess whether the European Commission properly examined the basis for the domestic industry's improved performance, which in turn sheds light on whether the European Commission provided a sufficient basis for inferring that these improvements would be temporary and readily reversible in the face of increasing imports (section 7.6.2.1.2). We next assess Turkey's contention that, even assuming the European Commission did indeed properly attribute the domestic industry's improved performance to the AD/CV measures, it nonetheless erred in inferring that these improvements would be temporary and readily reversible (section 7.6.2.1.3). Finally, we assess the role of the other factors that allegedly explained the domestic industry's improved performance (section 7.6.2.1.4).

7.6.2.1.1 Whether any improvement in the domestic industry's situation necessarily negates a finding of threat of serious injury

7.205. Turkey argues that the fact that the domestic industry improved towards the end of the POI demonstrates that serious injury was not "clearly imminent", irrespective of the reasons underlying the improvement.³⁵⁸ The European Union responds that the fact that the domestic industry improved during this period did not preclude a determination of a threat of serious injury in view of the European Commission's finding that the domestic industry nonetheless remained "fragile" and "vulnerable".³⁵⁹

7.206. We are of the view that the Agreement on Safeguards does not establish a categorical rule that precludes authorities from finding a threat of serious injury whenever the data show positive trends in the domestic industry's performance at a given point of the POI. Rather, Article 4.1(b) provides that "'threat of serious injury' shall be understood to mean serious injury that is clearly imminent". In our view, such assessments are necessarily fact-specific and to be made on a case-by-case basis. If the evidence in a proceeding shows that a domestic industry's performance is on a durable upward trajectory, it may be difficult to establish that serious injury is "clearly imminent". However, if the evidence indicates that serious injury is "clearly imminent" despite certain improvements in the domestic industry's performance, an authority may find a threat of serious injury. For instance, the evidence may show that the improvement is fleeting or illusory and that the domestic industry will continue to be at a competitive disadvantage *vis-à-vis* imports once the short-lived improvement recedes. Accordingly, the question of whether a determination of a

³⁵³ Turkey's first written submission, paras. 225, 231, and 244.

³⁵⁴ European Union's first written submission, para. 202.

³⁵⁵ European Union's first written submission, paras. 204-205.

³⁵⁶ European Union's first written submission, paras. 203, 205, and 213-215; opening statement at the first meeting of the Panel, paras. 80 and 83; and opening statement at the second meeting of the Panel, paras. 17 and 31.

³⁵⁷ Brazil's third-party submission, paras. 23-25.

³⁵⁸ Turkey's response to Panel question No. 26, para. 13. See also Turkey's opening statement at the second meeting of the Panel, para. 12; and first written submission, para. 244.

³⁵⁹ European Union's comments on Turkey's response to Panel question No. 26, paras. 12-14.

threat of serious injury complies with the Agreement on Safeguards when the domestic industry's performance improves at some point during the POI will depend on the facts and the quality of the authority's explanation as to the nature and implications of those improvements.

7.6.2.1.2 Whether the European Commission adequately explained the basis for the domestic industry's improved performance

7.207. We turn now to whether the European Commission adequately explained the basis for the domestic industry's improved performance, such that it could infer that this improvement was partial and transitory in the face of increasing imports and thus did not negate a finding of a threat of serious injury. We address two main points of contention in this section. The first point of contention concerns the parties' differing understandings of the European Commission's findings on why the domestic industry's performance improved. The second point of contention concerns whether the European Commission properly identified and substantiated that the AD/CV measures³⁶⁰ that were adopted during the POI were a factor that contributed to the domestic industry's improved performance.

7.208. The improvement in the domestic industry's performance towards the end of the POI is a matter of fact that is uncontested between the parties. Rather, the parties dispute whether the European Commission could reasonably infer that this improvement was "partial", "temporary", and readily "revers[ible]"³⁶¹, and whether, on that basis, the European Commission reasonably determined that the situation of the domestic industry was "fragile and vulnerable".³⁶² Given that the parties offer differing understandings of the European Union's findings on why the domestic industry's performance improved, we begin our evaluation by examining the relevant sections of the European Commission's determination on this point.

7.209. The European Commission found that the situation of the domestic industry had "deteriorated significantly" in the period 2013-2016, before "recover[ing] partially" in 2017.³⁶³ In particular, the European Commission observed both positive trends for the domestic industry from 2016 to 2017 in relation to domestic sales, consumption, production, production capacity, capacity utilization, prices, profitability, and cash flow, as well as negative trends in market share, stocks, and employment.³⁶⁴ The determination also sets out data indicating that the domestic industry improved its position relative to imports from 2016 to 2017 in terms of the respective rates at which it captured new demand and stemmed the loss of market share to imports.³⁶⁵ Although the European Commission indicated that the data for the first semester of 2018 could not be verified, the European Commission noted that these data confirmed that the "partial recovery" of the domestic industry was ongoing.366

7.210. The parties diverge as to what factors informed the European Commission's analysis of the domestic industry's improved performance from 2016 to 2017. The European Union contends that the European Commission identified five factors that explained this improvement, namely increased domestic consumption from 2013 to 2017, increased sales by the domestic industry, the cost of raw material remaining lower in 2017 than 2013, the recovery in prices to 2013 levels, and the imposition

³⁶⁰ Our reference to AD/CV measures in this section means those measures adopted by the European Union during the POI of the safeguard investigation that are mentioned in the provisional and definitive determinations. (See e.g. Provisional determination, (Exhibit TUR-3), recitals 52-53, 55-57, 59, 61, and 69; and Definitive determination, (Exhibit TUR-5), recitals 95 and 116). ³⁶¹ Definitive determination, (Exhibit TUR-5), recitals 87, 89, 90, 95, 97, and 114;

European Union's opening statement at the first meeting of the Panel, paras. 82-83; and Turkey's response to Panel question No. 26, paras. 16 and 20.

³⁶² European Union's opening statement at the second meeting of the Panel, paras. 17-19;

Turkey's second written submission, paras. 157, 161, and 175.

³⁶³ Definitive determination, (Exhibit TUR-5), recital 90.

³⁶⁴ Provisional determination, (Exhibit TUR-3), recitals 39-46; Definitive determination, (Exhibit TUR-5), recitals 67-73. These data pertain to the "global" level of analysis. Similar trends were observed by reference to the product families and product categories. (Definitive determination, (Exhibit TUR-5), recitals 73-86. See also Turkey's first written submission, para. 234; and European Union's first written submission, para. 211).

³⁶⁵ Definitive determination, (Exhibit TUR-5), recital 33 and table 4. See also Turkey's response to Panel question No. 26, para. 19; and European Union's second written submission, para. 75.

³⁶⁶ Definitive determination, (Exhibit TUR-5), recital 89; European Union's opening statement at the first meeting of the Panel, para. 83.

of AD/CV measures against several of the products at issue.³⁶⁷ The European Union also cites "efficiency gains by [the domestic] producers" in this regard.³⁶⁸ Turkey responds that three of these factors – prices, sales volume, and consumption – are descriptive rather than explanatory in nature, and in any event, neither consumption nor sales volume are mentioned in the European Commission's determination as factors explaining the domestic industry's improved performance.³⁶⁹

7.211. Based on our review of the European Commission's determination, we agree that consumption and sales volume do not appear to have been explicitly cited as factors contributing to the domestic industry's improved performance. At the provisional stage, the European Commission stated that "[p]rices recovered in 2017, given a general recovery of the steel market but also as a consequence of the various trade defence measures"; "some [product categories] have improved, most likely as a result of the recent imposition of anti-dumping and anti-subsidy measures"; and "the [domestic] steel industry has partially recovered for some product categories in 2017, notably due to trade defence measures".³⁷⁰ At the definitive stage, the European Commission affirmed its finding that "[t]his recovery was attributed, inter alia, to the effectiveness of the different trade defence measures that have been adopted, in particular since 2016".³⁷¹ Given this, the European Commission appears to have accorded particular weight to the AD/CV measures in stimulating the domestic industry's recovery, with "a general recovery of the steel market" also mentioned as a factor contributing to price increases. Additionally, the European Commission appears to have considered raw material costs to be a factor contributing to the domestic industry's improved profit levels. Specifically, the European Commission noted that "[t]he [domestic] industry achieved a level of profit of 6,2 % [in 2017] since cost of production (raw material), even if increasing, remained lower than in 2013".372

7.212. In summary, we understand the European Commission to have found that the domestic industry's recovery in 2017 arose principally from the application of AD/CV measures, in particular since 2016, and that lower raw material costs and a "general recovery of the steel market" also contributed to this recovery through improved profitability and increased prices. Increased prices were not themselves cited as a contributing factor. Rather, the European Commission treated increased prices as indicative of the improvements arising from the AD/CV measures and the general recovery of the market. Increased consumption and sales volumes were likewise not cited as contributing factors, but rather as indicative of the situation of the domestic industry in 2017.³⁷³

7.213. With this understanding in mind, we turn now to the parties' disagreement as to the proper inferences and conclusions to be drawn from the domestic industry's improved performance. The European Commission found that, in light of the nature of this improvement, the improvement was "partial", "temporary", and readily "revers[ible]"³⁷⁴, and it was on this basis that the European Commission justified its conclusion that the domestic industry was still "fragile" and "vulnerable" to a continued increase in imports despite the improvement.³⁷⁵ Turkey, by contrast, contends that the domestic industry was "performing outstandingly in 2017".³⁷⁶

7.214. Given the prominence of the AD/CV measures in the European Commission's explanation for the domestic industry's improved performance, we focus in this section on the parties' disagreement as to the proper inferences to be drawn from that factor in relation to the improved situation of the

³⁶⁷ European Union's second written submission, para. 55.

³⁶⁸ European Union's response to Panel question No. 26, para. 17.

³⁶⁹ Turkey's opening statement at the first meeting of the Panel, paras. 30-31; response to Panel question No. 26, para. 14.

³⁷⁰ Provisional determination, (Exhibit TUR-3), recitals 52, 55, and 69.

³⁷¹ Definitive determination, (Exhibit TUR-5), recital 95.

³⁷² Provisional determination, (Exhibit TUR-3), recital 45. See also Definitive determination,

⁽Exhibit TUR-5), recital 72. ³⁷³ Provisional determination, (Exhibit TUR-3), recitals 39-40. We note that the European Union argues that consumption and sales volume comprise part of the "general recovery of the [steel] market" (European Union's comments on Turkey's response to Panel question No. 26, para. 16 (emphasis original)). We address the role of the "general recovery of the steel market" as a contributing factor in section 7.6.2.1.4

below. ³⁷⁴ Provisional determination, (Exhibit TUR-3), recitals 55 and 69; Definitive determination,

⁽Exhibit TUR-5), recitals 87, 89, 90, 95, 97, and 114.

³⁷⁵ Definitive determination, (Exhibit TUR-5), recitals 87, 90, 95, and 114.

³⁷⁶ Turkey's first written submission, paras. 239 and 243-244.

domestic industry in 2017.³⁷⁷ The parties agree that this was the only factor identified by the European Commission to explain why the domestic industry improved its competitive relationship *vis-à-vis* imports after 2016.³⁷⁸ However, the parties disagree as to whether the European Commission's analysis provided a sufficient basis for finding that the AD/CV measures contributed to the domestic industry's improved performance in the first place.³⁷⁹

7.215. The European Commission explained the role of the AD/CV measures in improving the situation of the domestic industry, particularly by increasing profits to a "sustainable" level for certain product categories, as follows:

As far as profit is concerned, all product categories were sold at a loss or at a much reduced profit until 2016. Only 7 products could recover to a level of profit above 6 % in 2017. These products are significant in terms of EU production volume and six of them are currently subject to (recent) anti-dumping or countervailing duty measures. Note that these measures concern only some countries of origin. All other products remained either loss making (3 products) or only close to break-even (13 products).

...

For product categories 1, 2 and 4, the financial situation was negative in 2016, but became positive in 2017 following the imposition of anti-dumping and anti-subsidy measures against a number of countries like, amongst others, China and Russia.

....

In parallel, in order to remedy the injury caused by unfair trade imports, the Union has imposed a number of anti-dumping and anti-subsidy measures against imports of steel products. In total, there are currently no less than 19 anti-dumping or anti-subsidy measures against the unfairly traded imports of 14 product categories under investigation from various countries. During the period under investigation, i.e. 2013-2017, 13 new investigations determined that the EU steel industry suffered (or in one case was threatened to suffer) from material injury caused by unfair trade practices.³⁸⁰

7.216. Turkey contends that this analysis is deficient in a number of respects. First, Turkey argues that the European Commission failed to properly articulate its rationale for finding that the AD/CV measures contributed to the domestic industry's improved performance, because it did not identify the relevant measures and product categories concerned, nor did it explain how those measures impacted trends in injury factors.³⁸¹ Second, given this "lack of systematic analysis", Turkey contends that the European Commission drew incorrect conclusions.³⁸² In particular, Turkey asserts that the European Commission's focus on improvements in profitability for product categories subject to these measures ignored the improvements in profitability for product categories irrespective of the application of AD/CV measures show that those measures could not themselves explain the domestic industry's improved performance.³⁸⁴ The European Union responds that the domestic industry's improved performance "was due only in part" to the AD/CV measures, and that the improvement for products not subject to those measures "is explained by the general

³⁷⁷ We address other factors contributing to the domestic industry's improved performance in section 7.6.2.1.4 below.

³⁷⁸ European Union's response to Panel question No. 26, para. 16; Turkey's response to Panel question No. 26, paras. 16-20.

³⁷⁹ Turkey's responses to Panel question No. 26, para. 17; and No. 28(b), para. 24;

European Union's comments on Turkey's response to Panel question No. 28, paras. 26-29.

³⁸⁰ Provisional determination, (Exhibit TUR-3), recitals 53, 56, and 59.

³⁸¹ Turkey's comments on the European Union's response to Panel question No. 27, para. 23.

³⁸² Turkey's comments on the European Union's response to Panel question No. 27, para. 24.

³⁸³ Turkey's response to Panel question No. 26, para. 17.

³⁸⁴ Turkey's response to Panel question No. 26, para. 17; comments on the European Union's response to Panel question No. 27, para. 24.

recovery of the steel market".³⁸⁵ Moreover, the European Union emphasizes that product categories subject to AD/CV measures tended to return to sustainable levels of profitability, whereas those not subject to these measures tended to remain at unsustainable levels, despite improvements.³⁸⁶

7.217. We agree with Turkey that the European Commission's analysis of the role played by the AD/CV measures in the domestic industry's improved performance did not provide a sufficient basis to infer that the improvement was "temporary" and "revers[ible]".387 The European Commission's analysis contains insufficient detail to explain how imposing an AD/CV measure on a particular product at a particular point in time led to a subsequent improvement in the trends for the category encompassing that product. For instance, as evidenced in the portions extracted above (see paragraph 7.215), the analysis does not provide any information about the product coverage of a given AD/CV measure, the proportion of a given category that was impacted by that measure, the point in time at which that measure was imposed, and the subsequent improvements in trends for that category. We do not suggest that all such information is necessarily required in all instances. Rather, such information is illustrative of the kind of details that could permit an understanding of how the AD/CV measures contributed to the domestic industry's improved performance. As it stands, the information underpinning the European Commission's analysis on this point is sparse and incomplete and thus, standing alone, is insufficient to support the European Commission's inference that the improvement was temporary and reversible. This aspect of the European Commission's explanation falls short of the Article 4.1(b) requirement that "threat of serious injury" determinations be "based on facts".

7.218. We also consider it significant that increases in profitability were observed in 2017 irrespective of the adoption of AD/CV measures. According to the European Commission, significant price depression from imports had precluded the domestic industry from achieving sustainable profit levels in 2013-2016 despite falling raw material costs.³⁸⁸ Based on this, the European Commission reasoned that the subsequent improvement in profit levels for some product categories in 2017 was "most likely as a result of the recent imposition of anti-dumping and anti-subsidy measures".³⁸⁹ According to this reasoning, the adoption of AD/CV measures alleviated the downward price pressure from certain imports, which in turn enabled the domestic industry to benefit from the lower (albeit increasing) raw material costs and to fetch prices that were commensurate to sustainable profit levels whilst passing increased costs on to customers.³⁹⁰

7.219. However, as Turkey has noted, increases in profitability were also observed in 2017 for product categories for which AD/CV measures had *not* been recently adopted.³⁹¹ For instance, profit levels increased in 2016-2017 from 4.9% to 9.2% for category 8, from 2.3% to 5.8% for category 14, and from -3.1% to 3.9% for category 15.³⁹² These increases can be juxtaposed against those explicitly referenced by the European Commission as exhibiting improvements in 2017 following the imposition of AD/CV measures, namely categories 1, 2, and 4.³⁹³ The profit levels for these categories improved from 2016 to 2017 as follows: from -1.0% to 7.8% for category 1, from 0.6% to 9.8% for category 2, and from 7.9% to 11.7% for category 4.³⁹⁴ Meanwhile, profit levels for category 6 decreased in 2016-2017 from 4.6% to 3.1% despite the recent adoption of

 $^{^{385}}$ European Union's comments on Turkey's response to Panel question No. 28(b), paras. 25 and 29. As we explained in para. 7.214, we focus on the role of the AD/CV measures in the present section given the prominence of this factor in the European Commission's determination and given that this was the only factor identified by the European Commission that could explain the domestic industry's improved performance *vis-à-vis* imports. We address other factors contributing to the domestic industry's improved performance in section 7.6.2.1.4 below.

³⁸⁶ European Union's comments on Turkey's response to Panel question No. 26, para. 22.

³⁸⁷ Turkey's comments on the European Union's response to Panel question No. 27, para. 21.

³⁸⁸ Provisional determination, (Exhibit TUR-3), recital 55.

³⁸⁹ Provisional determination, (Exhibit TUR-3), recital 55.

³⁹⁰ European Union's second written submission, para. 62.

³⁹¹ Turkey's response to Panel question No. 26, para. 17.

³⁹² Provisional determination, (Exhibit TUR-3), annex III. The European Commission's determination does not enumerate those product categories affected by the imposition of AD/CV measures prior to 2017 or generally. Turkey has provided a table in this regard in Exhibit TUR-49. In the absence of any objections by the European Union as to its accuracy, we are guided by the contents of that table in understanding which product categories were affected by these measures, and when.

³⁹³ Provisional determination, (Exhibit TUR-3), recital 56.

³⁹⁴ Provisional determination, (Exhibit TUR-3), annex III.

anti-dumping measures in 2016. These data do not support the proposition that the fact and degree of improvement in profit levels in 2017 depended on the imposition of recent AD/CV measures.³⁹⁵

7.220. Accordingly, the domestic industry's ability to mitigate downward price pressure from imports and realize increased profit levels through higher prices in 2017 was not limited to the product categories recently subject to AD/CV measures. Rather, the domestic industry's improved competitiveness vis-à-vis imports was observable across other product categories too.³⁹⁶ Indeed, as Turkey points out, the domestic industry's improved competitiveness vis-à-vis imports in 2017 was also reflected in other data.³⁹⁷ In particular, domestic consumption grew 12% from 2013 to 2016.³⁹⁸ Imports outperformed the domestic industry in capturing this growth insofar as imports increased by 59%, compared to domestic sales which increased by just 5% during that period.³⁹⁹ By contrast, in 2016-2017 the domestic industry managed to substantially reduce the rate at which it was being outperformed by imports in capturing new demand.400 This provides a further indication that the domestic industry had improved its competitiveness vis-à-vis imports in 2017 following the prior "significant deterioration"⁴⁰¹ in 2013-2016.

7.221. As we have noted, the European Union acknowledged that the imposition of the AD/CV measures was the only factor identified by the European Commission to explain the domestic industry's improved competitiveness vis- \dot{a} -vis imports.⁴⁰² This factor, however, cannot explain the domestic industry's improved competitiveness in relation to product categories for which no AD/CV measures had been recently adopted. The absence of an explanation in this regard is noteworthy. First, the fact that the domestic industry could mitigate the price pressure from imports and improve profitability across a range of product categories in 2016-2017 irrespective of the application of AD/CV measures raises the possibility that something other than those measures was driving the improved competitiveness.⁴⁰³ Nothing before us suggests that such a possibility was explored by the European Commission.⁴⁰⁴ Second, even accepting that the AD/CV measures were

³⁹⁵ The European Union seeks to differentiate between those product categories whose profits improved to a "healthy" or "sustainable" profit level and that were subject to recent AD/CV measures, and those whose profits levels improved but failed to reach "healthy" or "sustainable" levels and that were not subject to such measures. We find this unconvincing. The European Union oscillates between 6% (European Union's response to Panel question No. 28(b), para. 27) and 8% (response to Panel question No. 27, para. 21; comments on Turkey's response to Panel question No. 26, para. 22) as the benchmark for the requisite level of profit without any explanation. We note that the European Commission stated that "[i]t is considered that the level profit below 6 % is insufficient to cover the investments needed to sustain the activity, as, in the majority of the recent investigations, the Commission has used a level of around 8 % profit as a sufficient profit level in this sector in order to cover investments". (Provisional determination, (Exhibit TUR-3), recital 53). We would expect to see additional explanation if this was indeed a pivotal aspect of the determination. We agree with Turkey in this regard that the European Commission did not adequately explain its use of these benchmarks as indicative of a "sustainable" profit level. (Turkey's comments on the European Union's response to Panel question 28(b). para. 27). Indeed, several interested parties argued during the investigation that the European Commission had found profit levels of between 3-7% to be adequate in a number of AD/CV investigations concerning the steel sector, but the European Commission did not respond directly to this claim. (Definitive determination, (Exhibit TUR-5), recitals 96-97). In any case, contrary to the European Union's argument, at least one category not subject to AD/CV measures exceeded the 8% threshold (category 8), whereas some categories that were subject to those measures fell below the 6% threshold (categories 5, 7, and 12). These data-points call into question the premise of the European Union's alleged distinction between products subject to AD/CV measures reaching healthy profit levels on the one hand, and those not subject to such measures exhibiting unsustainable profit levels, on the other hand.

³⁹⁶ We also note that *downward* trends in profitability from 2016-2017 were not limited to product categories not subject to AD/CV measures. For instance, profit levels for category 6 decreased in 2016-2017 from 4.6% to 3.1% despite the recent adoption of anti-dumping measures in 2016.

³⁹⁷ Turkey's response to Panel question No. 26, paras. 19-20. See also European Union's second written submission, para. 75.

³⁹⁸ Definitive determination, (Exhibit TUR-5), table 4.

³⁹⁹ Definitive determination, (Exhibit TUR-5), tables 2 and 4.

⁴⁰⁰ Definitive determination, (Exhibit TUR-5), recital 33, and tables 2 and 4.

⁴⁰¹ Definitive determination, (Exhibit TUR-5), recitals 90 and 95.

⁴⁰² European Union's response to Panel question No. 26, para. 16. According to the European Union, the other factors, namely the "general recovery of the steel market" and the lower raw material costs, benefitted both the domestic industry and imports.

⁴⁰³ Turkey's responses to Panel question No. 12, para. 44; No. 26, paras. 17 and 20; and No. 28(b), para. 24. ⁴⁰⁴ European Union's response to Panel question No. 12, paras. 45-48. The

European Commission's reference to a "general recovery of the steel market" and lower raw material costs

responsible for the domestic industry's improved competitiveness *vis-à-vis* imports in certain product categories, the European Commission does not appear to have addressed the industry's improved competitiveness in other categories.⁴⁰⁵ Its determination therefore leaves unexplained why the domestic industry began to gain ground on imports from 2016 to 2017 with respect to the imports of product categories not subject to the AD/CV measures. In the absence of any understanding as to why the domestic industry's performance improved *vis-à-vis* imports of those particular product categories, it was not reasonable for the European Commission to surmise that the industry nonetheless continued to be "vulnerable" to those imports. Moreover, the European Commission did not explain how its finding that the domestic industry was "vulnerable" was compatible with seemingly contradictory data cited in its determination. In particular, the domestic industry's performance improved after 2016 even as imports that were not subject to AD/CV measures continued to increase. The domestic industry's improvement in the face of increased volumes of imports not subject to AD/CV measures calls into question the European Commission's finding that the domestic industry is performent in the face of increased volumes of imports not subject to AD/CV measures calls into question the European Commission's finding that the domestic industry was "vulnerable" to increases of those imports, and thus warranted further explanation.

7.222. We recall, in this regard, that Article 4.1(b) of the Agreement on Safeguards requires a determination of a threat of serious injury to "be based on facts and not merely on allegation, conjecture or remote possibility". In our view, the aforementioned deficiencies in the European Commission's assessment regarding how the domestic industry improved its competitive position in relation to imports from 2016 to 2017 indicate that the European Commission's subsequent conclusion as to the "vulnerab[ility]" of the domestic industry to further competition from imports was not "based on facts" as required by Article 4.1(b).

7.6.2.1.3 Whether the European Commission improperly inferred that improvements owing to the AD/CV measures would be temporary and reversible

7.223. We recall that Turkey presents a further contention on the assumption, *arguendo*, that the European Commission correctly identified the AD/CV measures as a main factor contributing to the domestic industry's improved performance. In particular, Turkey argues that the European Commission erred by inferring that the positive effects of those measures would be temporary and reversible.⁴⁰⁶ We have already found that the European Commission's analysis of the role played by the AD/CV measures in the domestic industry's improved performance was deficient. We nonetheless consider it useful for the effective resolution of the dispute to examine Turkey's contention regarding the effects of the AD/CV measures.

7.224. According to Turkey, if AD/CV measures adopted during a safeguard investigation led to improvements in the domestic industry's performance, it stands to reason that those improvements will persist into the future.⁴⁰⁷ For Turkey, by adopting AD/CV measures, an authority seeks to ameliorate certain sources of injury, an approach that is subsequently borne out if the domestic industry's performance improves.⁴⁰⁸ Moreover, given that those measures can last for five years and potentially be renewed, Turkey contends that the resulting amelioration of certain sources of injury should be considered durable.⁴⁰⁹ The European Union responds that the European Commission found that imports subject to AD/CV measures were being replaced by imports from other sources and, moreover, the improvements in domestic consumption and prices together with the Section 232 measures attracted yet further imports to the EU market.⁴¹⁰

7.225. We recall that the European Commission found that there was a "significant deterioration" in the situation of the domestic industry in 2013-2016, followed by improvements in 2017 as a result

may explain aspects of the domestic industry's overall improvement, but it does not explain why the domestic industry improved its competitive position *vis-à-vis* imports, since those factors benefitted both the domestic industry and imports in similar ways. (European Union's response to Panel question No. 26, para. 16). See also section 7.6.2.1.4 below.

⁴⁰⁵ Turkey's responses to Panel question No. 12, para. 44; and No. 26, para. 20.

 $^{^{\}rm 406}$ Turkey's response to Panel question No. 26, paras. 16 and 20; second written submission, paras. 155-158.

⁴⁰⁷ Turkey's response to Panel question No. 26, paras. 16 and 20. See also Turkey's second written submission para. 158; and response to Panel question No. 12, para. 45.

⁴⁰⁸ Turkey's response to Panel question No. 12, para. 45.

⁴⁰⁹ Turkey's response to Panel question No. 26, para. 16.

⁴¹⁰ European Union's comments on Turkey's response to Panel question No. 26, paras. 17-19.

WT/DS595/R

of, *inter alia*, the AD/CV measures.⁴¹¹ According to the European Union, the period 2013-2016 is illustrative of the domestic industry's vulnerability to imports.⁴¹² However, according to the European Commission's own explanation, the AD/CV measures imposed on some imports had contributed to the subsequent improvement in 2017. To that extent, these measures had mitigated certain sources of injury – or "vulnerability" – in relation to imports. It follows from the European Commission's reasoning that, absent some explanation to the contrary, a reversion to the kind of negative conditions experienced by the domestic industry in 2013-2016 could arise only through sources of injury that had not been mitigated by the AD/CV measures.

7.226. In this regard, the European Union contends that the European Commission found that imports subject to the AD/CV measures were being replaced by imports from other sources, and that import levels continued to increase despite the measures.⁴¹³ The European Union presented data in these proceedings indicating that the proportion of the total import volume that was subject to AD/CV measures was 33% in 2015, before declining to 21% and 10% in 2016 and 2017 respectively.414 Turkey responds that these data were not referenced in the European Commission's determination and should be disregarded.⁴¹⁵ Even if these data had been referenced explicitly by the European Commission, we do not consider that they support the European Union's case. Rather, when considered in light of other trends, they show that the replacement of imports subject to AD/CV measures with imports from other sources did not preclude the improvement in the domestic industry's performance in 2016-2017. Indeed, the domestic industry was able to achieve this improved performance *despite* import volumes (notably, those not subject to AD/CV measures) continuing to increase in 2016-2017.416 This raises the possibility that the domestic industry had surmounted its vulnerability to increased imports⁴¹⁷, for instance through the alleviation of the price depression from 2013-2016 by the imposition of AD/CV measures.

7.227. The European Union points to declines in prices in the third quarter of 2018, whilst imports continued to increase, as evidence that the domestic industry was vulnerable to imports despite the measures.418 imposition of AD/CV However, we agree with Turkev that the European Commission's findings of an acceleration in import volumes and a decline in prices after 2017 were not accompanied by evidence of downward trends in injury factors for the domestic industry.⁴¹⁹ Given that the domestic industry had improved its performance in 2016-2017 despite increased imports from sources not subject to AD/CV measures, we see no basis for inferring - in the absence of evidence to the contrary - that this dynamic would change after 2017.⁴²⁰ Indeed, the data available to the European Commission for the first semester of 2018 appeared to show that the "partial recovery" of the domestic industry was ongoing despite continued increases in import volumes.⁴²¹ The mere prospect of further increases in imports was thus insufficient to

⁴¹¹ Provisional determination, (Exhibit TUR-3), recital 55; Definitive determination, (Exhibit TUR-5), recitals 90 and 95.

⁴¹² European Union's first written submission, paras. 205 and 207.

⁴¹³ European Union's opening statement at the second meeting of the Panel, para. 35; comments on Turkey's response to Panel question No. 26, para. 18.

⁴¹⁴ European Union's response to Panel question No. 28(a), para. 25.

⁴¹⁵ Turkey's comments on the European Union's response to Panel question No. 28(a), para. 26.

⁴¹⁶ Turkey's response to Panel question No. 12, paras. 49-50.

⁴¹⁷ Turkey's opening statement at the second meeting of the Panel, para. 17; opening statement at the first meeting of the Panel, para. 34.

⁴¹⁸ European Union's response to Panel question No. 26, para. 17; opening statement at the second meeting of the panel, paras. 19 and 31; and opening statement at the first meeting of the Panel, para. 83.

⁴¹⁹ Turkey's second written submission, paras. 159-160. See also Definitive determination, (Exhibit TUR-5), recital 89.

⁴²⁰ We note, in this regard, the European Commission's finding that "imports of the product categories concerned rose by 4 % from 2017 to the MRP without causing serious injury". (Definitive determination, (Exhibit TUR-5), recital 143). The European Commission mentioned no evidence indicating that the replacement of imports subject to AD/CV duties with other imports was accompanied by a replacement of sources of injury. On the contrary, the fact that the domestic industry's performance improved in 2016-2017 as the replacement of imports increased suggested that the domestic industry had improved its competitiveness *vis-à-vis* imports, thus enabling it to face further increased imports in the future (Turkey's opening statement at the first meeting of the Panel, para. 34), and yet the European Commission did not contend with these indications from the evidence that contradicted its conclusions.

⁴²¹ Definitive determination, (Exhibit TUR-5), recital 89; European Union's opening statement at the first meeting of the Panel, para. 83.

demonstrate that the situation of the domestic industry was "vulnerable and fragile".⁴²² Additionally, we do not consider that declining prices in the third quarter of 2018 were indicative of the "new price depression" from imports that the European Commission had forecast in its provisional determination.⁴²³ These declines could arise from factors unrelated to imports. For instance, when assessing any harm caused by the domestic industry's allegedly poor export performance, the European Commission explained the drop in the domestic industry's export prices in 2017 to be a result of reductions in costs of raw materials.⁴²⁴ Thus, the European Commission explicitly recognized that price declines could be driven by factors other than price pressure from competitors. Against that background, we consider it significant that the determination does not set out the data underlying its observation of a price decline in the third quarter of 2018, nor does it analyse these data to ascertain whether this price decline was being caused by increased volumes of imports.

7.228. In summary, according to the European Commission's findings, the domestic industry was not vulnerable to imports subject to AD/CV measures to the extent that those measures produced positive effects for the domestic industry and continued to be in effect. The European Commission, however, did not demonstrate how the replacement of imports subject to AD/CV measures with imports from other sources could erode the positive effects of those measures. On the contrary, the domestic industry's performance improved in 2016-2017 whilst imports increased, particularly those not subject to these measures. In our view, therefore, the European Commission's respective findings that the "further increase of imports in 2018 – in particular from those countries or exporters not subject to trade defence measures – is likely to prevent the industry from a full recovery and from benefiting from [the AD/CV] measures"⁴²⁵, and that the domestic industry was "vulnerable to further increases of imports" not subject to those measures⁴²⁶, lacked the factual basis that is required for "threat of serious injury" determinations under Article 4.1(b) of the Agreement on Safeguards.

7.6.2.1.4 Whether the European Commission identified other factors contributing to the domestic industry's improved performance that demonstrate its vulnerability to increased imports

7.229. Finally, we recall that the European Union contends that there were factors other than the AD/CV measures contributing to the domestic industry's improved performance in 2016-2017. In particular, the European Union contends that the European Commission identified increased domestic consumption from 2013 to 2017 and increased sales by the domestic industry⁴²⁷, the cost of raw material remaining lower in 2017 than 2013, and the recovery in prices to 2013 levels, as factors in this regard.⁴²⁸ The European Union seems to argue that these factors could recede and leave the domestic industry exposed to the negative effects of imports that were increasing after 2017 due to, *inter alia*, trade diversion from the Section 232 measures.⁴²⁹ We recall, however, that the European Union explained that improvements in these other factors benefitted the domestic

⁴²² Turkey's second written submission, para. 159; comments on the European Union's response to Panel question No. 27, para. 25.

⁴²³ In its provisional determination, the European Commission considered that increased imports being diverted from the US market would "unavoidably result in a new price depression and undercutting on the EU market". (Provisional determination, (Exhibit TUR-3), recital 35). In the definitive determination, the European Commission considered that "monthly imports into the Union started to increase mostly since June 2018" as a result of such trade diversion, and "[m]oreover, steel prices in the Union started to follow a declining trend since the third quarter of 2018". (Definitive determination, (Exhibit TUR-5), recital 89). In the present proceedings, the European Union argues that the decline in prices was caused by downward price pressure from the increased import levels attributable to trade diversion from the US market. (European Union's response to Panel question No. 26, para. 17). In essence, we understand the European Union to argue that the price decline in the third quarter of 2018 represented the materialization of the anticipated "new price depression" due to increased imports being diverted from the US market that the European Commission had forecast in its provisional determination.

⁴²⁴ See, e.g. Definitive determination, (Exhibit TUR-5), recital 123.

⁴²⁵ Provisional determination, (Exhibit TUR-3), recital 61. See also ibid. recital 57.

⁴²⁶ Provisional determination, (Exhibit TUR-3), recitals 35 and 61.

⁴²⁷ The European Union contends that these factors were part of the "general recovery of the steel

market" that was identified European Commission at Provisional determination, (Exhibit TUR-3), recital 52.

⁴²⁸ European Union's second written submission, para. 55. See para. 7.212 for our understanding of the European Commission's determination on this point.

⁴²⁹ European Union's response to Panel question No. 12, para. 47; second written submission, paras. 55, 60, 64, and 75; and comments on Turkey's response to Panel question No. 28, para. 29.

industry and imports in similar ways.⁴³⁰ Yet, the European Union has provided no reason why declines in those factors would not likewise affect the domestic industry and imports in similar ways, leaving their competitive relationship unaltered. The European Union has thus not shown how the domestic industry would be any more exposed to imports as a result of declines in those factors than had been the case in 2016-2017, when the domestic industry improved its performance and competitive position *vis-à-vis* imports despite imports continuing to increase. Accordingly, the role of these other factors does not obviate the deficiencies in the European Commission's explanation, identified above in paragraphs 7.226-7.227 as to the allegedly partial, temporary, and readily reversible nature of the domestic industry's improvement. There is no reasoning in the European Commission's determination along the lines that downward trends in demand and increased raw material costs would leave the domestic industry more vulnerable to competition from increased imports.

7.6.2.1.5 Conclusion on the explanation for the determination of threat of serious injury

7.230. In conclusion, we have found that the European Commission's explanation of the AD/CV measures as a main contributor to the domestic industry's improved performance lacks basic information concerning the timing, scope, and effectiveness of those measures. Further, the European Commission's explanation in this regard did not account for improvements in product categories for which no AD/CV measures were imposed, thus leaving those improvements (and the assumption that they would be reversed in the face of increased import levels) unexplained and without a factual basis. As we consider the European Commission's explanation of the industry's improved performance to be deficient, this deficiency necessarily undermines the European Commission's inference that the improvement was partial, temporary, and readily reversible, and the European Commission's characterization of the domestic industry as "vulnerable" and "fragile" despite this improvement.

7.231. Additionally, to the extent that the AD/CV measures were responsible for the domestic industry's improved performance, the European Commission's explanation that the replacement imports from other sources would cause harm to the domestic industry was not "based on facts" as required by Article 4.1(b). Rather, the period during which these imports from other sources replaced those subject to AD/CV measures coincided with the domestic industry's recovery. Indeed, the domestic industry's performance improved during this period despite import volumes continuing to increase. The European Commission did not set out a factual basis for inferring that increased volumes of replacement imports from other sources would harm the domestic industry into the future.

7.232. For all of these reasons, we agree with Turkey that the European Commission's explanation that the domestic industry was "in a fragile and vulnerable position" despite its improved performance was flawed. Given the centrality of this flawed explanation to the European Commission's determination of a threat of serious injury⁴³¹, we find that the European Commission's determination of a threat of serious injury was not "based on facts" as required by Article 4.1(b) of the Agreement on Safeguards. In view of this finding, we consider it unnecessary for the effective resolution of the dispute to reach findings on substantively the same issue under Articles 4.1(a) and 4.2(a) of the Agreement on Safeguards.

7.6.2.2 Whether the European Commission did not establish a high likelihood of serious injury in the very near future

7.233. Turkey claims that the European Commission acted inconsistently with Article 4.1(b) of the Agreement on Safeguards by failing to demonstrate, on the basis of facts, that there was a high degree of likelihood of serious injury to the domestic industry in the very near future.⁴³² In essence, Turkey argues that the European Commission conflated its finding of a threat of serious injury with a finding of a threat of increased imports.⁴³³ For Turkey, this is because the European Commission's determination assumed that increased import levels in the future would bring about serious injury without demonstrating how those increased import levels would lead to a

⁴³⁰ European Union's response to Panel question No. 26, para. 16.

⁴³¹ Provisional determination, (Exhibit TUR-3), recitals 55, 60-61, and 69; Definitive determination, (Exhibit TUR-5), recitals 90 and 111.

⁴³² Turkey's first written submission, para. 245.

⁴³³ Turkey's second written submission, para. 182.

significant overall impairment in the position of the domestic industry.⁴³⁴ Turkey thus contends that the European Commission's determination of threat of serious injury was based on "conjecture" rather than on "facts".⁴³⁵

7.234. The European Union responds that the European Commission based its determination of a threat of serious injury on a projection as to the overall state of the domestic industry that was informed by facts relating to the negative trends in certain injury factors, the increase in imports during the MRP, namely mid-2017 to mid-2018, declining prices in the third quarter of 2018, and the likely further increase in imports due among others to the Section 232 measures.⁴³⁶ The European Union emphasizes that the expected further increase in imports should be understood in the context of the domestic industry's recovery from a period in which its position had deteriorated significantly due to increased imports.⁴³⁷ For the European Union, the expected further increase in imports would reverse the recovery and "necessarily" cause a deterioration in injury factors such as price and profitability.⁴³⁸

7.235. Brazil, as a third party, argues that the future likelihood of increased imports cannot be a trigger for safeguard measures. Rather, the "threat" must relate to "the actual trend of imports that have experienced a recent, sharp, sudden, and significative increase".⁴³⁹ Switzerland contends that "[b]ecause a threat determination requires an analysis of the likelihood that the serious injury will occur in the near-term, the expected volume of imports as well as their likely effect on the domestic industry could be relevant to the competent authority's analysis".⁴⁴⁰ For the United Kingdom, a threat of serious injury determination entails an assessment of both historical and existing facts from the reference period accompanied by fact-based projections concerning future developments in the domestic industry's condition.⁴⁴¹

7.236. The parties do not contest that the European Commission relied on the expectation of a further increase in import volumes as a "critical element"⁴⁴² in demonstrating that the domestic industry was currently threatened with serious injury. In our view, the main issue before us is whether the European Commission's determination of a threat of serious injury evinces a sufficient factual basis for projecting that this expected increase in imports would bring about a significant overall impairment in the position of the domestic industry.⁴⁴³ We begin by setting out the European Commission's findings on the likely impact of this expected increase on the domestic industry. At the provisional stage, the European Commission examined the factors that would lead to a further increase in imports and stated that:

In this context, a significant increase of supply on the Union market caused by an influx of imports will result in a general downward price pressure, resulting in price levels comparable to 2016 with significant negative consequences on the profitability of the Union steel industry.⁴⁴⁴

⁴³⁴ Turkey's second written submission, paras. 182 and 185-186; response to Panel question No. 29, para. 30; and opening statement at the second meeting of the Panel, para. 19.

⁴³⁵ Turkey's first written submission, para. 248; second written submission, para. 187.

⁴³⁶ European Union's comments on Turkey's response to Panel question No. 30, para. 38.

⁴³⁷ European Union's opening statement at the second meeting of the Panel, para. 38; closing statement at the second meeting of the Panel, para. 18.

⁴³⁸ European Union's response to Panel question No. 30, para. 40; opening statement at the second meeting of the Panel, para. 38.

³⁹ Brazil's third-party submission, para. 22.

⁴⁴⁰ Switzerland's third-party response to Panel question No. 6, para. 8.

⁴⁴¹ United Kingdom's third-party response to Panel question No. 6, para. 21.

⁴⁴² Definitive determination, (Exhibit TUR-5), recital 93.

⁴⁴³ We note Turkey's clarification that, in its view, "the competent authorities are permitted to, and actually should, consider the expected rate and amount of the increase in import levels as part of the threat and/or causation analysis" and further that: "[a]s the Appellate Body emphasised, facts, by their very nature, pertain to the present and the past, and thus to the [POI]. Those facts then serve as a basis for a projection that there is a high degree of likelihood of serious injury to the domestic industry in the very near future. In this regard, this prospective analysis must not only assess whether imports will likely increase in the future, but also the expected evolution of all relevant injury factors in light of past trends". (Turkey's response to Panel question No. 39, para. 64 (fn omitted)).

⁴⁴⁴ Provisional determination, (Exhibit TUR-3), recital 68. See also European Union's response to Panel question No. 30, para. 39(a).

7.237. At the definitive stage, the European Commission stated that:

It should however be noted that – as indicated in table 12 below – monthly imports into the Union started to increase mostly since June 2018. Moreover, steel prices in the Union started to follow a declining trend since the third quarter of 2018. It is, therefore, not possible to observe the effects of these imports and price development on the situation of the Union industry during the first semester of 2018. Therefore, the recent data confirmed the delicate situation of the Union industry and the threat posed by the most recent increase in imports.

....

In the provisional Regulation, the Commission concluded that the situation of the Union industry deteriorated significantly in the period 2013-2016 and recovered partially in 2017. However, the Commission considered that the Union industry, despite the temporary improvement, was still in a fragile situation and under the threat of serious injury if the increasing trend in imports continued with the ensuing price depression and profitability drop below sustainable levels.

...

[T]he Commission concluded that the Union industry was in a fragile situation, recovering from a period where its situation had deteriorated significantly. This recovery was attributed, inter alia, to the effectiveness of the different trade defence measures that have been adopted, in particular since 2016. ... In this context, the Commission confirmed that the ongoing provisional recovery could quickly be reversed if a further increase of imports was to take place.

...

[D]espite the fact that in 2017 the profitability levels had significantly improved from previous years (where the [domestic] industry was either loss-making or break-even), this situation could rapidly be reversed if imports would continue to increase (or surge, as a result of *inter alia*, the US Section 232 measures). ... In this context, the established risk of trade diversion would be a key element that would negatively affect the current economic situation of the Union industry if measures are not adopted.⁴⁴⁵

7.238. It is apparent from these passages that the European Commission projected that a further increase in import levels would result in price depression and declines in the domestic industry's profitability. The factual basis for this projection was the negative impact of increased import levels on the domestic industry in 2013-2016.⁴⁴⁶ We consider this to be inadequate. It presumes that the market conditions prevailing in 2013-2016 would be replicated in the future, without accounting for the record evidence that suggests that those conditions may have changed. In particular, the European Commission identified the adoption of AD/CV measures as a factor that had impacted the market after 2016.⁴⁴⁷ It also identified a trend in which imports subject to those AD/CV measures were being replaced by imports from other sources.⁴⁴⁸ Thus, the composition of imports at the end of the POI had changed vis-à-vis 2013-2016.449 Meanwhile, as imports subject to AD/CV measures declined, overall import levels continued to increase for the remainder of the POI at the same time as the domestic industry's performance improved.⁴⁵⁰ Increased imports based after did their composition 2016 not result in deterioration in the on а domestic industry's performance in 2016-MRP. Contrary to the assertion of the European Union, the

⁴⁴⁵ Definitive determination, (Exhibit TUR-5), recitals 89, 90, 95, and 97. See also

European Union's response to Panel question No. 30, paras. 39(b)-(e).

⁴⁴⁶ Provisional determination, (Exhibit TUR-3), recital 74.

⁴⁴⁷ Provisional determination, (Exhibit TUR-3), recitals 52-53 and 69.

⁴⁴⁸ Provisional determination, (Exhibit TUR-3), recitals 55-56.

⁴⁴⁹ The European Commission alluded to this when it found that the main threat of increased imports into the future would arise from those imports not subject to AD/CV measures. (Provisional determination, (Exhibit TUR-3), recitals 35, 55, and 61).

⁴⁵⁰ We note that the domestic industry's improved performance appears to have continued from 2017 into the first semester of 2018 whilst the increase in imports accelerated. (Definitive determination, (Exhibit TUP 5), racital 80; European Union's enoming statement at the first meeting of the Papel, pare 82)

⁽Exhibit TUR-5), recital 89; European Union's opening statement at the first meeting of the Panel, para. 83).

data set out in the European Commission's determination do not show that a further increase in imports would "necessarily" cause a deterioration in prices and profitability.⁴⁵¹

7.239. We thus agree with Turkey that "[e]ven if imports were set to increase in 2018, this does not automatically lead to the conclusion that there is a significant overall impairment in the position of the domestic industry which is clearly imminent".⁴⁵² The European Commission did not identify data suggesting a correlation between increases in imports based on their composition at the end of the POI, on the one hand, and a negative impact on the situation of the domestic industry, on the other hand. The European Union points out that the European Commission found that prices started following a declining trend in the third quarter of 2018.⁴⁵³ However, as we stated earlier, we do not consider that this necessarily implies a "new price depression" from imports at the end of the POI.⁴⁵⁴ As we explained earlier, the European Commission itself recognized that price declines for the products at issue could be due to factors unrelated to imports, such as reductions in the cost of raw materials⁴⁵⁵, and the European Commission engaged in no examination in this regard.

7.240. In short, the impact of increased imports on the domestic industry in 2013-2016 did not provide a sufficient factual basis for projecting the impact of increased imports on the domestic industry after the end of the POI, nor did the European Commission identify an alternative factual basis for this projection. Moreover, the fact that the domestic industry's performance improved between 2016 and the MRP whilst imports continued to increase underscores that increased import levels cannot be assumed to bring about a significant overall impairment in the situation of the domestic industry, and the European Commission did not engage with that evidence, which contradicted its conclusion. We therefore find that the European Commission's determination of a threat of serious injury was not "based on facts" as required by Article 4.1(b) of the Agreement on Safeguards.⁴⁵⁶

7.6.2.3 Turkey's consequential arguments relating to the European Commission's determination of threat of serious injury

7.241. Turkey appears to argue that the inconsistency of the European Commission's determination of a threat of serious injury with the requirements set out in Articles 4.1(a), 4.1(b), and 4.2(a) of the Agreement on Safeguards also results in purely consequential inconsistencies with Article 2.1 of the Agreement on Safeguards and Article XIX:1(a) of the GATT 1994.⁴⁵⁷ In view of our findings under Article 4.1(b) in sections 7.6.2.1 and 7.6.2.2 above, we consider it unnecessary for the effective resolution of the dispute to reach findings on substantively the same issues under Article 2.1 of the Agreement on Safeguards and Article XIX:1(a) of the GATT 1994.

7.6.2.4 Overall conclusion on threat of serious injury

7.242. In conclusion, we find that the definitive safeguard is inconsistent with Article 4.1(b) of the Agreement on Safeguards because two central elements of the European Commission's finding of a threat of serious injury were not "based on facts", as required by that provision. First, the European Commission's finding that the domestic industry was "in a fragile and vulnerable position" despite its improved performance was not "based on facts". Second. the European Commission's finding that a further increase in import volumes in the future would bring about serious injury to the domestic industry was not "based on facts". In view of these findings, we do not consider it necessary for the effective resolution of the dispute to address Turkey's claim

⁴⁵¹ European Union's response to Panel question No. 30, para. 40.

⁴⁵² Turkey's second written submission, para. 182.

⁴⁵³ European Union's opening statement of the first meeting of the Panel, para. 91; response to Panel question No. 30, para. 39(b).

⁴⁵⁴ See para 7.227 and fn 423 above.

⁴⁵⁵ See, e.g. Definitive determination, (Exhibit TUR-5), recital 123.

⁴⁵⁶ We note that the parties also disputed whether the European Commission adequately showed that the *extent* of the negative implications of the anticipated increase in imports sufficed to meet the threshold of serious injury. (Turkey's second written submission, para. 186; European Union's comments on Turkey's response to Panel question No. 29, para. 32). In view of our finding that the European Commission did not articulate a factual basis for the proposition that the anticipated increased imports would have any such negative implications in the first instance, it is unnecessary for the effective resolution of the dispute to address these other arguments of the parties.

⁴⁵⁷ Turkey's first written submission, para. 225. As set out elsewhere in this Report, Turkey also presents other arguments in support of its claims under Article 2.1 and Article XIX:1(a).

based on substantively identical⁴⁵⁸ arguments under Article 4.1(a)⁴⁵⁹, Turkey's substantively identical arguments under Article 4.2(a)⁴⁶⁰, or Turkey's consequential arguments concerning these same aspects of the definitive safeguard under Article 2.1 of the Agreement on Safeguards⁴⁶¹ and Article XIX:1(a) of the GATT 1994.462

7.7 Causation

7.243. Turkey argues that the definitive safeguard is inconsistent with the first and second sentences of Article 4.2(b) of the Agreement on Safeguards because the European Commission failed to establish a causal link between the increased imports and the threat of serious injury to the domestic industry, and failed to ensure that injury caused by factors other than imports was not attributed to increased imports.⁴⁶³ Turkey also argues that purely as a consequence of these inconsistencies, the definitive safeguard is also inconsistent with Article 2.1 of the Agreement on Safeguards and Article XIX:1(a) of the GATT 1994. The European Union responds that the European Commission established a causal link between increased imports and the threat of serious injury and conducted a non-attribution assessment in accordance with those provisions.⁴⁶⁴

7.244. We have found above that the European Commission's determination of a threat of serious was inconsistent with Article 4.1(b). Article 4.2(b) is concerned with the injury authority's demonstration of "the existence of the causal link between increased imports of the product concerned *and serious injury or threat thereof*⁴⁶⁵ It presupposes the existence of a threat of serious injury. Having found that the European Commission did not properly demonstrate the existence of a threat of serious injury, we do not consider it necessary for the effective resolution of this dispute to separately review whether the European Commission followed the requirements of Article 4.2(b) in determining that the increased imports caused that threat of serious injury.⁴⁶⁶ For the same reasons, we do not consider it necessary to address Turkey's purely consequential arguments on this point under Article 2.1 of the Agreement on Safeguards⁴⁶⁷ and Article XIX:1(a) of the GATT 1994.468

7.8 Applying the safeguard only to the extent and time necessary to prevent serious injury

7.245. Turkey claims that the definitive safeguard is inconsistent with Articles 5.1 and 7.1 of the Agreement on Safequards, and Article XIX:1(a) of the GATT 1994, because it is applied beyond the extent and time necessary to prevent serious injury.469

7.246. We will consider, first, Turkey's arguments under Article 5.1 of the Agreement on Safeguards (section 7.8.1) and, second, Turkey's arguments under Article 7.1 of the Agreement on Safeguards and Article XIX:1(a) of the GATT 1994 (section 7.8.2).

⁴⁶⁵ Emphasis added.

⁴⁵⁸ See fn 345 above.

⁴⁵⁹ We recall that a panel may decline to review one or more claims if three conditions are met, i.e. that the panel has (a) already found that the same measure (b) is inconsistent with one or more other provisions of the covered agreements, and, in addition, that (c) findings under the additional claims are not necessary to resolve the dispute. (See e.g. Panel Report, Pakistan - BOPP Film (UAE) [appealed by Pakistan 22 February 2021], para. 7.49; and Appellate Body Report, Canada - Wheat Exports and Grain Imports, para. 133).

⁴⁶⁰ Turkey also presents other arguments in support of its claim under Article 4.2(a), but we have rejected those arguments. (See paras. 7.12-7.13, 7.41, and 7.68-7.74 above).

⁴⁶¹ Turkey also presents other arguments in support of its claim under Article 2.1, but we have rejected those arguments. (See paras. 7.12-7.13, 7.41-7.67, 7.69, and 7.74 above).

⁴⁶² Turkey also presents other arguments in support of its claim under Article XIX:1(a), and we have already found that the definitive safeguard is inconsistent with Article XIX:1(a).

⁴⁶³ Turkey's first written submission, para. 270.

⁴⁶⁴ European Union's first written submission, para. 226.

⁴⁶⁶ Turkey also presents other arguments in support of its claim under Article 4.2(b). We have rejected those arguments. (See paras. 7.12-7.13, 7.41, 7.68-7.69, and 7.74 above).

⁴⁶⁷ Turkey also presents other arguments in support of its claim under Article 2.1, but we have rejected those arguments. (See paras. 7.12-7.13, 7.41-7.67, 7.69, and 7.74 above). 468 Turkey also presents other arguments in support of its claim under Article XIX:1(a), and we have

already found that the definitive safeguard is inconsistent with Article XIX:1(a). ⁴⁶⁹ Turkey's first written submission, para. 312.

7.8.1 Article **5.1** of the Agreement on Safeguards

7.247. Turkey argues that the definitive safeguard was applied beyond the extent necessary to prevent or remedy serious injury, inconsistently with Article 5.1 of the Agreement on Safeguards, for the following three reasons:

- although it concluded that there was threat of injury based, among other things, on an increase in imports observed in the first six months of 2018, the European Commission did not take import volumes for the first six months of 2018 into account when it set the size of the TRQ component of the safeguard⁴⁷⁰;
- b. the European Commission's causation analysis was inconsistent with Article 4.2(b) of the Agreement on Safeguards⁴⁷¹; and
- c. by suspending the application of AD/CV duties to the extent of their overlap with the out-of-quota safeguard duty, the definitive safeguard improperly addressed the serious injury caused by dumped and subsidized imports via the safeguard.⁴⁷²

7.248. Below, we first recall the applicable requirements of Article 5.1, and we then address each of these arguments, in turn.

7.8.1.1 The applicable requirements of Article 5.1 of the Agreement on Safeguards

7.249. The first sentence of Article 5.1 of the Agreement on Safeguards provides:

A Member shall apply safeguard measures only to the extent necessary to prevent or remedy serious injury and to facilitate adjustment.

7.250. The first sentence of Article 5.1 requires Members seeking to apply a safeguard to do so only to the "extent necessary" to (a) prevent or remedy serious injury and (b) facilitate adjustment. A safeguard that exceeds these limitations will be inconsistent with Article 5.1 of the Agreement on Safeguards.

7.251. The first part of Article 5.1 sets the maximum permissible extent to which a safeguard may be applied by reference to "serious injury". Other provisions of the Agreement on Safeguards, and Article XIX:1(a) of the GATT 1994, provide relevant context to understand the phrase "serious injury" in Article 5.1. The phrase "serious injury" appears in Article XIX:1(a) and is used twenty times in the Agreement on Safeguards. Article 4 of the Agreement on Safeguards, which is titled "Determination of Serious Injury or Threat Thereof", defines serious injury or its threat, and contains requirements for the investigation of serious injury or threat. The phrase "serious injury" in Article 5.1 has the same meaning as in the remainder of the Agreement on Safeguards. Therefore, the injury relevant for the purposes of Article 5.1 of the Agreement on Safeguards is the injury determined to exist pursuant to Article 4 of the Agreement on Safeguards⁴⁷³, which is caused or threatened by the increase in imports that results from unforeseen developments.

7.252. To evaluate Turkey's claims, we consider whether Turkey has established that the safeguard applied by the European Union exceeded the maximum permissible extent to which a safeguard may be applied pursuant to Article 5.1.

⁴⁷⁰ Turkey's first written submission, para. 330.

⁴⁷¹ Turkey's first written submission, para. 332.

⁴⁷² Turkey's first written submission, para. 334.

⁴⁷³ See also Appellate Body Report, US – Line Pipe, para. 249.

7.8.1.2 Whether Turkey has established that the definitive safeguard is inconsistent with Article 5.1 of the Agreement on Safeguards

7.8.1.2.1 Whether the definitive safeguard is inconsistent with Article 5.1 because the European Commission did not take into account data from the first six months of 2018 in determining the size of the TRQs

7.253. We recall that the definitive safeguard combines TRQs and an out-of-quota duty. The size of the TRQs was based on the average imports in the period from January 2015 to December 2017 (TRQ period).⁴⁷⁴ Turkey argues that because the European Union based its finding of threat of serious injury on, among other things, the increase in imports observed during the first six months of 2018, but did not include the data on imports during those six months when calculating the size of the TRQs, the European Union did not apply the safeguard only to the extent necessary to prevent serious injury, as required by Article 5.1.475

7.254. The European Union responds that the European Commission ensured that the safeguard was "commensurate with the goals of preventing or remedying serious injury and of facilitating adjustment".476

7.255. Among the third parties, Japan argues that the Panel must examine whether the safeguard measure at issue is in a form and at a level appropriate to or commensurate with the goal of preventing or remedying serious injury and facilitating adjustment.477

7.256. Turning to the published determinations, we note that there was some variation among the periods on which the European Commission relied to conclude that there was an increase in imports that threatened injury. The European Commission examined the increase in imports based on data for the period from January 2013 to June 2018.⁴⁷⁸ In the section of the definitive determination titled "Threat of Serious injury", the European Commission examined trends in several injury factors for the years from 2013 to 2017⁴⁷⁹; it made a discrete finding that the level of imports in the period from July 2017 to June 2018 was higher than that in the period from January 2017 to December 2017 due to "the relatively high level of imports in the first semester of 2018"⁴⁸⁰; and it examined monthly imports into the European Union and the United States for the period from January 2018 to September 2018 as compared to monthly imports in the period from January 2017 to September 2017 to analyse the "likelihood of further increased exports".⁴⁸¹ Turkey argues that since the European Commission's determination of the threat of serious injury was based on import data which included the first six months of 2018, it should also have included the first six months of 2018 in the period that it chose as the TRQ period.⁴⁸² Turkey does not explain further why this is the case.

7.257. We first note that nothing in Article 5.1 of Agreement on Safeguards requires the period during which the Member concerned finds an increase in imports that causes or threatens to cause

⁴⁷⁴ Definitive determination, (Exhibit TUR-5), recital 144.

⁴⁷⁵ Turkey's first written submission, para. 330.

⁴⁷⁶ European Union's first written submission, para. 270.

⁴⁷⁷ Japan's third-party statement, para. 8.

 ⁴⁷⁸ Definitive determination, (Exhibit TUR-5), recital 39.
 ⁴⁷⁹ Definitive determination, (Exhibit TUR-5), recitals 66-86. The European Commission noted that while it requested the EU industry associations to provide data for the first semester of 2018, that information could not be verified. The European Commission also noted that it "could not draw any reliable conclusion based on the situation of the [domestic] industry during the first semester of 2018" but that "based on these 2018 data" it "could ... confirm[]" the trend of a partial recovery of the industry observed in 2017. (Definitive determination, (Exhibit TUR-5), recital 89).

⁴⁸⁰ Definitive Determination, (Exhibit TUR-5), recital 101.

⁴⁸¹ Definitive determination, (Exhibit TUR-5), recitals 99 and 103.

⁴⁸² Turkey's first written submission, para. 330; second written submission, para. 229. Though its arguments on this subject are not entirely clear, we understand Turkey to link its contention that the TRQ period should have included the first six months of 2018 to the fact that the European Commission also examined data that included the first six months of 2018 when it conducted its threat of injury analysis. However, as noted above, while the European Commission's threat of injury analysis considered import data from 2018, this data was not limited to the first six months of 2018. Instead, the European Commission examined import data from the first nine months of 2018 (i.e. from January to September 2018). Given this, Turkey's argument that the TRQ period should have included the first six months of 2018 does not account for the fact that the European Commission did not examine this same period of time when it conducted its threat of injury analysis.

serious injury to be identical to the period that the Member uses to establish the level of a safeguard, or for the two periods to have the same end-points. In fact, Article 5.1 and its context establish that the Agreement on Safeguards does *not* require the two periods to be identical. We note, for example, that the second sentence of Article 5.1 generally requires that the level of a safeguard that takes the form of a quantitative restriction be set on the basis of data for "the last three representative years for which statistics are available". By comparison, Article 4 of the Agreement on Safeguards, which concerns the determination of serious injury or the threat thereof, does not prescribe a default length of the POI for the injury analysis. Thus, even though the second sentence of Article 5.1 is limited to quantitative restrictions, the comparison between Articles 4 and 5.1 illustrates that the Agreement on Safeguards does not require that the time periods that are used for setting the level of the safeguard be the same as the period that is used for the injury analysis.

7.258. Second, we note that Turkey appears to be confusing two different concepts, i.e. (a) what is *necessary to prevent or remedy* serious injury, and (b) the *period over which* an increase in imports giving rise to that serious injury is *observed*. Article 5.1 requires a safeguard to be applied only to the extent necessary to prevent or remedy serious injury. As noted above, setting a TRQ based on the level of imports during the same period as the one used by the competent authority in its examination of the increase in imports causing injury is not always necessary to fulfil this requirement, and Turkey has not explained why it was necessary in this particular case.

7.259. Referring to the panel report in *Chile – Price Band System*, Turkey argues that the TRQs that the European Commission adopted were not applied "only to the extent necessary to prevent serious injury" because they lack the required rational connection with the objective of preventing serious injury.⁴⁸³ The panel in *Chile – Price Band System* found that for a safeguard to be applied only to the extent necessary to prevent or remedy serious injury and facilitate adjustment there must be "a rational connection" between the measure and the objective of preventing or remedying serious injury and facilitating adjustment.⁴⁸⁴ We note that, in that dispute, the panel found that the level of the safeguard duty depended, among other factors, on a price threshold calculated on the basis of recent international prices; the panel observed that that the price threshold thus calculated was not "indicative of a state below which the domestic industry will experience (a threat of) serious injury", and it therefore bore "no rational connection to a state of the domestic industry below which (a threat of) serious injury will be experienced".485 That report, therefore, underlines that the permissible extent of the safeguard in a particular case must not be based on considerations other than those that are connected to a state of the domestic industry below which the industry will experience serious injury. However, Turkey has not explained why TRQs established based on the data concerning the volume of steel that was imported in a period that included the first six months of 2018⁴⁸⁶ would be connected to a state of the domestic industry below which the industry will experience serious injury, whereas the TRQs that the European Commission set based on the volume of imports from 2015 to 2017 are not. We therefore do not consider that Turkey's reference to the panel report in Chile - Price Band System supports Turkey's argument.

7.260. Based on the above, we consider that Turkey has not established that the European Union acted inconsistently with the first sentence of Article 5.1 of the Agreement on Safeguards by not taking into account data from the first six months of 2018 in determining the size of the TRQs.

7.8.1.2.2 Whether the definitive safeguard is inconsistent with Article **5.1** because the European Commission did not conduct a proper causation analysis

7.261. Turkey also argues that the definitive safeguard is inconsistent with Article 5.1 of the Agreement on Safeguards because the European Commission did not establish a causal link between the increased imports and the threat of injury and did not conduct the non-attribution analysis as

⁴⁸³ Turkey's first written submission, para. 330 (referring to Panel Report, *Chile – Price Band System*, para. 7.183).

⁴⁸⁴ Panel Report, Chile – Price Band System, para. 7.183. (emphasis original)

⁴⁸⁵ Panel Report, Chile – Price Band System, para. 7.184.

⁴⁸⁶ Turkey does not explicitly state what the starting point of the TRQ period should have been in its view. Given that the gist of Turkey's argument appears to be that a period used in the threat of injury analysis should also be included in the TRQ period, one might infer that Turkey would take the view that the TRQ period should begin January 2013 (i.e. the same point as the beginning of the injury analysis). However, as noted, Turkey's submissions were not explicit on this point. It is only in its comments on the interim report that Turkey made explicit that it was arguing that the starting point of the TRQ period should be 2015. (Turkey's request for interim review, para. 39).

required under Article 4.2(b) of the Agreement on Safeguards.⁴⁸⁷ The European Union argues that because the European Commission properly established the existence of a causal link between the increased imports and the threat of serious injury, Turkey's argument should be dismissed.⁴⁸⁸

7.262. We note that Turkey's argument is purely consequential to its claim under Article 4.2(b) of the Agreement on Safeguards. Having already found that the definitive safeguard is inconsistent with Article 4.1(b) of the Agreement on Safeguards, we have considered it unnecessary to decide whether the definitive safeguard is also inconsistent with Article 4.2(b).⁴⁸⁹ Likewise, we do not consider it necessary to consider Turkey's purely consequential argument that the definitive safeguard is inconsistent with Article 5.1 as a result of the claimed inconsistency with Article 4.2(b).⁴⁹⁰

7.8.1.2.3 Whether the double remedy regulation rendered the definitive safeguard inconsistent with Article **5.1**

7.263. We recall that as defined by Turkey, the legal instruments setting out the definitive safeguard include the double remedy regulation.⁴⁹¹ This regulation suspends the application of AD/CV duties on a particular product to the extent they overlap with the out-of-quota safeguard duty applicable to that product.⁴⁹² According to Turkey, this means that the definitive safeguard addresses the injury caused by dumped and subsidized imports. Turkey argues that the definitive safeguard is therefore inconsistent with Article 5.1 of the Agreement on Safeguards, because this provision permits the application of a safeguard only to the extent necessary to address the serious injury caused by the increased imports within the meaning of Article 2.1 and Article XIX:1(a).⁴⁹³

7.264. The European Union responds that, as set out in the double remedy regulation, the European Commission suspended the AD/CV duties to the extent of their overlap with the out-of-quota safeguard duty in order to avoid "a greater effect than desirable", and thus the suspension cannot be characterized as a failure to apply the measure only to the extent necessary to prevent or remedy serious injury and to facilitate adjustment.⁴⁹⁴

7.265. Among the third parties, Japan argues that the Panel must examine whether the safeguard measure at issue is in a form and at a level appropriate to or commensurate with the goal of preventing or remedying serious injury and facilitating adjustment.⁴⁹⁵ The Republic of Korea comments that safeguards and AD/CV measures are not mutually exclusive and may be applied in combination.⁴⁹⁶

7.266. At the outset of our evaluation, we note that Turkey's claim does not concern the question whether a safeguard can be applied to dumped and subsidized imports. Turkey clarified in response to a question from the Panel that it is not arguing that a safeguard cannot be applied to a product that is also subject to AD/CV duties.⁴⁹⁷ Turkey argues that by suspending AD/CV duties to the extent

⁴⁹³ Turkey's first written submission, para. 334; second written submission, para. 233.

⁴⁹⁴ European Union's first written submission, para. 282. See also European Union's closing statement at the second meeting of the Panel, paras. 34-36.

⁴⁹⁶ Korea's third-party submission, para. 25.

⁴⁹⁷ Turkey's response to Panel question No. 42, para. 78. See also Turkey's comments on the European Union's response to Panel question No. 38, para. 49.

⁴⁸⁷ Turkey's first written submission, paras. 331-332.

⁴⁸⁸ European Union's first written submission, para. 277.

⁴⁸⁹ See para. 7.242 above.

 ⁴⁹⁰ We consider, and reject, Turkey's other arguments in support of its Article 5.1 claim at paras. 7.253-7.260 above and paras. 7.263-7.271 below.
 ⁴⁹¹ See section 2.1 above. See also Turkey's panel request, paras. 13 and 17; and response to Panel

 ⁴⁹¹ See section 2.1 above. See also Turkey's panel request, paras. 13 and 17; and response to Panel question No. 14, para. 52.
 ⁴⁹² Double remedy regulation, (Exhibit TUR-7), Article 1 (if the safeguard duty is lower, the difference

⁴⁹² Double remedy regulation, (Exhibit TUR-7), Article 1 (if the safeguard duty is lower, the difference between the anti-dumping or countervailing duty and the safeguard duty is also applied). In the definitive determination, the European Commission noted that in order to prevent the cumulation of AD/CV duties with safeguards when the safeguard TRQ is exceeded, the European Commission could, pursuant to a framework to be developed at a later stage, suspend or reduce the level of existing AD/CV duties that were being assessed on certain covered imports. (Definitive determination, (Exhibit TUR-5), recital 186). Subsequently, the European Commission adopted the double remedy regulation stipulating that in cases where the safeguard duty exceeds the equivalent rate of the higher of the AD/CV duty rate applicable to the same product categories, only the safeguard duty shall be collected.

⁴⁹⁵ Japan's third-party statement, para. 8.

they overlapped with the safeguard duty, the European Union addressed the injury caused by the dumped and subsidized imports via the safeguard measure.⁴⁹⁸ The question that we must address to resolve Turkey's claim, therefore, is whether Turkey has shown that *by suspending AD/CV duties* to the extent that they overlap with the safeguard duty, the European Union failed to apply the safeguard only to the extent necessary to prevent serious injury due to increased imports within the meaning of Article 2.1 and Article XIX:1(a), thereby acting inconsistently with Article 5.1 of the Agreement on Safeguards.

7.267. Turkey appears to suggest that merely because AD/CV duties were suspended to the extent of their overlap with the safeguard duty, the safeguard duty should be considered to perform the function of the suspended AD/CV duties, i.e. addressing the injury caused by dumping and subsidization, such that the safeguard duty becomes inconsistent with Article 5.1 of the Agreement on Safeguards. We do not agree. We note that the European Commission explained its determination to suspend any AD/CV duties to the extent that they overlapped with the safeguard duty by stating that:

[T]he combination of anti-dumping and/or anti-subsidy measures and safeguard measures ... would ... have an effect greater than that intended or desirable in terms of the Union's trade defence policy and objectives[.]⁴⁹⁹

7.268. As such, the European Commission did not indicate that it was suspending the AD/CV duties because they addressed the *same* injury as the safeguard. Instead, the European Commission indicated that the combined effect of all measures would have been greater than desirable. Turkey has not countered this explanation. Specifically, Turkey has not demonstrated that rather than choosing *not to address* the additional injury caused by dumping and subsidization (i.e. injury other than the serious injury caused by increased imports within the meaning of Article 2.1 and Article XIX:1(a)), the European Union was addressing that injury via the definitive safeguard by suspending the AD/CV duties to the extent of their overlap with the safeguard duty.⁵⁰⁰ We therefore consider that the suspension of the AD/CV duties to the extent of their overlap with the safeguard duty does not, on its own, establish that the safeguard duty was being applied beyond the extent necessary to prevent serious injury caused by increased imports within the meaning of Article 2.1 and Article XIX:1(a).

7.269. We also note the following statement made by Turkey:

Where there are already anti-dumping and countervailing measures in place, prior to the increase in imports resulting from unforeseen developments, a subsequent finding that the imposition of safeguard measures is warranted may result in the imposition of a safeguard measure to offset the serious injury caused by the increased imports in addition to the existing anti-dumping or countervailing duty.⁵⁰¹

7.270. Turkey thus acknowledges that a Member may apply a safeguard to offset the serious injury caused by increased imports even if those imports are also subject to AD/CV measures. Notwithstanding this, Turkey asserts that the definitive safeguard, and in particular the double remedy regulation, was nonetheless inconsistent with Article 5.1 of the Agreement on Safeguards because it suspended the AD/CV duties that overlapped with the safeguard duty.⁵⁰² In making these statements, Turkey appears to argue that an inconsistency with Article 5.1 may not have presented itself had the European Commission continued to apply both the full amount of the pre-existing AD/CV duties and the safeguard duty on the relevant imports, but that the European Commission acted inconsistently with Article 5.1 because it suspended part of the AD/CV duties. To us, this position is internally contradictory.

⁴⁹⁸ Turkey's second written submission, para. 233; response to Panel question No. 42, para. 80.

⁴⁹⁹ Double remedy regulation, (Exhibit TUR-7), recital 15.

⁵⁰⁰ In this regard, we note that the application of AD/CV measures to counteract dumping and/or subsidization in accordance with the Agreement on Implementation of Article VI of the General Agreement on Tariffs and Trade 1994 (Anti-Dumping Agreement) and the Agreement on Subsidies and Countervailing Measures (SCM Agreement) is a right that Members may, but are not mandated to, exercise. Members are also not mandated to apply the full amount of AD/CV duties that they are permitted to apply by the Anti-Dumping Agreement and the SCM Agreement.

⁵⁰¹ Turkey's response to Panel question No. 42, para. 80.

⁵⁰² Turkey's response to Panel question No. 42, para. 80.

7.271. Based on the above, we consider that Turkey has not established that the double remedy regulation resulted in the application of the definitive safeguard beyond the extent necessary to prevent serious injury.

7.8.2 Article 7.1 of the Agreement on Safeguards and Article XIX:1(a) of the GATT 1994

7.272. Turkey argues that because the European Commission did not perform a proper causation analysis in accordance with Article 4.2(b), the European Union was unable to ensure that the safeguard was applied only for the period of time necessary to address the serious injury attributed to increased imports. On this basis, Turkey claims that the definitive safeguard was inconsistent with Article 7.1 of the Agreement on Safeguards⁵⁰³ and Article XIX:1(a) of the GATT 1994.⁵⁰⁴ The European Union argues that Turkey's contention ought to be rejected.⁵⁰⁵

7.273. We note that Turkey's claim under Article 7.1 of the Agreement on Safeguards, and Turkey's relevant argument under Article XIX:1(a) of the GATT 1994, are purely consequential to Turkey's claim under Article 4.2(b) of the Agreement on Safeguards, and that we have not reached findings of inconsistency under Article 4.2(b). Given that we have already found the definitive safeguard to be inconsistent with other provisions of the Agreement on Safeguards, and that the additional findings requested by Turkey are not necessary to resolve the dispute, we exercise economy on Turkey's consequential claim under Article 7.1 of the Agreement on Safeguards, and we do not review Turkey's consequential argument⁵⁰⁶ under Article XIX:1 of the GATT 1994.

7.9 The allocation of shares in the TRQs

7.274. Turkey claims that the definitive safeguard is inconsistent with Article XIII:2(d) and the *chapeau* of Article XIII:2 of the GATT 1994 because it allocates country-specific shares in the TRQs based on a period that is not representative, and without taking due account of any special factors affecting the trade in the product concerned.⁵⁰⁷ Turkey claims that the definitive safeguard is also inconsistent with Article 5.2(a) of the Agreement on Safeguards, "for the same reasons".⁵⁰⁸ The European Union contends that the European Commission fully conformed with the requirements of Article XIII of the GATT 1994.⁵⁰⁹ The European Union also contends that Article 5.2(a) of the Agreement on Safeguards that Article 5.2(a) of the Agreement on the European Union also contends that Article 5.2(a) of the Agreement on Safeguards that Article 5.2(a) of the Agreement on Safeguards that Article 5.2(a) of the Agreement on the European Union also contends that Article 5.2(a) of the Agreement on the European Union these bases, the European Union requests the Panel to reject Turkey's claims.

7.275. In the following sections, we first address Turkey's claims under the relevant provisions of Article XIII:2 of the GATT 1994 (section 7.9.1), and then Turkey's claim under Article 5.2(a) of the Agreement on Safeguards (section 7.9.2).

7.9.1 Article XIII:2 of the GATT 1994

7.276. In the definitive determination, the European Commission determined that "a country specific [TRQ] should be allocated to countries having a significant supplying interest, based on their imports over the last 3 years".⁵¹¹ While the import volumes data from the first six months of 2018 were on the record before the European Commission, the European Commission did not take those data into account when it allocated the country-specific shares in the TRQs for the relevant product categories.

⁵⁰³ Article 7.1 of the Agreement on Safeguards provides:

A Member shall apply safeguard measures only for such period of time as may be necessary to prevent or remedy serious injury and to facilitate adjustment. The period shall not exceed four years, unless it is extended under paragraph 2.

⁵⁰⁴ Turkey's first written submission, para. 333; response to Panel question No. 13, para. 51.

⁵⁰⁵ European Union's first written submission, para. 279.

⁵⁰⁶ Turkey makes other arguments in support of its claim under Article XIX:1(a), and we have already found that the definitive safeguard is inconsistent with Article XIX:1(a).

⁵⁰⁷ Turkey's first written submission, para. 336.

⁵⁰⁸ Turkey's first written submission, para. 336.

⁵⁰⁹ European Union's first written submission, para. 296.

⁵¹⁰ European Union's first written submission, para. 285.

⁵¹¹ Definitive determination, (Exhibit TUR-5), recital 147.

7.277. First, Turkey argues that the European Commission's decision to allocate the country-specific shares in the TRQs based solely on import volume data from 2015 to 2017 and without considering data from the first six months of 2018 was inconsistent with Article XIII:2(d) of the GATT 1994 on the following alternative grounds: (a) the "previous representative period" used by the European Commission should have included the first half of 2018, because the first half of 2018 formed part of the MRP preceding the application of the safeguard, and because in the first half of 2018 the European Union had introduced new trade remedy measures (the representative period argument); or (b) there had been changes in imports shares from various countries following the imposition of new trade remedies measures in the first half of 2018, and these changes were a "special factor" of which the European Union did not take due account (the special factor argument).⁵¹² Turkey asserts that the Panel must address the special factor argument "if [it] were to be concluded that there was no obligation for the European Union to have included the first six months of 2018 into the representative period".⁵¹³

7.278. Second, Turkey argues that for the same reasons, this allocation of country-specific shares was inconsistent with the *chapeau* of Article XIII:2.⁵¹⁴

7.279. Below, we first set out the applicable requirements of Article XIII:2 of the GATT 1994. We then address Turkey's claim under Article XIII:2(d) of the GATT 1994, before turning to Turkey's claim under the *chapeau* of Article XIII:2 of the GATT 1994.

7.9.1.1 The applicable requirements of Article XIII:2 of the GATT 1994

7.280. Article XIII:2 *chapeau* and Article XIII:2(d) of the GATT 1994, in relevant part, provide:

In applying import restrictions to any product, contracting parties shall aim at a distribution of trade in such product approaching as closely as possible the shares which the various contracting parties might be expected to obtain in the absence of such restrictions and to this end shall observe the following provisions:

...

(d) In cases in which a quota is allocated among supplying countries the contracting party applying the restrictions may seek agreement with respect to the allocation of shares in the quota with all other contracting parties having a substantial interest in supplying the product concerned. In cases in which this method is not reasonably practicable, the contracting party concerned shall allot to contracting parties having a substantial interest in supplying the product shares based upon the proportions, supplied by such contracting parties during a previous representative period, of the total quantity or value of imports of the product, due account being taken of any special factors which may have affected or may be affecting the trade in the product.

7.281. The *chapeau* to Article XIII:2 provides that "in applying import restrictions" to a product, the contracting parties "shall aim at a distribution of trade ... approaching as closely as possible the shares which the various contracting parties might be expected to obtain in the absence of such restrictions". The ordinary meaning of "aim" includes "[t]o have (something) as an object, intention, or desired outcome ... to seek to achieve or obtain", and to "direct one's course towards a particular point or destination, or to make this one's object".⁵¹⁵ This provision therefore requires a Member, when applying import restrictions to a product, to seek to achieve or obtain a distribution of trade that approaches, "as closely as possible", the shares that other Members "might be expected to obtain" in the overall imports of the product concerned "in the absence of *such restrictions*".⁵¹⁶ The phrase *such restrictions* refers to the import restrictions that are applied by the Member concerned under Article XIII:2. The *chapeau* further specifies that the subparagraphs that follow *shall* be observed to achieve the "end" specified in the *chapeau*. The *Ad* Note to Article XIII:2(d) refers to

⁵¹² Turkey's first written submission, paras. 347-348; response to Panel question No. 43, para. 89.

⁵¹³ Turkey's response to Panel question No. 43, para. 89.

⁵¹⁴ Turkey's first written submission, para. 352; response to Panel question No. 44, para. 90.

⁵¹⁵ Oxford Dictionaries online, definition of "aim", v., meanings 6 and 7,

https://www.oed.com/view/Entry/4348?rskey=daocNi&result=4&isAdvanced=false#eid (accessed 22 September 2021).

⁵¹⁶ Emphasis added.

the *chapeau* as containing a "general rule". The provisions of Articles XIII:2(a) to XIII:2(d) specify ways in which the general rule in the *chapeau* of Article XIII:2 may be fulfilled in certain instances.⁵¹⁷

7.282. Subparagraph (d), which concerns "cases in which a quota is allocated among supplying countries" as the means to achieve the "end" stipulated in the *chapeau*, prescribes two methods through which the Member concerned may determine the "allocation of shares in the quota" among Members that have "a substantial interest in supplying the product concerned". The present dispute involves the second of these methods, which is set out in the second sentence of Article XIII:2(d). This method requires the Member applying the import restriction at issue to allocate shares in the quota for imports of the product concerned among the relevant contracting parties "based upon the proportions, supplied by such contracting parties during a previous representative period". Aside from specifying that the previous period to be used for this purpose must be *representative*, the provision does not prescribe how long the period must be or what must be the start and end points of this period. Notably, the provision refers to "a" previous representative period, not "the" previous representative period. Further, the second sentence of Article XIII:2(d) also requires the Member applying the import restriction at issue to take "due account" of "any special factors which may have affected or may be affecting the trade in the product" when allotting shares in the quota to the relevant Members.

7.9.1.2 Whether Turkey has established that the definitive safeguard is inconsistent with Article XIII:2(d) of the GATT 1994

7.283. Turkey contends that the European Commission acted inconsistently with Article XIII:2(d) of the GATT 1994 on the following two alternative⁵¹⁸ grounds: (a) the European Commission failed to properly choose the previous representative period for the purposes of Article XIII:2(d) by not including the first six months of 2018 in the representative period (the representative period argument)⁵¹⁹; and (b) the European Commission failed to take into account the change in the share of imports from the exporting countries following "the newly imposed trade defence measures" during the first half of 2018 as a "special factor" that justified the adjustment of the TRQ allocation (the special factor argument).⁵²⁰ We will first consider the representative period argument, and then the special factor argument.

7.9.1.2.1 The representative period argument

7.284. Turkey argues that the European Commission should have included the first six months of 2018 in the reference period used to allocate country-specific shares in the TRQs for relevant product categories, because the first half of 2018 formed part of the MRP preceding the application of the safeguard, and because in the first half of 2018 the European Union had introduced new trade remedy measures.⁵²¹ According to Turkey, by not including data from the first six months of 2018, and instead allocating the country-specific shares in the TRQs based on data from January 2015 to December 2017, the European Commission did not allocate country-specific shares in the TRQs for various product categories based on the MRP not distorted by the TRQs, and therefore acted inconsistently with Article XIII:2(d).⁵²²

7.285. The European Union argues that there is no general rule in Article XIII:2(d) of the GATT 1994 always requiring the use of the most recent three-year period preceding the entry into force of a TRQ.⁵²³ The European Union also asserts that the reference period used by the European Commission, i.e. 2015-2017, was also a recent one, and meets the criterion of a "previous representative period".⁵²⁴ According to the European Union, the consideration of average imports during the last three calendar years prior to the initiation of the investigation (in this instance, 2015-2017) fulfils the requirements of Article XIII of the GATT 1994.⁵²⁵ The European Union also

⁵¹⁷ Appellate Body Reports, *EC – Bananas III (Article 21.5 – Ecuador II) / EC – Bananas III (Article 21.5 – US)*, para. 338.

⁵¹⁸ Turkey's response to Panel question No. 43, para. 89.

⁵¹⁹ Turkey's first written submission, para. 351.

⁵²⁰ Turkey's response to Panel question No. 43, para. 89.

⁵²¹ Turkey's first written submission, para. 348.

⁵²² Turkey's first written submission, paras. 348 and 351.

⁵²³ European Union's first written submission, para. 288; comments on Turkey's response to Panel question No. 43, para. 70 (both referring to Panel Report, *EU – Poultry Meat (China)*, para. 7.353).

⁵²⁴ European Union's first written submission, para. 290.

⁵²⁵ European Union's first written submission, para. 291.

contends that unlike the obligation in respect of the examination of the increase in imports, the obligation in respect of the previous representative period did not require a significant focus on the first six months of 2018.⁵²⁶ The European Union adds that, in fact, it is important to use full years to ensure representativeness, because of seasonal variations.⁵²⁷

7.286. We note that in the definitive determination the European Commission determined that a country-specific TRQ should be allocated to countries having a share of more than 5% of imports for the product category concerned in the period 2015-2017, and that a global TRQ based on the average of the remaining imports in the same period should be allocated to all other supplying countries.⁵²⁸ The Commission based the size of the TRQs, country-specific as well as global, on the average volume of imports in the period 2015-2017, plus 5%.⁵²⁹

7.287. Referring to the panel reports in *EC* – *Bananas III (Article 21.5* – *Ecuador)* and *US* – *Line Pipe*, Turkey argues that it is necessary for the "previous representative period" for purposes of Article XIII:2(d) to be the MRP not distorted by restrictions, i.e. not distorted by the TRQ in question.⁵³⁰ Turkey argues that to achieve this, the European Commission should have allocated country-specific shares in the TRQs based on data including the first half of 2018.⁵³¹

7.288. We note that the second sentence of Article XIII:2(d) requires the allocation of country-specific shares in a TRQ based on proportions supplied by the relevant Members during "*a* previous representative period".⁵³² The use of the indefinite article "a" instead of the definite article "the" in reference to the notion of "previous representative period" shows that the Member applying the TRQ has a margin of discretion in selecting a previous period that it considers to be representative for the purpose of allocating country-specific shares in the TRQ at issue.⁵³³ Indeed, the panel in *EC – Bananas III (Article 21.5 – Ecuador)*, to which Turkey refers in support of its contention, noted that "Members have a degree of discretion in choosing a previous representative period".⁵³⁴ We also do not consider that the observation of the panel in *US – Line Pipe* referred to by Turkey that "trade flows before the imposition of a safeguard measure provide an objective, factual basis for projecting what might have occurred in the absence of that measure" supports the proposition that only the MRP before the *application* of the TRQ, including where applicable the most recent half-year, could constitute a valid previous representative period for the purpose of Article XIII:2(d).

7.289. The period 2015-2017 used by the European Commission to allocate country-specific shares in the TRQs comprised the most recent three calendar years before the *initiation* of the investigation.⁵³⁵ We have some difficulty with the notion that the European Union could be held to have acted inconsistently with Article XIII:2(d) of the GATT 1994 on the ground that the period it used to allocate shares in the TRQs was not the most recent, despite the inclusion of the three most recent full years before the initiation of the investigation. In our view, the European Commission's decision to consider the most recent three calendar years before the initiation of the investigation of the IRQs was within the margin of discretion available to the European Commission in selecting a previous representative period pursuant to Article XIII:2(d) of the GATT 1994. Therefore, we reject Turkey's contention that

⁵²⁶ European Union's first written submission, paras. 292-294.

⁵²⁷ European Union's first written submission, para. 295.

⁵²⁸ Definitive determination, (Exhibit TUR-5), recitals 146-147.

⁵²⁹ Definitive determination, (Exhibit TUR-5), recitals 144 and 146-147.

⁵³⁰ Turkey's first written submission, para. 351 (referring to Panel Reports, EC – Bananas III

⁽Article 21.5 – Ecuador), para. 6.39; and US – Line Pipe, para. 7.54).

⁵³¹ Turkey's first written submission, para. 351; second written submission, para. 237.

⁵³² Emphasis added.

 $^{^{533}}$ We note that the panel in *EU* – *Poultry Meat (China)* also took the view that "the reference to 'a' previous representative period in Article XIII:2(d) implies that there is no general rule that applies in all cases regarding the selection of the reference period". (Panel Report, *EU* – *Poultry Meat (China)*, para. 7.349).

⁵³⁴ Panel Report, *EC* – *Bananas III (Article 21.5* – *Ecuador)*, para. 6.50. We also note that the observation of the panel to which Turkey refers in support of its contention was made by the panel when discussing its view that "if data from a period are out of date or imports distorted because the relevant market is restricted, then using that period as a representative period cannot achieve the aim of the chapeau". (Ibid. para. 6.39). In this case, Turkey has not argued that data from the period 2015-2017, which were used by the European Commission to allocate country-specific shares in the relevant TRQs, were "out of date". Turkey has also not contended that imports in this period were distorted by restrictions.

⁵³⁵ The notice of initiation of the investigation is dated 26 March 2018. (Notice of initiation, (Exhibit TUR-1), p. 29).

the European Commission was obliged to include data from the first six months of 2018 in the period used as the basis to allocate country-specific TRQs on the basis that these were the most recent data undistorted by the TRQs.

7.290. Turkey also argues that the use of data relating to the first six months of 2018 was necessary because the European Commission had imposed a number of trade remedies in that period that resulted in a significant change in the shares of imports from a number of countries.⁵³⁶ Turkey contends that imports from countries not subject to AD/CV measures increased to the detriment of imports from countries subject to those measures.⁵³⁷ Turkey asserts that had the first six months of 2018 been included in the period based on which country-specific shares were allocated, Turkey would have received larger country-specific quotas and would have been allocated a country-specific TRQ for at least two additional product categories.⁵³⁸

7.291. We consider that the imposition of trade remedies by the European Commission in the first six months of 2018 does not establish that the period 2015 to 2017 could not have been a previous representative period for the purpose of the second sentence of Article XIII:2(d) of the GATT 1994. We do not see anything in the text of Article XIII:2(d) that makes a period in which new trade remedies have been imposed a mandatory part of the "previous representative period" for the allocation of country-specific shares in the TRQ at issue in a particular investigation. Turkey has not demonstrated that the imposition of new trade remedies in 2018 made it necessary to include the first six months of 2018 in this proceeding, and indeed it has not even identified the new trade remedies at issue, or explained how they differed from trade remedies already applied during 2015-2017. Therefore, in our view, the imposition of trade remedies by the European Commission in the first six months of 2018 and any associated change in the proportions of imports supplied by various countries does not, by itself, call into question the representativeness of the period 2015 to 2017 for the purpose of Article XIII:2(d) in this case. In this regard, we agree with the observation of the panel in EU - Poultry Meat (China) that "there is nothing unusual about Members applying WTO-consistent measures which may, directly or indirectly, affect the importation of certain products".⁵³⁹ Thus, we do not consider that the application of trade remedies in 2018 undermined the representativeness of the period 2015-2017 for the purposes of the second sentence of Article XIII:2(d) of the GATT 1994.

7.292. Finally, we note Turkey's assertion that the European Commission's consideration of data from the first six months of 2018 in its examination of increase in imports "confirms that the data relating to that period was reliable and relevant for the purpose of imposing the safeguard measures at issue".⁵⁴⁰ To the extent that Turkey is arguing that the inclusion of the first six months of 2018 in the examination of increase in imports obliged the European Commission to include the first six months of 2018 in the period based on which it allocated country-specific shares in the TRQs, we disagree. Article XIII:2(d) does not necessarily require the period used as the "previous representative period" to allocate country-specific shares in a TRQ and the period used by the competent authority to examine increased imports in a safeguard investigation to be identical. Indeed, in this case, even under Turkey's arguments the two periods would not correspond, given that the European Commission assessed the increase in imports based on data for 2013-June 2018, whereas we understand Turkey to be calling for an allocation of TRQ shares based on data for 2015-June 2018. We therefore reject Turkey's additional argument.

7.293. Based on the foregoing, we reject Turkey's contention that the European Commission was required to include the first six months of 2018 in the period based on which it allocated country-specific shares in the relevant TRQs to fulfil the requirements of Article XIII:2(d) of the GATT 1994. Having rejected Turkey's contentions concerning the representative period, we now

⁵³⁷ Turkey's first written submission, para. 348; response to Panel question No. 43, para. 88.

⁵³⁶ Turkey's first written submission, para. 348; second written submission, para. 239.

⁵³⁸ Turkey's first written submission, paras. 349-350.

⁵³⁹ Panel Report, *EU – Poultry Meat (China)*, para. 7.337. Although this observation was made by the panel when addressing the question whether certain sanitary and phytosanitary (SPS) measures that reduced China's ability to export the relevant products were a "special factor" under Article XIII:2(d), we consider that based on the same logic, AD/CV measures do not, in and of themselves, undermine the representativeness of the reference period selected to allocate shares in a TRQ in all circumstances.

⁵⁴⁰ Turkey's second written submission, para. 238.

proceed to examine the special factor argument, which Turkey presented as an alternative to the representative period argument.541

7.9.1.2.2 The special factor argument

7.294. Turkey argues that the change in the share of imports from the exporting countries following the new trade remedies imposed during the first half of 2018 was a "special factor" that required an adjustment to the TRQ allocations.⁵⁴² Turkey asserts that the reduction in imports following the imposition of the new AD/CV measures from the countries subject to those measures constitutes a special factor that has affected or is affecting the trade in the product concerned.⁵⁴³ Referring to the report of the panel in EU - Poultry Meat (China), Turkey argues that the reference in Article XIII:2(d) of the GATT 1994 to special factors that "may be affecting" trade implies that a special factor that the competent authority ought to consider in allocating shares in a TRQ may arise between the end of the representative period and the time of the allocation of the TRQ.⁵⁴⁴

7.295. We agree with Turkey that by referring to "special factors which ... may be affecting" trade in the product concerned, Article XIII:2(d) recognizes the possibility that developments outside the previous representative period could constitute special factors, of which due account must be taken in the allocation of shares. We also note that the panel in EU - Poultry Meat (China) found that "in certain exceptional (i.e. 'special' circumstances), changes in the imports shares held by different Members that have occurred between the end of the representative period selected and the time of the TRQ being allocated" could constitute a special factor under Article XIII:2(d) of the GATT 1994.545 However, in that dispute China put forth evidence before that panel that indicated that the increase in China's imports into the European Union following the relaxation of certain EU sanitary and phytosanitary (SPS) measures after the representative period at issue was "significant[]", "dramatic", "steady", "continuous" and "rapid".⁵⁴⁶ In contrast, Turkey has not placed sufficient evidence before us that would illustrate the existence, or the extent, of the purported "change in the share of imports from the exporting countries" and "the reduction in imports ... from countries subject to those measures" that, according to Turkey, took place following the imposition of the new AD/CV measures by the European Union in 2018.547

7.296. In its first written submission, Turkey provided data illustrating the difference between the country-specific TRQs that the European Commission allocated to Turkey for various product categories and Turkey's annual average import volumes for those product categories based on data from 2015 to June 2018.⁵⁴⁸ According to Turkey, these data show that Turkey would have received larger country-specific shares in the TRQs if the shares had been allocated based on data from 2015 to June 2018, and that Turkey also would have been allocated country-specific TRQs for two additional product categories.⁵⁴⁹ While we agree that the data period that Turkey believes the European Union should have used would have been more beneficial to Turkey for several product categories, this fact, without more, does not establish that special factors existed that the European Commission was required to account for when it allocated the country-specific TRQ shares. The data submitted by Turkey do not show that the difference in the proportions of EU imports from Turkey that Turkey points to arose due to the imposition of trade remedies in the first half of 2018, and indeed Turkey has not even identified those new trade remedies.

7.297. We also note that the data that Turkey provided show that for certain product categories, namely product categories 9, 19, 25, and 26, Turkey's annual average import volume based on data from 2015 to June 2018 was in fact lower than the country-specific TRQ that the European Commission allocated to Turkey based on data from 2015-2017.550 Further, for two product categories, 4 and 27, in respect of which Turkey argues that it would have received a country-specific TRQ had the European Commission used data from 2015 to June 2018, we note

⁵⁴¹ Turkey's response to Panel question No. 43, para. 89.

⁵⁴² Turkey's response to Panel question No. 43, para. 89.

⁵⁴³ Turkey's response to Panel question No. 43, para. 89.

⁵⁴⁴ Turkey's response to Panel question No. 43, para. 89 (referring to Panel Report, *EU – Poultry Meat* (China), para. 7.355). ⁵⁴⁵ Panel Report, *EU – Poultry Meat (China)*, para. 7.363. (emphasis added)

⁵⁴⁶ Panel Report, EU – Poultry Meat (China), paras. 7.364, 7.366, 7.369, and 7.377.

⁵⁴⁷ Turkey's response to Panel question No. 43, para. 89.

⁵⁴⁸ Turkey's first written submission, paras. 349-350.

⁵⁴⁹ Turkey's first written submission, paras. 349-350.

⁵⁵⁰ Turkey's first written submission, para. 349.

that Turkey's data show that the change in Turkey's share in the overall imports would have been only 1% (an increase from 4% to 5%).⁵⁵¹ This is sharply in contrast to the magnitude of the change in China's import share of the two products in respect of which the panel in *EU – Poultry Meat (China)* found that the change in import shares following the relaxation of the SPS measures was a special factor. In that proceeding, the evidence demonstrated that China's share for one of the products rose from 0% in the representative period (2006-2008) to 27.1% in 2009, 40.7% in 2010, and 52.8% in 2011. Likewise, for the other product, China's share changed from 0% in the representative period (2006-2008), to 8.7% in 2009, 17.6% in 2010, and 61.1% in 2011.⁵⁵² Notably, in respect of another product at issue in that case, for which China's import share changed from 0% in the representative period (2006-2008) to 2.8% in 2009, 1.5% in 2010, and 2.9% in 2011, the panel did not find that the change in import shares after the representative period constituted a special factor.⁵⁵³

7.298. Thus, the change in China's import shares of the relevant products following the relaxation of the SPS measures in EU – Poultry Meat (China) was of a significantly greater magnitude than the change in Turkey's import shares of several product categories following the imposition of the new AD/CV measures that Turkey refers to. Therefore, Turkey's reliance on the EU – Poultry Meat (China) panel report to support the special factor argument is inapposite.

7.299. For these reasons, we do not consider that Turkey has established the existence of a special factor that the European Commission improperly failed to take into account in allocating country-specific shares in TRQs for the relevant product categories. We therefore disagree with Turkey's special factor argument.

7.9.1.3 Whether Turkey has established that the definitive safeguard is inconsistent with the *chapeau* of Article XIII:2 of the GATT 1994

7.300. Turkey argues that by failing to take into account the first six months of 2018 in determining the country-specific shares of TRQs, the European Union also acted inconsistently with the *chapeau* of Article XIII:2 of the GATT 1994.⁵⁵⁴ According to Turkey, the country-specific shares determined on the basis of data from 2015-2017 do not approach as closely as possible the shares in the trade of the products concerned that Turkey would be expected to obtain in the absence of the safeguard measure.⁵⁵⁵ In response to a Panel question, Turkey indicated that the same facts and arguments underly Turkey's claims under Article XIII:2(d) and the *chapeau* of Article XIII:2 of the GATT 1994.⁵⁵⁶

7.301. The European Union argues that it acted in full conformity with Article XIII of the GATT 1994 and ensured that the allocation of the TRQs approached as closely as possible the shares that various Members may be expected to obtain in the absence of the TRQs.⁵⁵⁷

7.302. In our evaluation of Turkey's claim under Article XIII:2(d) of the GATT 1994 we have found that Turkey has not established that data for a period including January to June 2018 were representative of trade flows in the absence of restrictions, but data for the period 2015-2017 were not. We have also found that Turkey has not established that the changes in import shares of various countries following the introduction of trade remedies in early 2018 was a special factor that had to be taken into account in the allocation of country-specific shares in the TRQs. Given that Turkey's claim under the *chapeau* of Article XIII:2 of the GATT 1994 rest on these same grounds, we find that Turkey has not established that the European Union acted inconsistently with the *chapeau* of Article XIII:2 of the GATT 1994.

7.9.2 Article 5.2(a) of the Agreement on Safeguards

7.303. Turkey argues that "for the same reasons" as those underlying its claims under the relevant provisions of Article XIII:2(d) of the GATT 1994, the definitive safeguard is also inconsistent with

⁵⁵¹ Turkey's first written submission, para. 350.

⁵⁵² Panel Report, EU – Poultry Meat (China), para. 7.364.

⁵⁵³ Panel Report, EU – Poultry Meat (China), paras. 7.364 and 7.369.

⁵⁵⁴ Turkey's first written submission, para. 352.

⁵⁵⁵ Turkey's first written submission, para. 352.

⁵⁵⁶ Turkey's response to Panel question No. 44, para. 90.

⁵⁵⁷ European Union's first written submission, paras. 291 and 296.

WT/DS595/R

Article 5.2(a) of the Agreement on Safeguards.⁵⁵⁸ According to Turkey, given that the text of Article 5.2(a) is "to a large extent identical" to that of Article XIII:2(d), the European Union's failure to allocate country-specific shares in the TRQs based on a "previous representative period" would give rise to an inconsistency not only with Article XIII:2(d) of the GATT 1994 but also with Article 5.2(a) of the Agreement on Safeguards.^{559, 560}

7.304. Referring to the panel report in US – Line Pipe, the European Union contends that Article 5.2(a) of the Agreement on Safeguards is not applicable to TRQs.⁵⁶¹ Therefore, the European Union request the Panel to reject Turkey's claim under Article 5.2(a) of the Agreement on Safeguards.

7.305. Article 5.2(a) of the Agreement on Safeguards provides:

In cases in which a quota is allocated among supplying countries, the Member applying the restrictions may seek agreement with respect to the allocation of shares in the quota with all other Members having a substantial interest in supplying the product concerned. In cases in which this method is not reasonably practicable, the Member concerned shall allot to Members having a substantial interest in supplying the product shares based upon the proportions, supplied by such Members during a previous representative period, of the total quantity or value of imports of the product, due account being taken of any special factors which may have affected or may be affecting the trade in the product.

7.306. The European Union, relying on a finding of the panel in US – Line Pipe, argues that a TRQ is not a "quota" in the sense of Article 5.2(a) of the Agreement on Safeguards.⁵⁶² Turkey, on the other hand, disagrees with that finding of the panel in US – Line Pipe and argues that TRQs include a "quota" component, and that Article 5.2(a) applies to TRQs.⁵⁶³

7.307. While the parties disagree as to whether Article 5.2(a) of the Agreement on Safeguards applies to TRQs and therefore to the measure at issue, we consider that even assuming *arguendo* that it does, Turkey has not established that the European Union acted inconsistently with the provision. As noted above, Turkey asserts, on the same grounds as those that it relied on in the context of its claim under Article XIII:2(d) of the GATT 1994, that the European Union's failure to allocate country-specific shares in the TRQs based on a "previous representative period" gave rise to an inconsistency with Article 5.2(a) of the Agreement on Safeguards.⁵⁶⁴ However, as we have found above, Turkey has not established that the period used by the European Commission for the allocation of country-specific shares in TRQs did not qualify as a "previous representative period" under Article XIII:2(d) of the GATT 1994.⁵⁶⁵ On this *arguendo* basis, we consider that the relevant findings that we made when addressing Turkey's claim under Article XIII:2(d) of the GATT 1994 apply to Turkey's claim under Article 5.2(a) of the Agreement on Safeguards. Therefore, Turkey has not established that the definitive safeguard was inconsistent with Article 5.2(a) of the Agreement on Safeguards. Therefore, Turkey has not established that the definitive safeguard was inconsistent with Article 5.2(a) of the Agreement on Safeguards.

⁵⁵⁸ Turkey's first written submission, para. 353; second written submission, para. 242.

⁵⁵⁹ Turkey's second written submission, para. 242.

⁵⁶⁰ See also United States' third-party response to Panel question No. 9, para. 27;

Canada's third-party response to the Panel question No. 9, paras. 3-5; Switzerland's third-party response to Panel question No. 9, paras. 13-17; and Brazil's third-party response to Panel question No. 9, paras. 25-28. ⁵⁶¹ European Union's first written submission, para. 285 (referring to Panel Report, *US – Line Pipe*,

para. 7.69); response to Panel question No. 45, para. 100.

⁵⁶² European Union's response to Panel question No. 45, para. 100.

⁵⁶³ Turkey's response to Panel question No. 45, paras. 95-96.

⁵⁶⁴ Turkey's second written submission, para. 242.

⁵⁶⁵ We note that the parties have not argued that the notion of "a previous representative period" under Article 5.2(a) of the Agreement on Safeguards differs substantively from that under Article XIII:2(d) of the GATT 1994. We need not, and do not, take any view on this issue, because Turkey has argued that the definitive safeguard is inconsistent with Article 5.2(a) of the Agreement on Safeguards "for the same reasons" as those underlying Turkey's claim under Article XIII:2(d) of the GATT 1994, and we have disagreed with those reasons in addressing Turkey's claim under Article XIII:2(d) of the GATT 1994.

7.10 Reduction in the pace of liberalization and increased restrictiveness

7.308. Turkey claims that the definitive safeguard was inconsistent with Articles 7.4 and 5.1 of the Agreement on Safeguards because the first and the second review regulations⁵⁶⁶ made it more restrictive and reduced its pace of liberalization.⁵⁶⁷ The European Union contends that Turkey's claims under Articles 7.4 and 5.1 of the Agreement on Safeguards ought to be rejected.⁵⁶⁸

7.309. In the sections below, we first address Turkey's claim under Article 7.4 of the Agreement on Safeguards (section 7.10.1) and we then turn to Article 5.1 of the Agreement on Safeguards (section 7.10.2).

7.10.1 Article **7.4** of the Agreement on Safeguards

7.310. Turkey argues that the definitive safeguard was inconsistent with Article 7.4 of the Agreement on Safeguards for the following reasons:

- a. by reducing the pace of liberalization of the TRQs from 5% at the end of the first and the second years of the application of the definitive safeguard (as stipulated in the definitive determination) to 3% at the same points in time (pursuant to the first review regulation), the European Union acted inconsistently with Article 7.4 of the Agreement on Safeguards⁵⁶⁹; and
- b. certain modifications made to the definitive safeguard through the first and the second review regulations made the definitive safeguard more restrictive than originally determined, inconsistently with Article 7.4 of the Agreement on Safeguards.⁵⁷⁰

7.311. Below, we first recall the applicable requirements of Article 7.4 of the Agreement on Safeguards (section 7.10.1.1), and we then address each of these arguments, in turn (sections 7.10.1.2.1 and 7.10.1.2.2).

7.10.1.1 The applicable requirements of Article 7.4 of the Agreement on Safeguards

7.312. The first sentence of Article 7.4 of the Agreement on Safeguards provides:

In order to facilitate adjustment in a situation where the expected duration of a safeguard measure as notified under the provisions of paragraph 1 of Article 12 is over one year, the Member applying the measure shall progressively liberalize it at regular intervals during the period of application.

7.313. The first sentence of Article 7.4 of the Agreement on Safeguards applies to measures with an expected duration of over one year. The first sentence provides that the covered safeguards "shall" be liberalized progressively at regular intervals. Thus, the first sentence imposes the obligation on the Member applying a covered safeguard to liberalize the measure (a) "progressively" and (b) at "regular intervals". The ordinary meaning of "progressive" includes "[p]roceeding by steps or stages".⁵⁷¹ The provision does not prescribe what the magnitude of progressive liberalization should be at each liberalizing step. The ordinary meaning of "regular" includes "[r]ecurring or taking place repeatedly at (short) uniform intervals" and "[r]ecurring or repeated at fixed times (although not necessarily at uniform intervals)".⁵⁷² The provision does not prescribe how long the regular intervals between each liberalizing step must be, or at what point in time the progressive liberalization must begin.⁵⁷³ The first sentence of Article 7.4 also indicates, however, that the

⁵⁶⁶ See para. 2.4 above.

⁵⁶⁷ Turkey's first written submission, para. 355.

⁵⁶⁸ European Union's first written submission, para. 306.

⁵⁶⁹ Turkey's first written submission, para. 365.

⁵⁷⁰ Turkey's first written submission, para. 366.

⁵⁷¹ Oxford Dictionaries online, definition of "progressive", meaning 3

https://www.oed.com/view/Entry/152244?redirectedFrom=progressive#eid (accessed 22 September 2021). ⁵⁷² Oxford Dictionaries online, definition of "regular", meanings c and d

https://www.oed.com/view/Entry/161414?redirectedFrom=regular#eid (accessed 22 September 2021).

See also Panel Report, Ukraine - Passenger Cars, para. 7.362.

⁵⁷³ Panel Report, Ukraine – Passenger Cars, paras. 7.363-7.364.

purpose of the requirements contained in the sentence is "to facilitate adjustment" of the domestic industry.

7.10.1.2 Whether Turkey has established that the definitive safeguard is inconsistent with Article 7.4 of the Agreement on Safeguards

7.10.1.2.1 Whether the reduction in the pace of liberalization was inconsistent with Article **7.4**

7.314. We recall that in the definitive determination, the European Commission announced that it would increase the size of each TRQ by 5% at the end of the first and second years of the application of the definitive safeguard.⁵⁷⁴ Subsequently, in the first review regulation, the European Commission changed the pace of liberalization to "3% + 3% for the second and third year of application of the safeguard measures".⁵⁷⁵

7.315. Turkey argues that the requirement under Article 7.4 of the Agreement on Safeguards to "progressively liberalize" the safeguard "at regular intervals" implies that once the competent authorities have determined a schedule of liberalization, they cannot decrease the pace of liberalization.⁵⁷⁶ Turkey argues that the object and purpose of Article 7.4 of the Agreement on Safeguards, i.e. to "facilitate adjustment", also supports Turkey's view that Article 7.4 does not permit a reduction in the pace of liberalization once a schedule of liberalization has been announced, because a reduced pace would create a disincentive for the domestic industry to undertake adjustment.⁵⁷⁷ Turkey posits that the requirement of progressive liberalization is consistent with the reference in the preamble of the Agreement on Safeguards to "the importance of structural adjustment and the need to enhance rather than limit competition in international markets".⁵⁷⁸ Turkey argues that the immediate context provided by the second and third sentences of Article 7.4 of the Agreement on Safeguards by reducing the pace of liberalization from 5% to 3% for the second and the third years of the application of the definitive safeguard.⁵⁸⁰

7.316. The European Union argues that by setting the pace of liberalization at 3% for the second and third years of the application of the definitive safeguard, it fulfilled the requirement of progressive liberalization set out in Article 7.4 of the Agreement on Safeguards.⁵⁸¹ The European Union contends that nothing in the text of the Agreement on Safeguards supports Turkey's view that once a competent authority has determined a schedule of liberalization, it cannot decrease the pace of liberalization.⁵⁸² The European Union argues that acceptance of Turkey's view would imply that a Member that announces a pace of liberalization of 5% after each year of operation of the measure and then revises it to 3% will be found to have acted inconsistently with Article 7.4, but not a Member that applies the 3% liberalization pace after each year of operation of the measure without making any initial announcement. According to the European Union, this would incentivize Members to announce a low pace of liberalization initially, thus forestalling the announcement of a higher pace of liberalization that would provide a "more demanding signal to the domestic industry as regards how it is expected to adjust".⁵⁸³

7.317. Among the third parties, the Republic of Korea takes the view that, as there are no WTO disciplines governing the degree of liberalization that a Member must implement, as long as a safeguard measure applied for longer than one year is liberalized at regular intervals, the degree of each liberalization could vary.⁵⁸⁴ For the United Kingdom, Article 7.4 of the Agreement on Safeguards does not prohibit a reduction in the pace of liberalization after an initial schedule of liberalization had

⁵⁷⁴ Definitive determination, (Exhibit TUR-5), recital 188; First review regulation, (Exhibit TUR-9), recital 134.

⁵⁷⁵ First review regulation, (Exhibit TUR-9), recital 147.

⁵⁷⁶ Turkey's first written submission, para. 356; second written submission, paras. 247-249.

⁵⁷⁷ Turkey's second written submission, para. 252.

⁵⁷⁸ Turkey's second written submission, para. 253.

⁵⁷⁹ Turkey's first written submission, paras. 357-358; second written submission, para. 250.

⁵⁸⁰ Turkey's first written submission, para. 365; second written submission, para. 254.

⁵⁸¹ European Union's first written submission, paras. 300 and 302.

⁵⁸² European Union's first written submission, paras. 301-302.

⁵⁸³ European Union's first written submission, para. 302.

⁵⁸⁴ Korea's third-party statement, para. 31.

been set, as long as the revised pace of liberalization results in progressive liberalization.⁵⁸⁵ Switzerland argues that Article 7.4 of the Agreement on Safeguards does not permit a reduction in the pace of liberalization.⁵⁸⁶

7.318. We recall that Turkey refers to the first sentence of Article 7.4, which requires Members to progressively liberalize safeguards imposed for more than a year at regular intervals during the period of application, as the basis for its argument.⁵⁸⁷ The question before us, therefore, is whether the first sentence of Article 7.4 precludes a Member from reducing the rate of liberalization to be achieved at regular intervals after initially announcing a higher rate, as the European Union did in this case. We consider that nothing in the ordinary meaning of the text of Article 7.4 of the Agreement on Safeguards indicates that a Member is necessarily precluded from doing so. Article 7.4 requires Members to "progressively liberalize ... at regular intervals". As explained above, this means that there must be liberalization, in steps, at regular intervals. The liberalization of a safeguard measure can be "progressive", i.e. proceeding by steps or stages, even if a Member reduces the magnitude of liberalization to be achieved at "regular intervals" after initially announcing a different magnitude.⁵⁸⁸

7.319. Turkey argues that the "object and purpose of Article 7.4" is "to facilitate adjustment", and that a reduction in the pace of liberalization relative to the initially announced pace is contrary to that object and purpose.⁵⁸⁹ Turkey refers to the observation of the panel in Ukraine – Passenger *Cars* that "[d]elaying liberalization in this way could create a disincentive for the domestic industry to undertake appropriate efforts at adjustment from the outset of the period of application".⁵⁹⁰ We note that the panel in Ukraine - Passenger Cars expressed the concern that Turkey refers to in relation to the issue of a Member "not taking any liberalization steps until a late stage in the period of application of a safeguard measure".⁵⁹¹ That issue is not the same as the issue before us, i.e. whether Article 7.4 precludes a Member from reducing the magnitude of liberalization to be achieved at regular intervals after initially announcing a different magnitude. Therefore, we consider Turkey's reliance on this observation of the panel in Ukraine – Passenger Cars to be inapposite. Aside from this reference to the panel in Ukraine - Passenger Cars, Turkey has not provided any other basis in support of its assertion that a reduction in the pace of liberalization relative to the initially announced pace does not facilitate adjustment. We note that even if the Member applying the safeguard reduces the magnitude of liberalization to be achieved at regular intervals after initially announcing a different magnitude, the domestic industry will be exposed to a higher amount of competition from imports after each liberalization. Turkey has not shown why such liberalization will not facilitate adjustment. Therefore, we disagree with Turkey's argument concerning the objective of Article 7.4.

7.320. We recall Turkey's argument that the context provided by the first and the second sentence of Article 7.4 of the Agreement on Safeguards also supports its view that once announced, the pace of liberalization cannot be decreased. Referring to the report of the panel in *Argentina – Footwear (EC)*, Turkey argues that the only modifications to a safeguard that the second and third sentences contemplate are those that decrease its restrictiveness.⁵⁹² The second and third sentences of Article 7.4 provide:

If the duration of the measure exceeds three years, the Member applying such a measure shall review the situation not later than the mid-term of the measure and, if

⁵⁸⁹ Turkey's first written submission, para. 357; second written submission, para. 252.

⁵⁹⁰ Turkey's second written submission, para. 252 (referring to Panel Report, *Ukraine – Passenger Cars*, para. 7.362).

⁵⁹² Turkey's first written submission, para. 358; second written submission, para. 250 (referring to Panel Report, *Argentina – Footwear (EC)*, para. 8.303).

⁵⁸⁵ United Kingdom's third-party statement, para. 14; third-party submission, paras. 21-22.

⁵⁸⁶ Switzerland's third-party submission, para. 29.

⁵⁸⁷ Turkey's first written submission, para. 356.

⁵⁸⁸ We note that the panel in *Ukraine – Passenger Cars* found that "a Member can ... comply with its obligation in Article 7.4 even if it has not previously provided a timetable for progressive liberalization". (Panel Report, *Ukraine – Passenger Cars*, para. 7.360). The finding made by that panel accords with our understanding of Article 7.4 of the Agreement on Safeguards. In particular, as noted by the Republic of Korea, if there is no obligation in Article 7.4 to provide a timetable for liberalization, it seems difficult to conclude that a change in the magnitude of liberalization to be achieved at "regular intervals" relative to the initially announced magnitude would *per se* be a violation of Article 7.4. (See Korea's third-party submission, para. 32).

⁵⁹¹ Panel Report, *Ukraine – Passenger Cars*, para. 7.362.

appropriate, withdraw it or increase the pace of liberalization. A measure extended under paragraph 2 shall not be more restrictive than it was at the end of the initial period, and should continue to be liberalized.

7.321. The second sentence applies to safeguards whose duration exceeds three years. The second sentence requires the Member applying such a safeguard to withdraw the measure, or increase the pace of liberalization, "if appropriate" in light of a review to be performed not later than the mid-term of the measure. We note that nothing in the second sentence of Article 7.4 bears upon the question as to whether the pace of liberalization can be slowed relative to the initially announced pace in situations other than one in which an increase in the pace of liberalization is found to be "appropriate" pursuant to a review of a measure the duration of which exceed three years. Therefore, we do not consider that the context provided by the second sentence of Article 7.4 supports the view that the first sentence of Article 7.4 necessarily precludes a reduction in the pace of liberalization of a safeguard measure relative to the initially announced pace.

7.322. The third sentence of Article 7.4 applies to "[a] measure extended under paragraph 2", i.e. a safeguard that is extended beyond the duration of four years in accordance with Article 7.2 of the Agreement on Safeguards. The third sentence stipulates that such a measure shall not be more restrictive than it was at the end of the initial period of application. Although this sentence contemplates an upper limit to the restrictiveness of a certain category of safeguards, in respect of liberalization the sentence only states that the measure "should continue to be liberalized". This sentence says nothing about *the pace of* liberalization even in relation to the specific type of safeguards to which it applies. Thus, we do not see anything in the third sentence of Article 7.4 that supports Turkey's view that the first sentence of Article 7.4 necessarily precludes a reduction in the pace of liberalization relative to the initially announced pace.

7.323. We also consider that Turkey's reliance on the observation of the panel in *Argentina – Footwear (EC)* that "[t]he only modifications of safeguard measures that Article 7.4 contemplates are those that *reduce* its [sic] restrictiveness (i.e., to eliminate the measure or to increase the pace of its liberalisation pursuant to a mid-term review)" to be inapposite.⁵⁹³ The panel made this observation in the context of its evaluation of a claim under Article 12 of the Agreement on Safeguards concerning Argentina's alleged failure to notify certain modifications to the safeguard at issue that, in the EC's view, made the measure more restrictive than originally applied. The panel noted that the Agreement on Safeguards does not contemplate restrictive modifications to a safeguard, and hence does not contain notification requirements for such modifications. However, the panel explicitly declined to express a view as to the substantive compliance of such modifications with Article 7 of the Agreement on Safeguards.⁵⁹⁴ Thus, the panel report referred to by Turkey does not support Turkey's arguments concerning the Article 7.4 claim.

7.324. We are therefore not persuaded that the first sentence of Article 7.4 precludes a Member from reducing the magnitude of liberalization to be achieved at regular intervals after initially announcing a different magnitude.

7.325. Turkey contends that the European Commission's adjustment of the pace of liberalization on the ground that the liberalization foreseen in the definitive determination could cause the import volume in the third year of the application to reach the same level as that in 2018, was improper.⁵⁹⁵ Agreeing with Switzerland, Turkey contends that adjusting the pace of liberalization based on developments after the imposition of the definitive safeguard that are unrelated to the increased imports on the basis of which the safeguard was applied, would fail to achieve the purpose of facilitating adjustment to the new competitive conditions caused by the increased imports.⁵⁹⁶ Pursuant to the said adjustment, even though the TRQs would be liberalized by 3% for the second and the third years of the definitive safeguard's application instead of 5% as foreseen in the definitive determination, the domestic industry would nevertheless be exposed to higher levels of import competition because of that liberalization. Turkey has neither explained nor pointed to anything in the record evidence that would show why such liberalization would not facilitate adjustment of the domestic industry in this case.

⁵⁹³ Panel Report, *Argentina – Footwear (EC)*, para. 8.303. (emphasis original)

⁵⁹⁴ Panel Report, Argentina – Footwear (EC), para. 8.303.

⁵⁹⁵ Turkey' second written submission, para. 255.

⁵⁹⁶ Turkey's second written submission, para. 257.

WT/DS595/R

7.326. Thus, we disagree with Turkey's argument that by reducing the pace of liberalization of the TRQs from 5% for the second and the third years of the definitive safeguard's application as stipulated in the definitive determination, to 3% through the first review regulation, the European Union made the definitive safeguard inconsistent with Article 7.4 of the Agreement on Safeguards.

7.10.1.2.2 Whether certain modifications to the definitive safeguard resulted in an inconsistency with Article **7.4**

7.327. Turkey argues that the definitive safeguard is inconsistent with Article 7.4 of the Agreement on Safeguards because certain modifications introduced with the first and second review regulations made it more trade restrictive than originally determined.⁵⁹⁷ The European Union argues that the modifications to the definitive safeguard made through the first and second review regulations are consistent with Article 7.4 of the Agreement on Safeguards.⁵⁹⁸

7.328. Among the third parties, Switzerland argues that the Panel must assess whether the modifications that Turkey points to made the safeguard more restrictive and, if so, it must conclude that such modifications are inconsistent with Article 7.4 of the Agreement on Safeguards.⁵⁹⁹

7.329. Turkey identifies the following modifications to the definitive safeguard, made through the first and the second review regulations, as having made the definitive safeguard more restrictive than originally determined:

- a. the establishment of a limitation for each exporting country not to exceed a share of 30% of the global TRQ for product category 1^{600} ;
- b. the restriction of imports under product category 4.B to those exporters that could demonstrate an end-use in the automotive sector⁶⁰¹;
- c. the imposition of a cap of 30% per country on the use of the global TRQ for product categories 13 and 16^{602} ;
- d. the reduction of the pace of liberalization⁶⁰³;
- e. the decision to manage country-specific TRQs on a quarterly basis⁶⁰⁴;
- f. the division of the global TRQ for product category 1 into country-specific TRQs and a residual TRQ 605 ; and
- g. the revision of the system of access to the residual TRQ during the last quarter for the third year of the definitive safeguard for countries having a country-specific TRQs.⁶⁰⁶

7.330. We note that in respect of all the modifications listed in the previous paragraph, apart from items (b) and (d), Turkey has only identified the modifications concerned and asserted that the modifications made the definitive safeguard more restrictive than originally determined in a manner inconsistent with Article 7.4 of the Agreement on Safeguards. We do not consider that the mere identification of these modifications is sufficient, without any accompanying explanation, to demonstrate that the modifications made the definitive safeguard more restrictive than originally determined in a manner inconsistent with Article 7.4 of the Agreement on Safeguards. We therefore consider that the modifications made the definitive safeguard more restrictive than originally determined in a manner inconsistent with Article 7.4 of the Agreement on Safeguards. We therefore consider that Turkey has not established a *prima facie* case that the modifications identified in

⁵⁹⁷ Turkey's first written submission, para. 366; second written submission, para. 263.

⁵⁹⁸ European Union's first written submission, para. 305.

⁵⁹⁹ Switzerland's third-party submission, para. 31.

⁶⁰⁰ First review regulation, (Exhibit TUR-9), Article 1(1)(a), recital 26.

⁶⁰¹ First review regulation, (Exhibit TUR-9), Article 1(1)(a), recitals 29-40.

⁶⁰² First review regulation, (Exhibit TUR-9), Article 1(1)(b), recital 94.

⁶⁰³ First review regulation, (Exhibit TUR-9), Article 1(2)(b), recital 156.

⁶⁰⁴ Second review regulation, (Exhibit TUR-12), Article 1(1)(a) and Annex II, recital 40.

⁶⁰⁵ Second review regulation, (Exhibit TUR-12), Article 1(1)(a) and Annex II, recital 49.

⁶⁰⁶ Second review regulation, (Exhibit TUR-12), Article 1(1)(b) and Annex III, recital 88.

items (a), (c), (e), (f), and (g) above made the definitive safeguard more restrictive than originally determined in a manner inconsistent with Article 7.4 of the Agreement on Safeguards.

7.331. We therefore turn to address Turkey's argument concerning the modification to the definitive safeguard listed in item (b) in paragraph 7.329 above, namely the restriction of imports under product category 4.B to those exporters that could demonstrate an end-use in the automotive sector. In the first review regulation, the European Commission introduced a requirement for importers to demonstrate that the product concerned was intended for end-use in the automotive sector, in order to be categorized in product category 4.B (automotive end-use requirement).⁶⁰⁷ At the same time, the European Commission extended and revised the scope of product category 4.A to ensure that products that were no longer eligible to be categorized in product category 4.B could be categorized as belonging to product category 4.A.⁶⁰⁸ Turkey argues that with the introduction of the automotive end-use requirement, the access to the TRQ for product category 4.B became more burdensome, and thus more restrictive than originally determined and therefore inconsistent with Article 7.4 of the Agreement on Safeguards.⁶⁰⁹

7.332. We note, and the parties confirmed⁶¹⁰ in response to a question from the Panel, that the automotive end-use requirement was revoked retroactively through the amending regulation of 15 January 2020, less than four months after being introduced through the first review regulation, which entered into force on 1 October 2019.⁶¹¹ Given that the automotive end-use requirement is no longer in force, we need not consider Turkey's argument concerning the automotive end-use requirement in order to resolve the present dispute, and we decline to do so.

7.333. As regards Turkey's argument concerning item (d) in paragraph 7.329 above, namely the reduction in the initially announced pace of liberalization, we have already rejected Turkey's arguments concerning the consistency of this modification with Article 7.4 of the Agreement on Safeguards in section 7.10.1.2.1 above.

7.10.2 Article **5.1** of the Agreement on Safeguards

7.334. Turkey argues that by making the definitive safeguard more restrictive than originally determined and by reducing the pace of liberalization through the first and the second review regulations, the European Union failed to apply the definitive measure only to the extent necessary to prevent serious injury and to facilitate adjustment, thus acting inconsistently with Article 5.1 of the Agreement on Safeguards.⁶¹² Specifically, Turkey argues that by increasing the restrictiveness and reducing the pace of liberalization of the definitive safeguard when the assessment of injury or threat thereof remained the same, the European Union no longer applied the definitive safeguard only to the extent "necessary to prevent or remedy serious injury".⁶¹³ Further, according to Turkey, by making the definitive safeguard more restrictive and reducing the pace of liberalization, the European Union also created a disincentive for the domestic industry to undertake appropriate efforts at adjustment.⁶¹⁴

7.335. The European Union contends that the Panel must reject Turkey's claim under Article 5.1 of the Agreement on Safeguards. 615

7.336. We note that Turkey relies on the same facts and arguments to support its claim under Article 5.1 of the Agreement on Safeguards as it did to support its claim under Article 7.4 of the Agreement on Safeguards, addressed in section 7.10.1.2 above. We found above in

⁶¹² Turkey's first written submission, para. 367. We recall that Turkey also presented other arguments in support of its claim under Article 5.1, which we have examined separately, in section 7.8.1 above.

⁶⁰⁷ First review regulation, (Exhibit TUR-9), recital 35.

⁶⁰⁸ First review regulation, (Exhibit TUR-9), recital 37.

⁶⁰⁹ Turkey's second written submission, para. 262.

⁶¹⁰ Turkey's response to Panel question No. 53, para. 1; European Union's response to Panel question No. 53, para. 7.

⁶¹¹ European Commission Implementing Regulation (EU) 2020/35, (Exhibit TUR-10), recital 9 ("[t]he present Regulation should be given retroactive effect, thereby revoking, as from 1 October 2019, the end-use requirement introduced by the amending Regulation"); First review regulation, (Exhibit TUR-9), Article 3, recital 36.

⁶¹³ Turkey's first written submission, para. 367.⁶¹⁴ Turkey's first written submission, para. 367.

⁶¹⁵ European Union's first written submission, para. 306.

section 7.10.1.2.2 that Turkey did not provide any explanation as to how the relevant modifications to the definitive safeguard made through the first and the second review regulations that Turkey points to made the measure more restrictive in a manner that would give rise to an inconsistency with Article 7.4 of the Agreement on Safeguards.⁶¹⁶ We also rejected in section 7.10.1.2.1 Turkey's contention that the reduction in the pace of liberalization of the definitive safeguard necessarily creates a disincentive for the domestic industry to undertake adjustment. Given that Turkey's claim under Article 5.1 rests on the same grounds and suffers from the same flaws, we find that Turkey has not established that the European Union acted inconsistently with Article 5.1 of the Agreement on Safeguards.

7.11 Article II:1 of the GATT 1994

7.337. Turkey also claims that the definitive safeguard is inconsistent with Article II:1(b) of the GATT 1994. Turkey argues that the 25% out-of-quota duty set out in the definitive safeguard constitutes "other duties or charges" within the meaning of Article II:1(b), and that as a result of the claimed inconsistencies with Article XIX:1(a) of the GATT 1994, those duties or charges are not excused by Article XIX and are therefore inconsistent with Article II:1(b).⁶¹⁷

7.338. We recall that we have already found that the definitive safeguard (i) is inconsistent with Article XIX:1(a), because the European Commission did not establish that the increase in imports was "as a result of" the unforeseen developments it relied upon and did not identify the obligations whose effect resulted in the injurious increase in imports, and (ii) is inconsistent with Article 4.1(b) of the Agreement on Safeguards, because it rests on a determination of threat of injury that is not "based on facts". In view of this, we do not find it necessary to examine whether, for the same reasons, the definitive safeguard is also inconsistent with Article II:1 of the GATT 1994.

8 CONCLUSIONS, RECOMMENDATION, AND REQUEST FOR A SUGGESTION

- 8.1. For the reasons set out in this Report, we conclude as follows:
 - a. Turkey *has established* that the definitive safeguard is inconsistent with:
 - i. Article XIX:1(a) of the GATT 1994, because the European Commission did not ascertain that the increase in imports took place as a result of the unforeseen developments it had identified, and did not identify in its published reports the obligations whose effect resulted in the increase in imports; and
 - ii. Article 4.1(b) of the Agreement on Safeguards, because two central elements of the European Commission's determination of a threat of serious injury were not "based on facts" as required by that provision.
 - b. Turkey *has not established* that the definitive safeguard is inconsistent with:
 - i. Article XIX:1(a) of the GATT 1994 and Articles 2.1, 3.1, 4.1(c), 4.2(a), 4.2(b), and 4.2(c) of the Agreement on Safeguards because of the European Commission's approach to product scope;
 - ii. Article XIX:1(a) of the GATT 1994 and Articles 2.1 and 4.2(a) of the Agreement on Safeguards because of alleged errors in the European Commission's determination of an increase in imports;
 - iii. Article 5.1 of the Agreement on Safeguards because the European Commission did not take into account data from the first six months of 2018 in determining the size of the TRQs, or because of the double remedy regulation, or because the European Union reduced the pace of liberalization or allegedly made the measure more restrictive;

⁶¹⁶ As noted in para. 7.332 above, we need not make findings on Turkey's arguments concerning the automotive end-use requirements as that requirement is no longer in force.

⁶¹⁷ Turkey's first written submission, paras. 375-380.

- iv. Article XIII:2(d) and the *chapeau* of Article XIII:2 of the GATT 1994 and Article 5.2(a) of the Agreement on Safeguards because the European Commission did not take into account data from the first six months of 2018 in allocating the TRQs; and
- v. Article 7.4 of the Agreement on Safeguards because the European Union reduced the pace of liberalization and allegedly made the measure more restrictive.
- c. We do not consider it necessary to decide whether the definitive safeguard is inconsistent with:
 - i. Articles 4.1(a) and 4.2(a) of the Agreement on Safeguards for reasons substantively identical to those put forward under Article 4.1(b) of the Agreement on Safeguards;
 - ii. Article 2.1 of the Agreement on Safeguards as a consequence of the inconsistency with Article 4.1(b) of the Agreement on Safeguards;
 - iii. Article 4.2(b) of the Agreement on Safeguards because of errors in the determination of causation;
 - iv. Articles 2.1, 5.1, and 7.1 of the Agreement on Safeguards as a consequence of the claimed inconsistency with Article 4.2(b) of the Agreement on Safeguards; and
 - v. Article II:1(b) of the GATT 1994 as a consequence of the inconsistency with Article XIX:1(a) of the GATT 1994.
- d. The automotive end-use requirement is no longer in force, and we do not consider it in reviewing the definitive safeguard.
- e. The provisional safeguard is no longer in force, and we do not make findings on its consistency with the covered agreements.

8.2. Under Article 3.8 of the DSU, "where there is an infringement of the obligations assumed under a covered agreement, the action is considered *prima facie* to constitute a case of nullification or impairment". We conclude that, to the extent that the definitive safeguard is inconsistent with the GATT 1994 and the Agreement on Safeguards, it has nullified or impaired benefits accruing to Turkey under those agreements.

8.3. Pursuant to Article 19.1 of the DSU, we recommend that the European Union bring its measure into conformity with the Agreement on Safeguards and the GATT 1994.

8.4. In addition to requesting recommendations, Turkey asks us to suggest that the European Union implement our recommendation by revoking the safeguard at issue, pursuant to Article 19.1 of the DSU.⁶¹⁸

8.5. Article 19.1 of the DSU contains two prongs. First, "[w]here a panel or the Appellate Body concludes that a measure is inconsistent with a covered agreement, it *shall* recommend that the Member concerned bring the measure into conformity with that agreement".⁶¹⁹ Thus, panels are required to make such a recommendation when an inconsistency is established. Second, "[i]n addition ... the panel or Appellate Body *may* suggest ways in which the Member concerned could implement the recommendations".⁶²⁰ Thus, suggestions are additional to recommendations, and the choice whether to make such suggestions falls squarely within the discretion of each panel.⁶²¹

⁶¹⁸ Turkey's first written submission, para. 382.

⁶¹⁹ Emphasis added; fn omitted.

⁶²⁰ Emphasis added.

⁶²¹ See also e.g. Appellate Body Report, US – Continued Zeroing, para. 389.

8.6. Some panels have made suggestions under Article 19.1 of the DSU.⁶²² Others have preferred to decline to do so, reasoning in particular that the choice on the manner of implementation is, in the first place, for the Member concerned.⁶²³

8.7. In this case, we decline to make a suggestion under Article 19.1 of the DSU.

 ⁶²² Panel Reports, Argentina – Poultry Anti-Dumping Duties, paras. 8.3-8.7; US – Offset Act
 (Byrd Amendment), para. 8.6; US – Cotton Yarn, paras. 8.4-8.5; Guatemala – Cement II, paras. 9.4-9.6; US – Underwear, paras. 8.1-8.3; Mexico – Steel Pipes and Tubes, paras. 8.7-8.12; Ukraine – Passenger Cars, paras. 8.7-8.8; US – Lead and Bismuth II, para. 8.2; India – Quantitative Restrictions, paras. 7.1-7.7; EC – Bananas III (Article 21.5 – Ecuador), paras. 6.154-6.159; EC – Trademarks and Geographical Indications (Australia) and EC – Trademarks and Geographical Indications (US), paras. 8.4-8.5; EC – Export Subsidies on Sugar (Australia), EC – Export Subsidies on Sugar (Brazil), and EC – Export Subsidies on Sugar (Thailand), paras. 8.6-8.8; and Pakistan – BOPP Film (UAE) [appealed by Pakistan 22 February 2021], para. 9.6.
 ⁶²³ See e.g. Panel Report, Australia – Anti-Dumping Measures on Paper, para. 8.6.